DETERMINATION OF MERGER NOTIFICATION M/18/063 -

BERENDSEN (ELIS) / KINGS LAUNDRY

Dated 8 July 2019



Competition and Consumer Protection Commission

# Table of Contents

Section	Page
1. Introduction	4
Introduction	4
The Undertakings Involved	4
Rationale for the Proposed Transaction	5
Phase 1: Preliminary Investigation ("Phase 1")	5
Contacts with the Parties	5
Third Party Submissions	6
The Phase 1 Determination	7
Phase 2: Full Investigation ("Phase 2")	7
Contacts with the Parties	7
Phase 2 Proposals	
Third Party Submissions10	D
2. Industry Background – Flat linen Rental and Maintenance 12	
Industry Overview 12	1
Nature of competition 13	
Increasing demand for outsourced flat linen rental and maintenance services 14	4
3. Relevant Product and Geographic Markets 16	6
Horizontal Overlap10	
Relevant Product and Geographic Markets16	6
4. Competitive Assessment – unilateral effects	
Counterfactual	3
Competitive Assessment	
The hospitality market	5
The nature of pre-merger competition55	
Market Structure	
Market Concentration	
Competitive Effects Analysis 62	
The healthcare market94	
The nature of pre-merger competition94	
Market Structure	
Market Concentration	
Competitive Effects Analysis	
Coordinated Effects 130	
Vertical Relationship	
Efficiencies	
5. Remedies to address SLC concerns	
Proposals Submitted by Berendsen	
6. Ancillary restraints	
<b>7.</b> Conclusion	
8. Determination	
9. Berendsen's Proposals 146	
<b>10.</b> Appendix 1 – Summary of Customers' and Self-suppliers' Responses 157	7



Competition and Consumer Protection Commission

11.	Appendix 2 – The Amárach Survey	167
12.	Appendix 3 – Economic Reports	168
13.	Appendix 4 – Organisations with OPLs	169
14.	Appendix 5 – Glossary	171

# List of Tables and Figures

## Table/Figure number Page Table 1: Hospitality sector respondents' replies to a SSNIP question, by respondent size Table 2: Revenue from the provision of flat linen rental and maintenance services to hospitality customers in the State, 2017 (the parties' estimates) [> REDACTED] ...... 57 Table 3: Revenue from the provision of outsourced flat linen rental and maintenance services to hospitality customers in the State, 2014 - 30 June 2018 [X REDACTED] .. 58 Table 4: Revenue from the provision of outsourced flat linen rental and maintenance services in the hospitality market by region, 2017 [X REDACTED] ...... 60 Table 5: The HHI in the hospitality market, H1 2018......61 Table 6: Average price of flat linen items in the hospitality market in 2018 [🏁 Table 7: Linen laundries' turnover in the hospitality market by region, 1 January 2017 -31 December 2017 [🏁 REDACTED] ...... 73 Table 9: Berendsen's diversion ratios in the hospitality market, 2014 - 30 June 2018 Table 10: Kings Laundry's diversion ratios in the hospitality market, 2014 - 30 June Table 11: Revenue from the provision of flat linen rental and maintenance services to healthcare customers in the State, 2017 (the parties' estimates) [> REDACTED]...... 95 Table 12: Revenue from the provision of flat linen rental and maintenance services to Table 14: Bidders facing Berendsen – the healthcare market 2014 – 30 June 2018 🎼 Table 15: Bidders facing Kings Laundry – the healthcare market, 2014 - 30 June 2018 Table 16: Winners in tenders for flat linen rental and maintenance services in the healthcare market that Berendsen participated in, 2014 – 30 June 2018 [X Table 17: Winners in tenders for flat linen rental and maintenance services in the healthcare market that Kings Laundry participated in, 2014 – 30 June 2018 [X Table 18: Berendsen's diversion ratios in the healthcare market, 2014 - 30 June 2018 [% REDACTED]......114



Table 19: Kings Laundry's diversion ratios in the healthcare market, 2014 - 30 J	lune
2018 [※ REDACTED]	115
Table 20: Healthcare respondents' views on the impact of the Proposed Transa	action
	122
Table 21: Profile of Customers that provided responses	158
Table 22: Profile of Self-suppliers that provided responses	159
Table 23: Examples of criteria used in selection of provider of flat linen rental a	and
maintenance services	
Table 24: Customers' views on the potential impact of the Proposed Transaction	on163
Table 25: Equipment used in OPLs	
Table 26: Self-suppliers' views on the potential impact of the Proposed Transa	ction 166
Table 27: Organisations with OPLs	
Table 28: Organisations switching from/to OPLs, 1 January 2014 to 30 June 20	18170
Table 29: Recently opened hospitals and care centres that purchase outsource	ed flat
linen rental and maintenance services	170
Table 30: Glossary of frequently used terms	171

Competition and Consumer Protection

Commission

2
5
8
8
9
9
4
4
e
8
8
9
9
9
9



# 1. INTRODUCTION

# Introduction

- 1.1 On 7 August 2018, in accordance with section 18(1)(a) of the Competition Act 2002, as amended ("the Act"), the Competition and Consumer Protection Commission (the "Commission") received notification of a proposed transaction whereby Berendsen Ireland Limited ("Berendsen") would acquire sole control of Kings Laundry Limited ("Kings Laundry") (the "Proposed Transaction"). Berendsen is ultimately controlled by Elis S.A. ("Elis").
- 1.2 The Proposed Transaction is to be implemented pursuant to a Sale and Purchase Agreement ("SPA") dated 25 July 2018 between Berendsen and Oranday Holdings Limited, Coconutcove Limited and Fenbrook Nominees Limited<sup>1</sup> (collectively the "Vendors" and each a "Vendor").<sup>2</sup> Pursuant to the SPA, Berendsen would acquire the full legal and beneficial ownership of the entire issued share capital of Kings Laundry.

# The Undertakings Involved

The Acquirer - Berendsen<sup>3</sup>

- 1.3 Berendsen is an indirect wholly-owned subsidiary of Elis. Elis is a public limited company listed on the Euronext Paris stock exchange. Elis specialises in the rental and maintenance<sup>4</sup> of flat linen,<sup>5</sup> professional garments (i.e., workwear) and hygiene and wellness equipment across Europe and Latin America.
- 1.4 Within the State, Elis carries out almost all its activities through Berendsen. Berendsen's activities can be summarised as follows:
  - rental and maintenance of workwear to customers in various sectors;<sup>6</sup>
  - rental and maintenance of flat linen to customers in the healthcare<sup>7</sup> and the hospitality<sup>8</sup> sectors;
  - rental and maintenance of mats;
  - rental and maintenance of mops;

<sup>3</sup> Elis acquired full ownership and control of Berendsen on 12 September 2017. The Berendsen business, including Berendsen Ireland Limited was rebranded as Elis on 25 June 2018. See <u>https://ie.elis.com/insights/news/ie/berendsen-continues-under-the-nameelis/.</u>

<sup>6</sup> For example, Berendsen provides workwear to customers in the construction and manufacturing sectors.

<sup>&</sup>lt;sup>1</sup> Fenbrook Nominees Limited is the legal owner of a shareholding in Kings Laundry of which the beneficial owner is [...].

<sup>&</sup>lt;sup>2</sup> These three Vendors are collectively the legal and beneficial owners of the entire issued share capital of Kings Laundry.

<sup>&</sup>lt;sup>4</sup> The term maintenance refers to the collection and processing (i.e., sorting, washing, drying, ironing, folding and packing) of used and soiled items and the delivery of clean items. Items may also be repaired as part of maintenance.

<sup>&</sup>lt;sup>5</sup> Flat linen is an industry term describing offered linen products including flat-ironed bed sheets, pillowcases and tablecloths.

<sup>&</sup>lt;sup>7</sup> Customers in this sector include acute hospitals, specialist hospitals, private hospitals, care homes and nursing homes.

<sup>&</sup>lt;sup>8</sup> Customers in this sector include hotels, ferries, hostels, restaurants, guesthouses, leisure centres/spas and golf clubs.



- rental and maintenance of cleanroom garments to organisations operating cleanrooms such as pharmaceutical and high-tech industries;<sup>9</sup> and
- provision of washroom services.<sup>10</sup>
- 1.5 Berendsen operates eight facilities in the State, [...] of which cater for the maintenance of flat linen.<sup>11</sup> Berendsen Northern Ireland Limited operates one facility in Northern Ireland.
- 1.6 For the financial year ending 31 December 2017, Elis had a worldwide turnover of approximately €2.21 billion, of which €[...] was generated in the State.<sup>12</sup>

#### The Target – Kings Laundry

- 1.7 Kings Laundry, a private limited company registered in the State, provides rental and maintenance of flat linen to customers in the healthcare and the hospitality sectors.<sup>13</sup>
- 1.8 Kings Laundry operates two facilities in the State catering for the maintenance of flat linen.<sup>14</sup>
- 1.9 For the financial year ending 31 December 2017, Kings Laundry had a worldwide turnover of approximately €[...], of which €[...] was generated in the State.<sup>15</sup>

# **Rationale for the Proposed Transaction**

1.10 The Parties<sup>16</sup> stated the following in the notification:

"The Proposed Transaction is an additional step in Elis' development in Ireland. It complements Elis' existing operations in Ireland and enables Elis to offer services to a wider range of customers in combination with a well-known player."

## Phase 1: Preliminary Investigation ("Phase 1")

#### **Contacts with the Parties**

1.11 On 14 September 2018, the Commission served a Requirement for Further Information ("First RFI") on Berendsen and Kings Laundry, pursuant to section 20(2) of the Act. This

<sup>&</sup>lt;sup>9</sup> These customers operate within environments controlled for particular contaminants, such as dust, airborne microbes, aerosol particles and chemical vapours (cleanrooms).

<sup>&</sup>lt;sup>10</sup> Berendsen provides products such as toilet paper, soap dispensers and hand sanitizer.

<sup>&</sup>lt;sup>11</sup> These [...] facilities are located at [...].

<sup>&</sup>lt;sup>12</sup> According to the parties, [...].

<sup>&</sup>lt;sup>13</sup> According to the parties, revenues from customers in the hospitality sector account for approximately [...]% of Kings Laundry's turnover.

<sup>&</sup>lt;sup>14</sup> These two facilities are located in South City Business Park, Tallaght, Dublin 24, D24 NV2T and Unit 25, GB Business Park, Little Island, Cork, Co. Cork T45 NW20.

<sup>&</sup>lt;sup>15</sup> The remaining turnover was generated in [...].

<sup>&</sup>lt;sup>16</sup> Throughout this Phase 2 Determination the Commission refers to Berendsen and Kings Laundry collectively as "the Parties".



adjusted the deadline within which the Commission was required to conclude its assessment of the Proposed Transaction in Phase 1.

- 1.12 Upon receipt of all of the responses to the RFI, the "appropriate date" (within the meaning of section 19(6)(b)(i) of the Act) became 30 November 2018.<sup>17</sup>
- 1.13 During the Phase 1 investigation, the Commission requested and received, on an ongoing basis, further information and clarifications from the Parties.

## Third Party Submissions

- 1.14 No third party submission was received by the Commission during the Phase 1 investigation.
- 1.15 During the Phase 1 investigation, the Commission circulated questionnaires<sup>18</sup> and carried out follow-up communication with various third parties, including:
  - a number of competitors of Berendsen and Kings Laundry that are or were active in the supply of rental and maintenance of flat linen in the State and Northern Ireland as identified by the Parties in the Merger Notification Form<sup>19</sup> ("Competitors");<sup>20</sup>
  - a number of customers in the healthcare and hospitality sectors that purchase rental and maintenance of flat linen services ("Customers").<sup>21</sup> The Parties identified these customers as their largest customers in terms of turnover;<sup>22</sup>
  - a number of hotels currently operating their own on-premise laundries ("OPLs") as identified by the Parties in the Merger Notification Form ("Self-suppliers");<sup>23</sup> and
  - A number of hospitals currently operating their own on-premise laundries ("OPLs") as identified by the Parties in the Merger Notification Form ("Selfsuppliers").<sup>24</sup>
- 1.16 The Commission received a response from the majority of the third parties to whom it sent a questionnaire (81%<sup>25</sup>). In each case, the Commission also contacted those third parties by telephone and/or e-mail to clarify and/or seek further detail in relation to their responses. Customers' and Self-suppliers' responses are summarised in Appendix 1 of this Phase 2 Determination.

 <sup>&</sup>lt;sup>17</sup> The "appropriate date" is the date from which the time limits for making both Phase 1 and Phase 2 Determinations begin to run.
 <sup>18</sup> Through the remainder of this Phase 2 Determination, these questionnaires will be referred to as "Questionnaire to Competitors", "Questionnaire to Customers" and "Questionnaire to Self-suppliers".

<sup>&</sup>lt;sup>19</sup> Merger Notification Form refers to the merger notification form relating to the Proposed Transaction, as was submitted to the Commission on 7 August 2018.

<sup>&</sup>lt;sup>20</sup> These are: [...].

<sup>&</sup>lt;sup>21</sup> These are: [...].

<sup>&</sup>lt;sup>22</sup> The Customers which responded to the questionnaire account for approximately [...]% and [...]% of Kings Laundry's turnover and [...]% and [...]% of Berendsen's turnover, in healthcare and hospitality sectors respectively.

<sup>&</sup>lt;sup>23</sup> These are: [...]. Moreover, the Commission contacted [...] seeking to visit its OPL.

<sup>&</sup>lt;sup>24</sup> These are: [...]. Moreover, the Commission contacted [...] seeking to visit its OPL.

<sup>&</sup>lt;sup>25</sup> 34 out of 42 third parties.



# **The Phase 1 Determination**

- 1.17 Having considered all the available information in its possession at the time, the Commission was unable to form the view at the conclusion of the Phase 1 investigation that the result of the Proposed Transaction would not be to substantially lessen competition in any market for goods or services in the State.
- 1.18 On 9 January 2019, the Commission determined, in accordance with section 21(2)(b) of the Act, to carry out a full investigation under section 22 of the Act in relation to the Proposed Transaction.

# Phase 2: Full Investigation ("Phase 2")

### **Contacts with the Parties**

- 1.19 During the Phase 2 investigation, the Commission requested and received further information and clarifications from the Parties.
- 1.20 On 18 January 2019, the Parties submitted two reports relating to the competitive constraints posed by: (i) OPLs operated by healthcare and hospitality organisations (the "Report on OPLs"); and (ii) Celtic Linen (the "Report on Celtic Linen").
- 1.21 The Commission conducted site visits to facilities of both Berendsen and Kings Laundry. The Commission visited Kings Laundry's Cork facilities on 13 February 2019 and its Dublin facilities on 6 March 2019. The Commission visited Berendsen's Dublin facilities on 14 February 2019.
- 1.22 On 12 February 2019, the Commission served a Requirement for Further Information ("Second RFI") on Berendsen and Kings Laundry, pursuant to section 20(2) of the Act. Berendsen fully responded to the RFI on 27 February 2019 and Kings Laundry fully responded on 8 March 2019. This adjusted the deadline within which the Commission was required to conclude its determination of the Proposed Transaction in Phase 2.
- 1.23 The Commission issued its Assessment to the Parties on 28 March 2019 in accordance with its Mergers and Acquisitions Procedures.<sup>26</sup>
- 1.24 The Parties requested access to the Commission's file.<sup>27</sup> According to the Commission's procedures,<sup>28</sup> the Parties are to be given access to the Commission's file upon request, during the 15 working day period following their receipt of the Commission's Assessment. The Commission provided the Parties with a schedule of all of the documents included in the file on 1 April 2019 and access to the file was granted during the 15 working day period.
- 1.25 During the Commission's access to the file process, the Parties identified several documents of interest to them to which they had been granted partial access and to which they requested full access. Information that was confidential to third parties had

<sup>&</sup>lt;sup>26</sup> See the Commission's document titled "Mergers and Acquisitions Procedures", dated 31 October 2014 and in particular paragraph 3.8.

<sup>&</sup>lt;sup>27</sup> Berendsen made request on 29 March 2019, and Kings Laundry made request on 3 April 2019.

<sup>&</sup>lt;sup>28</sup> Article 5.1, of the Commission's document titled "Access to the File in Merger Cases", dated 31 October 2014



been redacted in the version of the documents provided to the Parties and could not be released without the consent of the relevant third party. However, in order to ensure that the Parties had access to all the information they required to understand and address the concerns raised in the Commission's Assessment, the Commission issued non-confidential summaries of these confidential materials to the Parties. The Commission also provided non-confidential summaries of any information on which it had relied in the Assessment and any information that could be exculpatory in the Parties' defence. In parallel, the Commission sought the consent from the relevant third parties to disclose their confidential information to advisors of the Parties under certain terms. Confidential information was disclosed where written consent was given by a third party.

- 1.26 The Parties made a joint written submission on 25 April 2019 in response to the Commission's Assessment (the "Parties' Response"). The Parties' submission included a submission by Frontier Economics (the "Frontier Economics Report").
- 1.27 The Parties, having regard to confidentiality requirements, made a joint oral submission to Commission Members on 3 May 2019 at which they and their legal and economic advisors participated in the oral submission.
- 1.28 Following the oral submissions on 3 May 2019, the Commission made additional market enquiries to address several points raised by the Parties in their written and oral submissions. The Parties were given access to all additional information gathered in this regard. The Parties were also given the opportunity to make further submissions, which they did in writing on 28 May 2019 and 4 June 2019, and orally on 29 May 2019.<sup>29</sup>
- 1.29 During the Phase 2 investigation, the Commission held various meetings and conference calls with the advisors of the Parties to discuss identified competition concerns and how they would be addressed.

## Phase 2 Proposals

- 1.30 The Proposed Transaction concerns the acquisition of the whole of Kings Laundry's business: that is, the supply of outsourced flat linen rental and maintenance services to both the hospitality and the healthcare markets. On completion of the Phase 2 review, the Commission has found that the Proposed Transaction will likely result in an SLC only in the healthcare market.
- 1.31 Following several engagements with the parties to discuss the identified SLC concerns in the healthcare market, Berendsen, on a without prejudice basis, submitted proposals on 14 June 2019 (the "First Proposals") pursuant to Section 20(3) of the Act. The submission of these proposals by Berendsen extended the deadline within which the Commission was required to make its determination of the Proposed Transaction in Phase 2. The extension added 15 working days to the Phase 2 period, bringing the review period to a total of 135 working days, in accordance with section 22(4B) of the Act. In the First Proposals Berendsen committed to either behavioural commitments, or to the divestment of certain contracts.

<sup>&</sup>lt;sup>29</sup> Further submissions are referenced in the Determination as "Parties' Letter" and the appropriate date.



- 1.32 The Commission rejected the First Proposals on the basis that they were inappropriate and insufficient to address the identified SLC concerns in the healthcare market. In light of this, the Commission did not find it necessary to market test the First Proposals.
- 1.33 On 24 June 2019 Berendsen, on a without prejudice basis, submitted revised proposals (the "Second Proposals") to the Commission committing to dispose of a package of healthcare contracts, [...].
- 1.34 The Commission considered that there were implementation risks associated with the Second Proposals which would render the remedy ineffective if they materialised. However, subject to the Parties addressing those implementation risks, the Second Proposals provided a divestment package which the Commission could market test.
- 1.35 Over the period 25 June to 1 July 2019, the Commission market tested the Second Proposals in order to establish whether they were likely to be appropriate, proportionate and effective in addressing the identified SLC concerns in the healthcare market. A summary of the results of the market testing is provided in paragraphs 5.8 to 5.13.
- 1.36 While the Commission considered that the substance of the Second Proposals may have addressed its competition concerns, it had significant remaining concerns regarding the implementation of the measures. These were post-completion proposals that posed serious implementation risks. The Commission engaged extensively with the Parties in relation to these concerns. Ultimately, however, the Parties were unable to address the implementation risks associated with the Second Proposals.
- 1.37 On 8 July 2019, Berendsen submitted a revised set of proposals (the "Third Proposals"). The Commission considers that the Third Proposals submitted by the Parties address its SLC concerns in the healthcare market (the "Proposals"). The Proposals are included as Chapter 9 in this Determination. The divestment of a package of healthcare contracts as described in the Proposals will allow the approved third party purchaser to provide effective competition to the merged entity and Celtic Linen. Berendsen's commitment to refrain from completing the Proposed Transaction until it has completed the divestment of the healthcare contracts addresses the Commission's concerns regarding implementation risks.
- 1.38 The Commission has taken the Proposals into account and, in light of the Proposals (which form part of the basis of the Commission's determination) has determined, in accordance with section 21(2)(a) of the Act, that the result of the proposed acquisition whereby Berendsen would acquire sole control of Kings Laundry will not be to substantially lessen competition in any market for goods or services in the State, and, accordingly, that the acquisition may be put into effect.

## **Contacts with Competitors**

1.39 During the Phase 2 investigation, the Commission extended the investigative work undertaken during Phase 1, and carried out a site visit to [...] facilities on 21 February



2019 and a site visit to [...] facilities on 8 May 2019. The Commission also held further phone calls with a number of Competitors:  $[...]^{30}$ ,  $[...]^{31}$ ,  $[...]^{32}$ ,  $[...]^{33}$  and  $[...]^{34}$ .

#### Contacts with customers and other third parties

- 1.40 During the Phase 2 investigation, the Commission had further contact with third parties, including customers in the healthcare and hospitality sectors, the Office of Government Procurement (the "OGP"), suppliers of laundry equipment,<sup>35</sup> and the National Standards Authority of Ireland (the "NSAI"). The Commission also carried out a site visit to a hotel which operates an OPL.<sup>36</sup>
- 1.41 The Commission commissioned Amárach Research ("Amárach") to conduct a survey of customers in both the hospitality and healthcare sectors (the "Amárach Survey") to inform its investigation, and to supplement the Market Enquiries undertaken directly by the Commission. Amárach contacted randomly selected organisations operating in the healthcare and hospitality sectors. Amárach Survey's sample excluded organisations already contacted directly by the Commission in the course of its own Market Enquiries. As noted in paragraph 1.15 above, the organisations contacted directly by the Commission included those identified by the Parties in the Merger Notification Form as being their top customers. The Commission therefore attaches particular weight throughout its analysis to the views of these more significant customers.
- 1.42 Amárach completed a total of 182 interviews. 67% of respondents were from the hospitality sector and the remaining 33% of respondents were from the healthcare sector.
- 1.43 The Amárach Survey is attached to this Phase 2 Determination in Appendix 2.
- 1.44 The Commission sought economic advice from PMCA Economic Consulting with regards to aspects of market definition and the likely impact of the Proposed Transaction on any market for goods or services in the State.
- 1.45 Following receipt of the Parties' written and oral submissions to its Assessment, the Commission sought specialist economic advice from DotEcon Ltd. ("DotEcon"). The DotEcon economic advice addressed specific arguments raised by the Parties.<sup>37</sup>

## **Third Party Submissions**

1.46 No third party submission was received by the Commission during the Phase 2 investigation.

<sup>&</sup>lt;sup>30</sup> Phone call was held on 20 February 2019.

<sup>&</sup>lt;sup>31</sup> Phone call was held on 26 February 2019.

<sup>&</sup>lt;sup>32</sup> Phone call was held on 19 February 2019.

<sup>&</sup>lt;sup>33</sup> Phone call was held on 19 February 2019.

<sup>&</sup>lt;sup>34</sup> Phone call was held on 20 February 2019.

<sup>&</sup>lt;sup>35</sup> [...].

<sup>&</sup>lt;sup>36</sup> [...].

<sup>&</sup>lt;sup>37</sup> For information purposes only, the DotEcon report (the "DotEcon Report") is presented alongside this Phase 2 Determination in Appendix 3. The views expressed by DotEcon are not necessarily the views of the Commission.



# 2. INDUSTRY BACKGROUND – FLAT LINEN RENTAL AND MAINTENANCE

2.1 Both Berendsen and Kings Laundry supply outsourced flat linen rental and maintenance services. This section provides an industry overview and describes some of the recent trends in the rental and maintenance of flat linen in the State.

# **Industry Overview**

- 2.2 The demand for outsourced flat linen rental and maintenance services<sup>38</sup> arises primarily in the hospitality and healthcare sectors. Customers in these sectors require a wide range of linen items including, but not limited to: bed linen; table linen; towels; and bath robes. The Parties indicated that, in addition to linen items, customers in the healthcare sector require rental and maintenance of surgical scrubs and maintenance of certain miscellaneous items.<sup>39</sup> Customers in both sectors typically specify item type, size, colour and provisional annual volume of required linen items. Certain hospitality customers (e.g., 5 star hotels) also specify linen quality (in terms of thread count or GSM<sup>40</sup>) and may also need their logo embroidered on linen items. Finally, customers specify timing and frequency of flat linen pick-up and delivery as well as some key performance indicators such as the maximum acceptable level of rejected linen.<sup>41</sup>
- 2.3 The linen maintenance process is similar across both healthcare and hospitality sectors with the same techniques and equipment used to wash, dry and iron soiled linen returned from customers in both sectors.<sup>42</sup> Thus, linen laundries catering for customers in both sectors tend to process linen in the same facility. However, linen items in the healthcare sector can get infected or contaminated with infectious diseases. For this reason, healthcare customers require linen laundries to adhere to certain linen processing guidelines.<sup>43</sup> In summary, linen laundries serving healthcare customers are required to:
  - a) Deploy a barrier wall system in order to prevent cross-contamination;
  - b) Wash linen items at specified temperature for a specified period of time;
  - c) Inject staff involved in the process with hepatitis B injections;
  - d) Give protective clothing and gloves to laundries' staff dealing with linen items from healthcare customers;
  - e) Maintain a higher level of cleanliness at a linen laundry facility compared to a laundry facility that deals only with hospitality linen; and

<sup>&</sup>lt;sup>38</sup> Hereinafter referred to as "flat linen rental and maintenance services".

<sup>&</sup>lt;sup>39</sup> Such items include, for example, patient gowns and slings.

<sup>&</sup>lt;sup>40</sup> In short, GSM refers to the fabric weight and density of towels in grams-per-square-meter. A towel with a higher GSM measurement is thicker and plusher than one with a low GSM, meaning it can absorb more water due to the higher pile and is usually made from a more luxurious type of cotton.

<sup>&</sup>lt;sup>41</sup> Customers may reject delivered linen because it is either (i) not clean or (ii) damaged (e.g., torn).

 <sup>&</sup>lt;sup>42</sup> It should be noted that the equipment used will differ depending on the type of linen (e.g., whether it is bed or table linen).
 <sup>43</sup> For example, see the HSE's guidelines for managing laundry and linen in healthcare facilities which can be accessed at:

https://www.hse.ie/eng/about/who/healthwellbeing/infectcont/sth/gl/ipcc-guidelines-section-9.pdf.



- f) Maintain a higher level of cleanliness in trolleys and vehicles carrying linen items.
- 2.4 The Parties indicated that contaminated linen arrives in alginate bags and is put through an initial chemo-thermal wash to prevent the spread of microorganisms before joining all other soiled linen. Figure 1 illustrates the flat linen maintenance process for healthcare customers.





N.B. The dry/press/finish steps vary depending on the type of item being laundered.

#### Source: information provided by the Parties

- 2.5 Linen laundries typically operate one or more facilities with a fleet of heavy goods vehicles ("HGVs") and vans delivering/collecting clean/soiled linen from multiple customer sites each day. In this supply model, linen laundries tend to earn higher margins if they are able to fill their vehicles by:
  - a) selling multiple services to their customers (e.g., flat linen and dust mats rental and maintenance services);
  - b) having a high density of customers in their areas of operation;
  - c) having facilities in close proximity to their customers; and/or
  - d) selling services to large customers.



- 2.6 In terms of pricing, the Commission's review of the Parties' contracts indicated that the Parties charge customers [...].<sup>44</sup> Prices are usually set out in contracts and are [...] during the duration of the contract. It should be noted that the largest customers of linen laundries frequently avail of [...].
- 2.7 Services provided by linen laundries can be grouped into three broad categories:
  - a) maintenance only where linen items are owned by the customer, but maintained by the linen laundry;
  - b) rental and maintenance where linen items are provided (and, where necessary replaced) by and maintained by the linen laundry;
  - c) managed service where the linen laundry not only rents and maintains linen, but also runs the customer's linen stock room and the distribution of linen throughout the customer's facilities such as a hospital or hotel.
- 2.8 Finally, a distinction can be made in terms of how linen laundries manage their linen stock:
  - a) pooled linen system where linen items are shared amongst customers (i.e., a customer may receive clean linen items previously used by another customer); and
  - b) dedicated linen system where linen items are dedicated to a single customer (i.e., a customer receives the same linen items that were sent to the linen laundry for washing).

## Nature of competition

- 2.9 The Parties submitted that customers in healthcare and hospitality sectors can select their providers of flat linen rental and maintenance services by: (i) approaching them directly and asking for a quote; or (ii) issuing a formal tender where providers are selected based on specific requirements.<sup>45</sup> Competition between linen laundries therefore takes place when a customer is considering its contract options.
- 2.10 The Parties submitted that customers in the hospitality sector use both approaches when selecting linen laundries, while the majority of healthcare customers use formal tender procedures. The Commission's market investigation broadly supported the Parties' views. For example, the Amárach Survey indicated that 23% of the respondents from the healthcare sector issue formal tenders with 19% requesting quotes with just over a third 35% saying that they extend, rollover or re-contract with their existing provider. In comparison, only 16% of respondents from the hospitality sector stated that they issue formal tenders whereas just over half, 51%, requested quotes and 17% said that they extended, rolled over or re-contracted with their existing provider.<sup>46</sup> A

<sup>&</sup>lt;sup>44</sup> This was also confirmed by Customers interviewed directly by the Commission. See paragraph 10.11 below.

<sup>&</sup>lt;sup>45</sup> See page 14 of the Merger Notification Form.

<sup>&</sup>lt;sup>46</sup> See slide 32 of the Amárach Survey.

similar pattern was observed during the Commission's interviews with significant Customers carried out in its direct Market Enquiries.<sup>47</sup>

- 2.11 The Commission notes that direct negotiations between laundries and customers include the roll-over of existing agreements. Therefore, the degree to which there is competition for customers using a direct negotiation approach will depend on the extent to which such customers proactively seek out bidders and invite them to make an offer, or potential providers proactively contact potential customers (i.e., make cold calls).
- 2.12 The Commission notes that procurement for some (but not all) public hospitals is currently carried out by the HSE. The OGP is developing a Framework Agreement for outsourced flat linen rental and maintenance services to be supplied to public bodies (the "OGP's Framework").<sup>48</sup> On 3 June 2019, the OGP issued a Request for Tender ("RFT") for the establishment of the OGP's Framework.<sup>49</sup> The duration of the OGPs Framework will be 2 years with a possibility of 2 one-year extensions. The OGP's Framework is divided into 6 lots<sup>50</sup> and is expected to be operational in Q3 2019.<sup>51</sup> In common with other procurement frameworks, the OGP's RFT invites bidders to tender for admission to the OGP's Framework, with tenderers required to meet a set of qualification criteria such as experience and financial and economic standing. Subsequent mini-tenders will be held for specific contracts and only suppliers admitted to the OGP's Framework will be able to compete for these contracts. Admission to the framework does not guarantee any contract award. Although there is no mandatory requirement for public bodies to use the OGP's Framework, its establishment would, to a large extent, replace the HSE's tender practices.

# Increasing demand for outsourced flat linen rental and maintenance services

2.13 The Commission notes that generally the demand for outsourced flat linen rental and maintenance services has been continuously growing in recent years. This is demonstrated in Figure 2 which indicates that combined revenue for outsourced flat linen rental and maintenance services increased from €[...] in 2014 to €[...] in 2017 (an increase of 20%) with a Compound Annual Growth Rate of 6.2% during this period.

<sup>48</sup> The Framework Contract will be available for use by all public sector bodies seeking to purchase linen and laundry services.

<sup>&</sup>lt;sup>47</sup> See paragraph 10.14 below.

<sup>&</sup>lt;sup>49</sup> See <u>https://irl.eu-supply.com/app/rfq/publicpurchase\_frameset.asp?PID=148064&B=ETENDERS\_SIMPLE&PS=1&PP=.</u>

<sup>&</sup>lt;sup>50</sup> Four of these lots cover the provision of outsourced flat linen rental and maintenance services.
<sup>51</sup> See a note of phone call with the OGP dated 14 May 2019



### Figure 2: Revenue from the provision of flat linen rental and maintenance services, 2014 – 2017 [*X* REDACTED]

Source: The Commission's analysis using information provided by the Parties and the Competitors.

- 2.14 The majority of this growth is accounted for by the increased demand from customers in the hospitality sector due to growing level of domestic<sup>52</sup> and overseas tourism<sup>53</sup> in recent years. For example, a recent survey of members of Irish Hotels Federation<sup>54</sup> (the "IHF survey") indicates that 70% of survey respondents reported business growth when comparing the summer of 2018 with the summer of 2017. An increase in summer advance bookings was also reported by the majority of respondents (67%).
- 2.15 Demand for flat linen rental and maintenance services from customers in the healthcare sector is also growing (albeit modestly) with internal Berendsen's documents noting [...].<sup>55</sup>
- 2.16 In general, the Commission has observed that demand for flat linen rental and maintenance services tends to grow in line with Gross Domestic Product ("GDP"). For example, Berendsen told the Commission that [...].<sup>56</sup> In addition, Berendsen confirmed that the flat linen rental and maintenance sector is growing in line with growing tourism and the economy.<sup>57</sup> Kings Laundry told the Commission that its turnover grew during the recession as it acquired customers from competitors [...].<sup>58</sup>

15

<sup>&</sup>lt;sup>52</sup> For example, the IHF survey indicates that 60% of surveyed hotels and guesthouses reported an increase in domestic visitor numbers. See also Fáilte Ireland's Tourism Barometer - December 2018 which can be accessed at: http://www.failteireland.ie/Research-and-Insights.aspx.

<sup>&</sup>lt;sup>53</sup> For example, the Central Statistics Office (the "CSO") reports that overseas trips in Q2 2018 increased by 6.6% when compared with the same period in 2017. Growth in the hospitality sector is also highlighted in Irish Times Article which can be accessed at: https://www.irishtimes.com/business/commercial-property/ireland-reaches-820-hotels-in-bumper-year-for-sector-1.3316496

<sup>&</sup>lt;sup>54</sup> The report can be accessed at: <u>https://www.ihf.ie/sites/default/files/upload/june\_barometer.pdf</u>.

<sup>&</sup>lt;sup>55</sup> For example, see slide 4 of Berendsen's document titled [...] submitted by Berendsen in response to Question 9 of the First RFI. Also see the HSE's page 69 of the National Service Plan 2019 in which it is noted that acute bed capacity will increase by 277 beds by Q1, 2020 (2%).

<sup>&</sup>lt;sup>56</sup> The Commission's meeting with Berendsen on 30 January 2019.

<sup>&</sup>lt;sup>57</sup> The Commission's meeting with Berendsen on 30 January 2019.

<sup>&</sup>lt;sup>58</sup> The Commission's meeting with Kings Laundry on 2 February 2019.

# 3. RELEVANT PRODUCT AND GEOGRAPHIC MARKETS

# **Horizontal Overlap**

Coimisiún um Iomaíocht agus

Cosaint Tomhaltóirí

3.1 There is a horizontal overlap between Berendsen and Kings Laundry in the State with respect to the supply of outsourced flat linen rental and maintenance services to customers in: (i) the healthcare sector; and (ii) the hospitality sector.

## **Relevant Product and Geographic Markets**

#### Views of the Parties

#### Relevant product markets

- 3.2 The Parties took the view that the supply of flat linen rental and maintenance services<sup>59</sup> to: (i) healthcare; and (ii) hospitality sectors form distinct product markets. In this regard, the Parties noted that: *"The customer bases of the two segments are clearly distinct, and the nature of the products purchased by and the requirements of customers in each of the two segments are very different."*<sup>60</sup> At the same time, the Parties noted that *"...from a supply-side perspective, there are limited technical barriers to a business active in one of these segments beginning to supply customers in another."*<sup>61</sup>
- 3.3 The Parties also highlighted material differences in conditions of competition between the healthcare and hospitality segments<sup>62</sup> and were of the view that the approach taken by the UK's Office of Fair Trading ("OFT") in Johnson Apparelmaster Ltd / Cannon Textile Care Business<sup>63</sup> should be adopted.
- 3.4 The Parties proposed that the supply of flat linen rental and maintenance services to healthcare customers constitutes a single product market, but highlighted that within this market there is a wide range of different types of customers.<sup>64</sup> The Parties then noted that *"The level of overlap between the Parties in servicing these customer groups differs"*<sup>65</sup>, as Kings Laundry does not provide flat linen rental and maintenance services to public healthcare customers. While the Parties were of the view that this fact was not adequately taken into account by the Commission, they confirmed their position that the supply of flat linen rental and maintenance services to healthcare customers constitutes a single product market.<sup>66</sup>

<sup>&</sup>lt;sup>59</sup> The Commission notes that the Parties were of the view that rental and maintenance of surgical scrubs and maintenance of certain miscellaneous items should be included in the relevant market for the provision of flat linen rental and maintenance services to the healthcare sector. The Parties noted that surgical scrubs and other ancillary items typically form part of a broader package of contracted services that includes flat linen. Footnote 30 of the Merger Notification Form.

<sup>&</sup>lt;sup>60</sup> Page 17 of the Merger Notification Form.

<sup>&</sup>lt;sup>61</sup> Page 17 of the Merger Notification Form.

<sup>&</sup>lt;sup>62</sup> Page 18 of the Merger Notification Form.

<sup>&</sup>lt;sup>63</sup> See OFT's decision of 29 March 2012 titled <u>"Anticipated acquisition by Johnsons Apparelmaster Limited of the Cannon Textile Care Business"</u>, case number ME/5325/12 ("Johnsons Apparelmaster/Cannon").

<sup>&</sup>lt;sup>64</sup> Frontier Economics Report, paragraph 2.6.

<sup>&</sup>lt;sup>65</sup> Frontier Economics Report, paragraph 2.7.

<sup>&</sup>lt;sup>66</sup> Frontier Economics Report, paragraph 2.8.



- 3.5 Additionally, the Parties were of the view that self-supply of flat linen and its maintenance via OPLs should be included in the relevant product markets.<sup>67</sup> In this regard, the Parties proposed that the OFT's reasoning in excluding self-supply in *Johnsons Apparelmaster/Cannon* does not apply in this case because:
  - a) there was a range of hospitals and hotels operating their own OPLs;
  - b) there was evidence of customers switching from purchasing outsourced flat linen rental and maintenance services to self-supply;
  - c) there were a large number of suppliers of industrial/commercial laundry equipment in the State, thus demonstrating a demand for such equipment from both healthcare and hospitality customers;
  - d) customers used the threat of switching to OPL in order to achieve better prices and better customer service;
  - e) the costs associated with setting up OPL facilities were relatively low and would account for a small share of a customers' revenue;
  - f) some customers already had OPL facilities that are used for maintenance of certain items (e.g., towels). For such customers, the costs of maintenance for more items in-house would be lower than for a customer without an OPL facility; and
  - g) OPLs were useful for hotels that have seasonal demand as it allowed them to retain key staff throughout the year.
- 3.6 The Parties in their Report on OPLs reiterated some of the arguments presented in the Merger Notification Form, but also set out additional arguments:
  - a) there was a precedent of national competition authorities ("NCAs") assessing OPL either as part their assessment of market definition or as part of their assessment of customers' countervailing buyer power ("CBP");<sup>68</sup>
  - b) OPLs were referred to as one of the main competitors in Elis' financial statements;<sup>69</sup> and
  - c) additional reasons why the OFT's reasoning in excluding self-supply in *Johnsons Apparelmaster/Cannon* did not apply in this case.
- 3.7 In their written Response, the Parties were of the view that the Commission: (i) underestimated the level of switching to OPLs; and (ii) failed to apply the hypothetical monopolist tests correctly.<sup>70</sup> In this regard, the Parties noted that OPLs posed a substantially higher constraint on Kings Laundry than Berendsen during the analysed

<sup>&</sup>lt;sup>67</sup> Pages 18 and 19 of the Merger Notification Form.

<sup>&</sup>lt;sup>68</sup> In addition to Johnsons Apparelmaster/Cannon, the Parties referred to case CA/1135/92 – Spring Grove / Conkenner and case ME/3633/08 - Fishers Services Ltd / The Sunlight Service Group.

<sup>&</sup>lt;sup>69</sup> These statements are available at: <u>https://www.corporate-elis.com/en/investor-relations</u>.

<sup>&</sup>lt;sup>70</sup> Frontier Economics Report, paragraphs 2.2 and 2.10.



period.<sup>71</sup> The Parties claimed that the evidence and analysis does not support the exclusion of OPLs from the relevant market. The Parties noted that, for hospitality customers, switching from Kings Laundry to OPL is more prevalent than switching from Kings Laundry to Berendsen. The Parties believed that there are errors in the application of the hypothetical monopolist (SSNIP) test, and in the use of survey data.<sup>72</sup>

3.8 The Parties also stated that customers could opt to use launderettes providing a washonly service.73

#### Relevant geographic markets

Coimisiún um Iomaíocht agus

- 3.9 In relation to the geographic scope of the relevant market(s), the Parties considered that it should be at least State-wide, referring to European Commission's (the "EC's") decision in Case No M.8399 - CWS-BOCO / RENTOKIL INITIAL TARGET BUSINESSES.<sup>74</sup> In this regard, the Parties noted that: "Whilst the nature of the products means that the costs of serving a customer are likely to be higher the farther they are from the plant, the relatively small size of the State, especially relative to other EU Member States, such as the United Kingdom, means that the costs of transporting linen even over a substantial distance are not prohibitive."75
- Furthermore, the Parties noted that:<sup>76</sup> 3.10
  - a) their competitors are generally active across the State;
  - b) suppliers that originally operated at a local or regional level have expanded activities into other regions;
  - c) the map of customers submitted by the Parties indicates that some of the Parties' customers are located at substantial distances from the nearest Parties' facilities catering for flat linen; and
  - d) there is a prevalence in national tenders by hotel groups to award contracts to suppliers of flat linen rental and maintenance services on a national basis.
- 3.11 In this regard, the Parties argued that in its Assessment, the Commission understated the role of regional providers<sup>77</sup> noting that "...laundries can have a national presence irrespective of the number of sites they have."78
- 3.12 Finally, while not explicitly stating that the geographic market should be broadened to include Northern Ireland, the Parties noted that linen laundries operating in Northern Ireland such as Lilliput and Limavady Linen, serve customers in the State from their facilities in Northern Ireland and are competing with the Parties. Thus, the Parties

<sup>&</sup>lt;sup>71</sup> Frontier Economics Report, paragraph 2.10.a.

<sup>&</sup>lt;sup>72</sup> Frontier Economics Report, paragraphs 2.9 to 2.11.

<sup>&</sup>lt;sup>73</sup> Page 19 of the Merger Notification Form.

<sup>&</sup>lt;sup>74</sup> See the EC's decision in <u>Case No M.8399 - CWS-BOCO / RENTOKIL INITIAL TARGET BUSINESSES</u> ("CWS-boco/Rentokil").

<sup>&</sup>lt;sup>75</sup> Page 19 of the Merger Notification Form.

<sup>&</sup>lt;sup>76</sup> Page 19-20 of the Merger Notification Form.

<sup>&</sup>lt;sup>77</sup> Frontier Economics Report, paragraphs 2.14 to 2.17.

<sup>&</sup>lt;sup>78</sup> Frontier Economics Report, paragraph 2.16.



considered that constraint from these competitors should be taken into account in the Commission's competitive assessment.<sup>79</sup>

#### Views of third parties

3.13 The Competitors' and Customers' views largely confirmed the differences between the requirements of customers in the healthcare and hospitality sectors in terms of: (i) the type of linen items; and (ii) hygiene standard specification. For example, [...] stated that:

"Hospitality customers have no hygiene requirements. Some private healthcare customers require a hygiene certification such as BS EN 14065, some have no hygiene certification requirements. Large public healthcare customers require hygiene certification such as BS EN 14065."<sup>80 81</sup>

- 3.14 Customers from the healthcare sector confirmed that a linen laundry's ability to demonstrate that it has processes in place to prevent cross-contamination from soiled linen is an essential requirement. In this regard, the Amárach Survey indicated that 80% of respondents considered that a linen laundry's ability to adhere to procedures required by an organisation (e.g., bio contamination control or RABC) is an essential or very important factor when choosing suppliers of outsourced flat linen rental and maintenance services.<sup>82</sup>
- 3.15 The Competitors were of the view that it is relatively straightforward to switch the supply of outsourced flat linen rental and maintenance services between the two sectors. All six of the Competitors that provided views on this issue noted limited barriers to switching the supply of outsourced flat linen rental and maintenance services from the hospitality sector to the healthcare sector and *vice versa*. The main barriers cited to supply-side substitution were associated with the switch of supply from hospitality to healthcare customers:
  - a) deployment of a barrier wall system in order to prevent cross-contamination;
  - b) physical changeover of flat linen stock;
  - c) more administrative work in the healthcare sector associated with the preparation of bids in response to ITTs; and
  - d) adhering to higher hygiene standards when providing flat linen rental and maintenance services to customers in the healthcare sector.
- 3.16 In general, the Competitors opined that none of these financial barriers would prevent a swift and cost-effective switch of supply from one sector to another. This is particularly the case for a provider switching supply from the healthcare sector to the hospitality sector, because in that case a provider of outsourced flat linen rental and maintenance services would only be required to change linen stock. In contrast, switching supply from the hospitality sector to the healthcare sector would require not only a changeover of

<sup>&</sup>lt;sup>79</sup> Pages 20 to 21 of the Merger Notification Form.

 $<sup>^{\</sup>rm 80}\left[...\right]$  reply to question 20 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>81</sup> Note that BS EN 14065 is the internationally-recognised standard for certifying biocontamination control systems for the laundry processing of textiles.

<sup>&</sup>lt;sup>82</sup> Slide 36 of the Amárach Survey.

Coimisiún um Iomaíocht agus

Cosaint Tomhaltóirí

flat linen stock, but also additional investment in terms of deployment of a barrier wall system, administration and obtaining and adhering to higher hygiene standards.

- 3.17 In relation to OPLs, Customers that responded to the Questionnaire to Customers unanimously considered that setting up OPLs would not be a viable option in response to a 5% to 10% price increase<sup>83</sup> by their supplier of outsourced flat linen rental and maintenance services.<sup>84</sup> The main reasons provided can be summarised as follows:
  - a) customer sites were unsuitable to host OPL facilities due to a lack of space required for such a facility;
  - b) substantial costs associated with switching to self-supply including (but not limited to): purchasing linen stock, equipment required for maintaining linen, OPL staff and other operational costs (e.g., electricity); and
  - c) lack of expertise in operating a laundry.
- 3.18 Even Customers that currently do some of their laundry in-house (e.g., some hotel groups) noted that their OPLs do not have the capacity to accommodate all of their flat linen maintenance requirements. For example, [...] noted that the costs of setting up OPL facilities (in addition to OPLs currently servicing two of its hotels) in order to cater for the whole group's needs would significantly exceed the hypothetical 5%-10% increase in annual expenditure on outsourced flat linen rental and maintenance services.<sup>85</sup>
- 3.19 The Commission received mixed evidence from Competitors on the issue of OPLs. Five out of the eight Competitors that provided views on this issue indicated that switching to OPLs would not be a viable alternative for their customers purchasing outsourced flat linen rental and maintenance services. These competitors cited costs associated with setting up an OPL and lack of available staff as the main reasons for their views. [...] noted that in its experience, the opposite has occurred (i.e., customers switch from OPLs to outsourced flat linen rental and maintenance services).<sup>86</sup> Three out of eight Competitors that considered a switch to OPLs to be a viable option cited examples of hospitality customers switching to OPLs.
- 3.20 Suppliers of laundry equipment were of the view that setting up an OPL is a viable option for customers in both healthcare (in particular for nursing homes) and the hospitality sectors. Suppliers of laundry equipment stated that they have observed an increased demand for laundry equipment from self-supplying organisations in both sectors. These views were expressed in respect of organisations setting up OPLs in existing facilities (e.g., [...]), as well as organisations setting up OPLs in newly built facilities.
- 3.21 One supplier of laundry equipment<sup>87</sup> noted that organisations can switch to OPL as long as they have sufficient space. Thus, hotels in certain urban areas (e.g., Dublin city

<sup>&</sup>lt;sup>83</sup> This is a typical price range used in the small but significant non-transitory increase in price ("SSNIP") test. The SSNIP test seeks to identify the smallest market within which a hypothetical monopolist could profitably impose a SSNIP without a sufficient number of consumers/service purchasers switching to alternative products to render the price increase non-profitable.
<sup>84</sup> See paragraph 10.27 below.

<sup>&</sup>lt;sup>85</sup> [...] response to question 11 of the Questionnaire to Customers.

<sup>&</sup>lt;sup>86</sup> [...] response to question 19 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>87</sup> Phone call record with [...].

centre) may have space constraints that would prevent them from switching. In addition, the same supplier noted that setting up an OPL tends to be cyclical and depends on the following: the relevant finance personnel; the housekeeper; the availability of staff for the OPL; poor outsourced service quality (e.g., the number of linen items rejected due to being dirty); and the management of the hotel.

- 3.22 Furthermore, this supplier of laundry equipment noted that while some hotels and hospitals do the laundry of some of the linen items in-house, they also outsource their flat linen rental and maintenance services requirements in respect of other items. For instance, it was noted that while [...] has an OPL, it also outsources rental and maintenance of surgical scrubs. In addition, certain hotels have an OPL in respect of towels, but outsource linen and maintenance services in respect of other products (e.g., bed linen).
- 3.23 In relation to wash-only launderettes, the majority of Customers that responded to the Questionnaire to Customers indicated that they would not consider using a wash-only launderette in response to a 5%-10% price increase of flat linen rental and maintenance services.<sup>88</sup> [...] noted that while using wash-only launderettes is a potential option, there would be concerns whether wash-only launderettes would be able to ensure daily delivery of linen.<sup>89</sup> [...] noted that purchasing all necessary linen items (instead of renting them) would also require a substantial capital investment.<sup>90</sup>
- 3.24 Finally, the Commission notes Competitors' views on the likely scope of the geographic market. The Competitors considered that linen laundries based in Northern Ireland do not tend to have significant activities in the State. In this regard, [...] stated that it expects to expand its activities in the State due to the fact that some of its customers in Northern Ireland are building additional sites in the State and want [...] to provide services to their new sites in the State.<sup>91</sup> Similarly, linen laundries based in the State are expanding their activities in Northern Ireland.

#### Views of the Commission

#### Previous determinations

- 3.25 In its merger determination *M/05/045 HTS International / National Linen*<sup>92</sup>, the Commission's predecessor the Competition Authority (the "Authority") made a distinction between different types of activities in which undertakings involved in the relevant transaction were engaged and identified flat linen rental and maintenance services as a distinct category of services. There was no definitive view on the precise definition of the relevant product and geographic markets in that case.
- 3.26 In its decision CA/1135/92 Spring Grove / Conkenner (the "1993 Decision"), the Authority examined arrangements for the purchase of the entire issued capital of

<sup>&</sup>lt;sup>88</sup> See paragraph 10.28 below.

<sup>&</sup>lt;sup>89</sup> [...] response to question 12 of the Questionnaire to Customers.

 $<sup>^{\</sup>rm 90}$  [...] response to question 12 of the Questionnaire to Customers.

<sup>&</sup>lt;sup>91</sup> The Commission's telephone call with [...] dated 20 February 2019.

<sup>&</sup>lt;sup>92</sup> The determination in this case can be accessed at: <u>https://www.ccpc.ie/business/mergers-acquisitions/merger-notifications/m05045-hts-international-national-linen/.</u>



Conkenner Limited by Spring Grove Ireland Limited from Initial Services (International) Limited pursuant to Section 4 of the Competition Act 1991.

- 3.27 The Authority defined a broad market for the provision of textile rental and associated services, but assessed the likely competitive impact of the relevant transaction by reference to four distinct service categories<sup>93</sup> provided on a national basis.<sup>94</sup> It should also be noted that, in that case, the Authority decided that "… in-house provision is a sufficiently close substitute for bought-in services for it to be considered part of the same market".<sup>95</sup>
- 3.28 The previous determinations set out above are of very limited value in terms of addressing the relevant product market in this case, in particular given how long ago they were decided, and given market developments since then.

#### Relevant product markets

- 3.29 Having regard to the available evidence and product market definitions adopted in other jurisdictions, the Commission identified 4 questions relevant to the identification of the product market definition in this case:
  - a) does the supply of flat linen rental and maintenance services to: (i) healthcare; and (ii) hospitality customers form distinct product markets? (discussed in paragraphs 3.32 to 3.443.43 below);
  - b) is self-supply of flat linen and its maintenance services via OPLs in the same product market as the supply of outsourced flat linen rental and maintenance services? (discussed in paragraphs 3.45 to 3.78 below);
  - c) does the supply of different categories of flat linen service constitute distinct product markets? (discussed in paragraphs 3.79 to 3.85 below); and
  - d) does the supply of rental and maintenance of workwear constitute a distinct product market from the rental and maintenance of flat linen? (discussed in paragraphs 3.86 to 3.92 below).
- 3.30 The Commission notes that the Parties agreed that (i) there are separate relevant markets for healthcare and hospitality; (ii) that the supply of flat linen maintenance-only services is not included in the healthcare or the hospitality market; and that (iii) the provision of workwear rental and maintenance services (with the exception of scrub suits and other miscellaneous items provided by linen laundries to healthcare customers) is not included in the healthcare and hospitality markets. As these aspects of the relevant product markets are not disputed, the Commission provides a summary outline of its consideration below.
- 3.31 The Parties did not agree that self-supply should be excluded from the relevant markets.

 <sup>&</sup>lt;sup>93</sup> These were: Hand Drying Facilities, Workwear Rental, Dustmat Rental and Linen Services. See paragraph 6 of the 1993 Decision.
 <sup>94</sup> While the Authority did not come to a definitive view on the precise definition of geographic markets, market share estimates were made on the basis of companies' State-wide turnover. *Ibid*, paragraph 6.

<sup>&</sup>lt;sup>95</sup> Paragraph 32 of the 1993 Decision.



# Does the supply of outsourced flat linen rental and maintenance services to: (i) healthcare; and (ii) hospitality customers form distinct product markets?

#### Demand for linen items

- 3.32 The Commission's view is that differences in product characteristics between the type of outsourced flat linen required by healthcare and hospitality customers indicates that they are not close substitutes.
- 3.33 From a demand-side perspective, the examination of the Parties' contracts with their customers as well as ITTs issued by customers indicated that, in general, organisations in the hospitality sector require different linen items when compared to organisations in the healthcare sector. For example, while organisations in both sectors need bed linen, rented linen items differ in terms of their composition and size.<sup>96</sup> Information provided by Berendsen indicates that a significant share of rented linen items by its customers in the hospitality sector is accounted for by double and king size bed linen items ([...]%).<sup>97</sup> These items are not required by Berendsen's healthcare customers. Similarly, bed blankets account for [...]% of all rented items by Berendsen healthcare customers.

#### **Differences in processing**

- 3.34 The Commission's view is that the specific requirements of healthcare customers for processing means that a significant number of customers would not find processing associated with the supply of linen to hospitality customers to be a good substitute for the processing associated with the supply of linen to healthcare customers.
- 3.35 There are certain differences between customers' requirements for linen processing. In particular, ITTs issued by healthcare customers (e.g., hospitals) require adherence to certain hygiene standards that specify the level of quality required for processing of flat linen in terms of bio-contamination control.<sup>98</sup>
- 3.36 Hygiene standard specification is not uniform across customers in the healthcare sector, with nursing/care homes generally not requiring linen laundries' adherence to specific hygiene standards.<sup>99</sup> It should also be noted that not all hospitals require linen laundries to have hygiene standard accreditations.<sup>100</sup> However, according to Kings Laundry, such accreditations "...does improve a tender submission particularly in the healthcare sector".<sup>101</sup>

<sup>&</sup>lt;sup>96</sup> For example, as noted in paragraph 2.2 above, certain hotels specify the quality of linen to be supplied by linen laundries. More generally, customers in the hospitality sector require bed sheets for larger beds.

<sup>&</sup>lt;sup>97</sup> Berendsen's response to Question 8 of the Second RFI. Similar information was provided by Kings Laundry. See Kings Laundry's response to Question 7 of the Second RFI.

<sup>&</sup>lt;sup>98</sup> The Commission understands that there are two commonly cited standards: (i) NHS Executive HSG(95)18. <u>"Hospital laundry arrangements for used and infected linen</u>", dated 21 April 1995; and (ii) and BS EN 14065:2002, <u>"Textiles – Laundry Processed Textiles – Biocontamination Control System</u>", dated 6 December 2002 (known in the industry as the risk analysis biocontamination control ("RABC") standard). In 2016 this standard was superseded by BS EN 14065:2016.

<sup>&</sup>lt;sup>99</sup> The Commission also notes the Parties' argument that wash-only launderettes can provide services to both hospitality customers and nursing/care homes which indicates that the Parties' consider flat linen maintenance requirements to be similar for these categories of customers.

<sup>&</sup>lt;sup>100</sup> For example, while Kings Laundry [...], it has not obtained formal accreditation to date.

<sup>&</sup>lt;sup>101</sup> Kings Laundry's response to Question 21 of the First RFI.



- 3.37 The Commission notes that, unlike in the healthcare sector, customers in the hospitality sector do not specify hygiene standards to be used by linen laundries when processing flat linen. However, linen laundries can apply a healthcare linen maintenance process when maintaining flat linen used by hospitality customers. In fact, Kings Laundry indicated that it uses [...] at its facilities.<sup>102</sup>
- 3.38 The Commission observed some degree of demand-side substitutability between the two sectors. In general, customer contracts and ITTs do not specify what linen management system must be used<sup>103</sup> with linen laundries frequently using a pooling system. In this regard, Berendsen noted that [...].<sup>104</sup> Berendsen stated that [...].<sup>105</sup> Kings Laundry also indicated that operation of a pooling system for flat linen that is used in both sectors would be possible, if the linen laundry is adhering to the linen processing standard requested by healthcare customers.

#### Differences in competitive conditions

- 3.39 The Commission notes the Parties' views in relation to different levels of competition in the healthcare and hospitality sectors. Currently, there are five linen laundries providing outsourced flat linen rental and maintenance services to both healthcare and hospitality sectors in the State.<sup>106</sup> There is no linen laundry focusing on the provision of outsourced flat linen rental and maintenance services to the healthcare sector only. In comparison, in addition to linen laundries operating in both sectors, there are six linen laundries currently focusing on the hospitality sector only in the State.<sup>107</sup>
- 3.40 Hence, the number of competitors providing flat linen rental and maintenance services to customers in the hospitality sector is significantly higher than the number of competitors servicing customers in the healthcare sector. For example, presence measured by market shares demonstrates that linen laundries such as [...], while being somewhat successful in the hospitality sector have not yet been able to penetrate the healthcare sector to any material extent.<sup>108</sup> The differing competitive landscapes in the healthcare and hospitality sectors thus point to distinct product markets.

#### Precedent

3.41 The Commission notes the OFT's reasoning in *Johnsons Apparelmaster/Cannon*. The activities of both parties in this transaction included the provision of flat linen rental and maintenance services. In assessing the segmentation of the product market by textile type, <sup>109</sup> the OFT stated that there was "*mixed evidence*" on the degree of supply-

<sup>&</sup>lt;sup>102</sup> Kings Laundry's response to Question 21 of the First RFI.

<sup>&</sup>lt;sup>103</sup> As noted in paragraph 2.2 above, certain hotels require their logo embroidered on linen items, thus making such items unusable by other customers of linen laundry.

<sup>&</sup>lt;sup>104</sup> Berendsen's response to the Commission's queries dated 30 November 2018.

<sup>&</sup>lt;sup>105</sup> Berendsen's response to the Commission's queries dated 30 November 2018.

<sup>&</sup>lt;sup>106</sup> These are: Berendsen, Kings Laundry, [...]. Kings Laundry currently services private hospitals and nursing/care homes while [...]

play a very minor role in the healthcare sector. In addition, CWS-boco provides scrub suit rental and maintenance services.

<sup>&</sup>lt;sup>107</sup> These are: [...] provides services to healthcare customers in [...].

<sup>&</sup>lt;sup>108</sup> See the Commission's market share analysis in paragraphs 4.56 to 4.76 and 4.212 to 4.222 below.

<sup>&</sup>lt;sup>109</sup> The OFT identified four product categories: workwear garments, healthcare, hospitality and washroom services.



side substitutability available to it<sup>110</sup> and examined the relevant transaction with reference to each type of textile on a *"cautious basis"*.<sup>111</sup>

3.42 The Commission's view is consistent with precedent in other jurisdictions. For example, very recently, the UK's Competition and Markets Authority ("CMA") in the case *Vanilla Group / Washstation*<sup>112</sup> considered that: (i) specific requirements for laundry services from higher education customers; and (ii) required experience in the supply of services to higher education customers collectively represented a barrier for providers of laundry services when switching capacity from other customer segments.<sup>113</sup> The Commission considers that a similar barrier is faced by linen laundries that currently supply outsourced flat linen rental and maintenance services to customers in only the hospitality sector.

#### Conclusion on distinct product markets for healthcare and hospitality customers

- 3.43 The Commission notes that Kings Laundry has not provided outsourced flat linen rental and maintenance services to public healthcare customers to date. Thus, within the healthcare sector, currently the Parties activities overlap only with respect to private hospitals and nursing homes. However, while the Commission takes account of variations in how the Parties address different customer segments, it maintains that the supply of outsourced flat linen rental and maintenance services to healthcare customers constitutes a single product market, and notes that the Parties agree with this.
- 3.44 For the reasons set out in paragraphs 3.32 to 3.42, the Commission considers that the supply of outsourced flat linen rental and maintenance services to: (i) healthcare customers; and (ii) hospitality customers form two distinct product markets. In coming to this conclusion, the Commission has considered differences in the nature of demand from healthcare and hospitality customers, and differences in the conditions of supply.

# Is the self-supply of flat linen and its maintenance services via OPLs in the same product market as the supply of outsourced flat linen rental and maintenance services?

- 3.45 The Commission's investigation does not support the inclusion of the self-supply of outsourced flat linen rental and maintenance services via OPLs in the relevant product markets. Rather, the Commission is of the view that it is more appropriate to consider self-supply in the analysis of competitive effects when examining CBP.
- 3.46 Having found that there are separate relevant healthcare and hospitality markets, the Commission's analysis continues by assessing each market in turn.

#### Healthcare market

3.47 The Commission notes that some healthcare customers currently self-supply flat linen through the use of OPL and that this is the case across different customer types within the healthcare market (i.e., it is true for some hospitals and some nursing homes). However, in order to be considered part of the relevant market, the Commission would

<sup>&</sup>lt;sup>110</sup> Johnsons Apparelmaster/Cannon, paragraph 30.

<sup>&</sup>lt;sup>111</sup> *Ibid*, paragraph 30.

<sup>&</sup>lt;sup>112</sup> See <u>https://www.gov.uk/cma-cases/vanilla-group-washstation-merger-inquiry.</u>

<sup>&</sup>lt;sup>113</sup> See paragraph 6.34 of the CMA's report titled "JLA and Washstation A report on the completed acquisition by JLA

New Equityco Limited of Washstation Limited" dated 11 October 2018.

expect to see evidence that healthcare customers operating OPLs are competing with linen laundries that provide outsourced flat linen rental and maintenance services by

linen laundries that provide outsourced flat linen rental and maintenance services by, for example, participating in tenders issued by purchasers of outsourced flat linen rental and maintenance services. This is not the case.

3.48 For completeness, the Commission has considered the degree of switching between healthcare customers who purchase outsourced flat linen rental and maintenance services, and customers who self-supply.

#### Propensity to switch

- 3.49 The Commission notes that none of the Customers in the healthcare sector contacted directly by the Commission through the Customer Questionnaires considered switching to self-supply to be a viable option in response to a 5% to 10% price increase by their supplier of outsourced flat linen rental and maintenance services.<sup>114</sup>
- 3.50 The Amárach Survey indicated that none of the respondents from the healthcare sector would consider switching to self-supply in response to a 5%-10% price increase imposed by their current provider of outsourced flat linen rental and maintenance services.<sup>115</sup> A subset of respondents who indicated that they would switch to another provider were then asked to identify their most likely action in response to a hypothetical 5% to 10% price increase by all suppliers of outsourced flat linen rental and maintenance services. Just one respondent from the healthcare sector considered switching to self-supply in response to this market-wide price increase.<sup>116</sup>
- 3.51 The Commission notes the Parties' argument that the Commission's Assessment failed to compare the actual level of customer switching in response to a hypothetical 5% to 10% price increase presented in the Amárach Survey to a critical level of switching that would render such price increase by a hypothetical monopolist unprofitable. The Commission's approach to the hypothetical monopolist test took into account the fact that there is a significant variation between customers in both healthcare and hospitality sectors in terms of their size and their spend on outsourced flat linen rental and maintenance services. In circumstances where there is a significant variation between customer size, the Commission considers that the relevant question to consider is what is the level of actual sales loss rather than what is the number of lost customers.<sup>117</sup> Thus, the Commission.<sup>118</sup> As noted in paragraph 10.27 below, those Customers unanimously considered that switching to an OPL would not be a viable option in

<sup>&</sup>lt;sup>114</sup> See paragraph 3.17 above.

<sup>&</sup>lt;sup>115</sup> See slide 51 of the Amárach Survey.

<sup>&</sup>lt;sup>116</sup> See slide 55 of the Amárach Survey.

<sup>&</sup>lt;sup>117</sup> In this regard, in paragraph 4.16 of the document titled "<u>Good practice in the design and presentation of customer survey evidence</u> <u>in merger cases</u>", dated May 2018, the CMA reference number CMA78 (the "CMA's Survey Guidelines") the CMA notes that "In most merger situations, the unit that we are conceptually most interested in is the value of sales in monetary terms. The diversion ratio therefore becomes the value of sales that are diverted to the merger Party over the total value of lost sales". The Commission notes that in circumstances where all customers have similar level of expenditure on services in question, these two metrics would provide similar results.

<sup>&</sup>lt;sup>118</sup> The significance of these Customers to the Parties (in terms of turnover) is highlighted in paragraph 1.15 above.



response to a 5% - 10% price increase of outsourced flat linen rental and maintenance services.

- 3.52 As noted in paragraph 1.41 above, the purpose of the Amárach Survey is to inform the Commission's investigation and to supplement the Market Enquiries undertaken directly by the Commission. As part of the Amárach Survey, the respondents were asked to identify their most likely action in response to a hypothetical 5% to 10% price increase<sup>119</sup> by only their supplier of outsourced flat linen rental and maintenance services.<sup>120</sup> The respondents that indicated that they would switch to another provider were then asked to identify their most likely action in response to a hypothetical 5% to 10% price increase by all suppliers of outsourced flat linen rental and maintenance services.<sup>121</sup>
- 3.53 The Commission has reviewed its approach to survey design, and in particular to the design of questions testing reaction to a SSNIP. The Commission considers that in the case where a respondent stated that it would do nothing if its supplier alone were to increase its price (i.e., when they have an option to choose other suppliers that do not increase prices), it is entirely appropriate to assume that this respondent would do nothing if every supplier increased prices. In this case, the customer would not have the option to choose an alternative supplier that did not increase prices. Thus, the Commission considered it appropriate to confine the question about what a respondent would do in response to a market-wide price increase to those respondents who stated that they would switch to another provider in response to their own supplier's price increase.

#### Review of Parties' evidence

- 3.54 Information received from the Parties and Competitors indicates that customers mainly switch to an alternative provider of flat linen rental and maintenance services rather than set up OPLs. In this regard, the Commission notes that the Parties were able to cite only one example of an organisation in the healthcare sector switching to OPL.<sup>122</sup> On the other hand, the Commission's investigation revealed a number of examples where healthcare organisations shut down their OPLs and started to purchase outsourced flat linen rental and maintenance services. These include four hospitals [...].<sup>123</sup> Furthermore, information provided by Berendsen indicates that in 2018 [...] nursing homes with OPLs started to purchase outsourced flat linen rental and maintenance services from Berendsen.<sup>124</sup>
- 3.55 Elis' financial statements further support the exclusion of self-supply from the relevant product market. See, for example, Figure 3, an extract of page 43 of Elis' 2017 annual report, which highlights that once the services have been outsourced they are very

<sup>&</sup>lt;sup>119</sup> See footnote 83 above.

<sup>&</sup>lt;sup>120</sup> Slides 51 to 54 of the Amárach Survey.

<sup>&</sup>lt;sup>121</sup> Slides 55 of the Amárach Survey.

<sup>&</sup>lt;sup>122</sup> During the Commission's meeting with Berendsen on 30 January 2019, Berendsen cited [...] and noted that this [...] switched to self-supply in 2017.

<sup>&</sup>lt;sup>123</sup> See the record of Commission's meeting with Berendsen on 30 January 2019 and the record of the Commission's meeting with [...].

<sup>&</sup>lt;sup>124</sup> Berendsen's response to Question 10 of the Second RFI.



Competition and Consumer Protection Commission

difficult to be brought back in-house.<sup>125</sup> The Commission notes that findings of its investigation concur with Elis's view that switching tends to be one way, and that once services are outsourced, they are not likely to be brought back in-house.

Figure 3: Extract from Elis' 2017 Annual Report (I)

# The indispensable nature of the Group's services

The flat linen, workwear and HWB appliance rental, laundry and maintenance services the Group provide are essential to its customers' businesses, particularly those in the Hospitality and Healthcare markets. Once these services have been outsourced they are very difficult to repatriate since doing so would require substantial capital expenditure.

Furthermore, the average monthly fee that Group customers pay for the supply and maintenance of flat linen, workwear and HWB appliances is relatively cheap compared to their other expenses. The Group estimates that half of its customers pay less than €150 a month.

Lastly, the Group is able to develop new and complementary products and services for sale to its current customer base.

3.56 While Elis' financial statements refer to OPLs in the section titled "MAIN COMPETITORS" (see Figure 4 below), <sup>126</sup> the Parties' internal documents generally refer to OPLs as [...].<sup>127</sup> Berendsen's main competitors in Ireland, according to Elis, were: "CWS-Rentokil<sup>128</sup>, Celtic, Kings".

#### Figure 4: Extract from Elis' 2017 Annual Report (II)

#### 1.7.2 MAIN COMPETITORS

It is important to note that certain potential customers may choose not to outsource the supply and maintenance of their flat linen, workwear or HWB appliances, preferring to use in-house solutions. For example, some companies make their employees responsible for laundering their workwear in return for a payment that is supposed to cover the costs of laundering. Other companies install in-house laundries to launder flat linen used in their business or workwear worn by their employees. The Group also faces competition from shared hospital laundries and "ESAT" laundries (entities that employ disabled workers).

For the target market, consisting of businesses that have opted to use a rental and maintenance service, the Group's competitors vary depending on the operating sector and the type of services provided by the Group.

<sup>&</sup>lt;sup>125</sup> See Elis' document titled "2017 Registration Document" dated 25 April 2018, which can be accessed at: <u>https://www.corporate-elis.com/en/investor-relations</u>.

<sup>&</sup>lt;sup>126</sup> Page 40 and 41 of Elis' 2017 annual report.

<sup>&</sup>lt;sup>127</sup> For example, see page 3 of Berendsen's internal document titled [...].

<sup>&</sup>lt;sup>128</sup> As noted in paragraph 4.65 below, CWS-Rentokil (trading as CWS-boco in Ireland) ceased to provide flat linen rental and maintenance services in Ireland in 2017.



Competition and Consumer Protection Commission

For HWB appliance rental and maintenance services in particular, the Group is in competition with cleaning companies which may be customers of the Group for this type of service, and facility management companies which offer a full range of services including HWB appliance rental and maintenance.

Several major groups operate in all three business segments. Following the merger announced between certain European activities of Rentokil Initial and CWS Boco, the latter will be Elis's main competitor in continental Europe. The Group is also faced with competition from the main national players, the most important of which are indicated below:

- France: Rentokil Initial, Kalhyge (formerly RLD), Anett, Sdez;
- ➔ UK: Johnson Service, Initial, PHS, SynergyHealth;
- Ireland: CWS-Rentokil, Celtic, Kings;
- Sweden: Textilia, CWS-Rentokil;
- Denmark: DFD;
- Poland, Baltic counties: CWS-Rentokil, Lindström;

- ➔ Austria: Salesianer;
- Czech Republic, Hungary, Slovakia: Lindström, Salesianer;
- Netherlands: CWS-Rentokil, CleanLease, LIPS;
- Norway: Nortekstil, Sentralvaskertiet, Stil;
- Finland: Lindström;
- Brazil: Alsco, Servizi Italia, Renova;
- Spain: Ilunion, PHS, CWS-Rentokil, L'Emporda;
- Germany: CWS-Rentokil, Mewa, Bardusch, Alsco;
- Switzerland: CWS-Rentokil, Bardusch, LBG;
- ➔ Portugal: Serlima;
- Belgium-Luxembourg: CWS-Rentokil, Cleanlease, Sterima, Mewa;
- Italy: Servizi Italia, Servizi Ospedalieri, Pedersoli, Alsco, Rentokil Initial;
- Chile: Golden Clean, Lavinur.
- 3.57 Figure 5 highlights an extract of page 41 of Elis' 2017 annual report which refers to a general trend toward outsourcing with the reasons for customers switching to outsourcing flat linen rental and maintenance services (amongst other services provided by Elis) listed on page 42 of Elis' 2017 annual report.<sup>129</sup>

Figure 5: Extract from Elis' 2017 Annual Report (III)

#### 1.7.3 MARKET DEMAND DRIVERS FOR RENTAL, LAUNDRY AND MAINTENANCE SERVICES

The following general and sector-specific trends are the main drivers of demand for flat linen, workwear and HWB appliance rental, laundry and maintenance services in the Hospitality, Healthcare, Industry, and Trade and Services markets:

#### Outsourcing

Demand for the Group's flat linen, workwear and HWB products and services in the Hospitality, Healthcare, Industry, and Trade and Services end markets is driven by a general trend toward outsourcing. The Group's customers in France tend to outsource these services to a single service provider, such as the Group. According to a study carried out by KPMG, about one-third of textile rental and laundry services were outsourced in France and in Europe as a whole in 2013. The chart below shows the outsourcing rates for flat linen, workwear and HWB appliance rental, laundry and maintenance services in Europe in 2013 as well as outsourcing potential at maturity:



Source: KPMG (Survey, March 2017).

According to ETSA, the market's upside potential <sup>(1)</sup> (the study did not provide an exact date, only a time horizon) should mainly be driven by an outsourcing rate that is likely to double in Europe to reach 55% to 65% according to market estimates.

# 3.58 The level of outsourcing flat linen rental and maintenance services amongst customers is relatively high. The Parties estimated that in 2017 the level of outsourcing was

<sup>&</sup>lt;sup>129</sup> Cited reasons include focus on customers' core business, reduction of fixed costs and more efficient management of expenditure, simplification of personnel management, freeing up space for other use, improve the quality of textile laundering and care, enhance brand image and reputation and choose supplier that supports sustainable development.



approximately 63% in the healthcare sector.<sup>130</sup> Although not a direct comparison, this situation contrasts with the situation described in the 1993 Decision where it was noted that the level of outsourcing for textile rental and associated services was 25%.<sup>131</sup>

3.59 The Commission's investigation revealed that the linen laundries industry expects the HSE to close its remaining OPLs and outsource flat linen rental and maintenance services to linen laundries in the short to medium term. For example, in paragraph 6.20 of the report titled *"CELTIC LINEN LIMITED, REPORT OF THE INDEPENDENT EXPERT PURSUANT TO SECTION 511 OF THE COMPANIES ACT 2014"* dated 28 September 2016 (the "IER report") it is noted that:

"One of the growth potential areas in the Healthcare sector relates to in house laundries operated by the HSE. Management understand that these laundries are inefficient and are likely to be outsourced by the HSE in the future. The incremental market is estimated to be worth  $\notin$ 7.0 million annually."<sup>132</sup>

3.60 [...] told the Commission that currently it [...], but noted that purchasing modern laundry equipment would require significant capital investment.

Conclusion on inclusion of self-supply via OPLs in the relevant healthcare product market

- 3.61 The Commission considers that self-supply via OPL is not in the same relevant product market as outsourced flat linen rental and maintenance services for the following reasons:
  - a) the fact that some potential purchasers of outsourced flat linen rental and maintenance services currently operate OPLs does not in itself indicate that self-supply is an effective substitute for customers that currently outsource flat linen rental and maintenance services. As noted in paragraph 3.58, currently the majority of customers in the healthcare sector prefer outsourcing to self-supply.
  - b) the fact that a very limited set of customers switched to self-supply and did not revert back to linen laundries does not in itself merit the inclusion of self-supply in the relevant product market. The relevant question in assessing this issue is whether a sufficient number of customers currently purchasing outsourced flat linen rental and maintenance services would switch to self-supply, thus preventing the merged entity from imposing price increases. The Commission's investigation revealed this not to be the case.<sup>133</sup> For example, the Parties were able to cite only one example of a customer in the healthcare sector switching to self-supply;
  - c) the existence of several suppliers of industrial/commercial laundry equipment (including second hand equipment) in the State does not necessarily contradict

<sup>&</sup>lt;sup>130</sup> The Commission's analysis of information set out in page 22 of the Merger Notification Form.

<sup>&</sup>lt;sup>131</sup> In paragraph 7 of the 1993 Decision, the Authority quotes the relevant parties' estimate of 75% for customers that self-supply textile rental and associated services.

<sup>&</sup>lt;sup>132</sup> Paragraph 6.20 of the IER report. Also see page 1 of the document titled [...], slide 9 of the document titled [...] dated July 2017 submitted by Berendsen in response to Question 9 of the RFI and page 9 of the document titled "Celtic Linen Limited Investment Opportunity" dated September 2016 submitted by Kings Laundry in response to question 10 of the First RFI.

<sup>&</sup>lt;sup>133</sup> See paragraph 3.17 above.

Competition and Consumer Protection Commission

evidence that most customers do not consider self-supply to be a good substitute for the purchase of outsourced flat linen rental and maintenance services. Notwithstanding the views expressed by suppliers of laundry equipment, the Commission points to high rates of outsourcing flat linen rental and maintenance services and evidence in the Parties' internal documents noting [...].<sup>134</sup> The Parties in the Report on OPLs state that they themselves routinely purchase second hand equipment.<sup>135</sup> Thus, the existence of several suppliers of industrial/commercial laundry equipment does not in and of itself merit the inclusion of self-supply in the relevant product market; and

- d) the Commission does not agree with the Parties' view that the comparison of OPL set-up costs with customer's revenue is an appropriate test in determining whether self-supply should be included in the relevant product market. In the Commission's view, the appropriate test is whether in response to a 5%-10% price increase a switch to OPL would be economically justifiable. In this regard, the Commission refers to the views of Customers summarised in paragraph 3.17 above. The Parties themselves list a wide range of costs (including lost opportunity costs) that customers must take into account when considering a switch to self-supply.<sup>136</sup>
- 3.62 The Commission does not agree with the Parties that no factor in the OFT's reasoning for excluding self-supply in *Johnsons Apparelmaster/Cannon* applies in this case. In particular, the Commission notes that adhering to certain hygiene standards is a very significant factor for healthcare customers when considering whether hosting an OPL is a viable alternative to outsourcing of flat linen rental and maintenance services. This was confirmed by several Customers from the healthcare sector.<sup>137</sup> These Customers noted that ensuring prevention of biocontamination requires significant expertise and is one of the reasons why switching to self-supply in response to a 5-10% price increase would not be a viable option.
- 3.63 To conclude, the Commission is of the view that self-supply of flat linen rental and maintenance services in the healthcare sector should not form a part of the relevant product market for the reasons set out in paragraphs 3.47 to 3.62 above.

#### Hospitality market

3.64 Similarly to the healthcare market, the Commission recognises that some hospitality customers currently self-supply flat linen through the use of OPLs. However, the evidence does not support the inclusion of OPLs in the relevant product market. In particular, the Commission investigation has not shown that hospitality customers operating OPLs are competing with linen laundries that provide outsourced flat linen rental and maintenance services<sup>138</sup> by, for example, participating in tenders issued by purchasers of outsourced flat linen rental and maintenance services.

<sup>&</sup>lt;sup>134</sup> Page 1 of the document titled [...] submitted by Berendsen in response to question 9 of the First RFI.

<sup>&</sup>lt;sup>135</sup> Paragraph 33 of the Report on OPLs.

<sup>&</sup>lt;sup>136</sup> See <u>https://www.kingslaundry.com/what-is-linen-rental/</u> accessed on 5 December 2018.

<sup>&</sup>lt;sup>137</sup> For example, see the Commission's minutes of meeting with [...].

<sup>&</sup>lt;sup>138</sup> The Commission notes one minor exception. [...] indicated that it provides table linen rental and maintenance services to a restaurant located nearby.



3.65 For completeness the Commission has considered the degree of switching between hospitality customers who purchase outsourced flat linen rental and maintenance services, and customers who self-supply.

#### Propensity to switch

- 3.66 The Commission notes that none of the Customers in the hospitality sector contacted directly by the Commission through the Customer Questionnaires (i.e., the most significant customers identified by the Parties) considered switching to self-supply to be a viable option in response by a 5% to 10% price increase by their supplier of outsourced flat linen rental and maintenance services.<sup>139</sup>
- 3.67 The Amárach Survey indicated that just 2% of respondents from the hospitality sector would consider switching to self-supply in response to a 5%-10% price increase imposed by their current provider of outsourced flat linen rental and maintenance services.<sup>140</sup> A subset of respondents who indicated that they would switch to another provider were then asked to identify their most likely action in response to a hypothetical 5% to 10% price increase by <u>all</u> suppliers of outsourced flat linen rental and maintenance services. In total 11%<sup>141</sup> of respondents from the hospitality sector considered switching to self-supply<sup>142</sup> to be a viable option. It should also be noted that the likelihood of switching to self-supply decreases with the customer size, and that it is a less viable option the larger the customer. This is illustrated in Table 1 below. The level of actual sales loss as a result of customers switching to OPL is lower than the level of customer loss reported in the Amárach Survey.<sup>143</sup>

Hospitality Sector						
	<10	10-24	25-49	50-100	100+	Total
	<u>0</u>	<u>8</u>	<u>4</u>	<u>12</u>	8	<u>32</u>
Keep purchasing current linen laundry services	0%	63%	50%	66%	50%	59%
Cancel my current service and start to self- supply linen laundry services	0%	38%	25%	17%	13%	22%

Table 1: Hospitality sector respondents' replies to a SSNIP question, by respondent size

<sup>139</sup> See paragraph 3.17 above.

<sup>140</sup> See slide 51 of the Amárach Survey.

<sup>&</sup>lt;sup>141</sup> This figure is calculated from the 39% who said they would switch provider in response to their own provider price increase. Out of this subsample, 22% said that they would switch to self-supply in response to an industry-wide price increase. The figure is a combination of the initial 2% who said they would switch to self-supply plus the 22% of the subsample.

<sup>&</sup>lt;sup>142</sup> Slides 51 and 55 of the Amárach Survey.

<sup>&</sup>lt;sup>143</sup> In this regard, see slide 19 of the Amárach Survey where respondents' spend by respondent size is reported.



Competition and Consumer Protection Commission

Other specify	0%	0%	0%	0%	0%	0%
Don't know	0%	0%	25%	17%	38%	19%

Source: The Amárach Survey

3.68 The Commission notes the Parties' comments in relation to the treatment of "Don't know" and "Other" responses.<sup>144</sup> In this regard, the CMA's Survey Guidelines state the following:

"However, if the customer answers that they would divert to another supplier but in a subsequent question says they do not know which supplier, then this answer is partially informative because they have stated that they would have diverted their expenditure rather than staying with the merger Party or **exiting the market**.<sup>145</sup> In these circumstances, usual practice is to allocate 'don't know' responses in the same proportions as those who have explicitly named the retailer to which they would divert. (emphasis added)"<sup>146</sup>

- 3.69 The Commission interprets the CMA's statement to apply only in the context of estimating diversion ratios between competitors rather than a diversion ratio to an 'exiting the market' option (i.e., stop purchasing and start self-supplying the required services). Therefore, the Commission does not agree that a proportion of respondents who have not explicitly stated whether they will switch to self-supply or not should be assumed as switchers to self-supply. In any case, the Commission reiterates that the Amárach Survey results are not considered definitive in themselves, but are considered alongside empirical data/evidence, where available, in particular, alongside information gathered from the Parties, Competitors and Customers via Market Enquiries.
- 3.70 Information received from the Parties and Competitors indicates that, notwithstanding examples provided by the Parties,<sup>147</sup> customers mainly switch their provider of flat linen rental and maintenance services rather than set up OPLs. In this regard the Commission refers to the following evidence:
  - a) information provided by Berendsen indicates that in the period from 1 January 2014 to 30 June 2018, [...] out of [...] hospitality customers that left Berendsen set up an OPL with the remaining customers switching to Berendsen's competitors;<sup>148</sup>
  - b) information provided by Kings Laundry indicates that in the period from 1 January 2014 to 30 June 2018, [...] out of [...] hospitality customers that left

<sup>&</sup>lt;sup>144</sup> Frontier Economics Report, paragraph 2.10 ii

<sup>&</sup>lt;sup>145</sup> Note that switching to self-supply is an exit from the merchant market

<sup>&</sup>lt;sup>146</sup> Paragraph 4.24 of the CMA's Survey Guidelines.

<sup>&</sup>lt;sup>147</sup> Also see Appendix 4 which sets out organisations that switched to and/from OPLs.

<sup>&</sup>lt;sup>148</sup> In total, Berendsen listed [...] hospitality customers, but for [...] of these customers it did not indicate whether customers switched to Berendsen's competitors or set up OPLs.

Kings Laundry set up OPLs<sup>149</sup> with the remaining customers switching to Kings Laundry's competitors;<sup>150</sup> and

- c) information provided by [...] indicates that in the period from 1 January 2014 to 30 June 2018, [...], set up OPLs with the remaining customers switching to [...] competitors<sup>151</sup>.
- 3.71 In this regard, the relevant evidence is the actual switching rate to OPLs across all major providers of outsourced flat linen rental and maintenance services. The switching figures presented in paragraph 3.70 indicate that this rate is very low.
- 3.72 The level of outsourcing of flat linen rental and maintenance services amongst customers is relatively high. The Parties' estimated that in 2017 the level of outsourcing was approximately 90% in the hospitality sector.<sup>152</sup> This contrasts with the situation described in the 1993 Decision where it was noted that the level of outsourcing for textile rental and associated services was 25%.<sup>153</sup>
- 3.73 Some Customers from the hospitality sector (e.g., [...]) contacted directly by the Commission through the Customer Questionnaires noted that they plan to close OPLs that currently cater for some of their linen maintenance requirements and outsource these services to linen laundries. In this regard, the Amárach Survey indicated that 5% of respondents from the hospitality sector that currently operate OPLs stated their intention to purchase outsourced flat linen rental and maintenance services.<sup>154</sup> Furthermore, information provided by Berendsen indicates that in 2018 [...] hotels that previously self-supplied flat linen rental and maintenance services outsourced these services to Berendsen.<sup>155</sup>

#### Conclusion on inclusion of self-supply via OPLs in the relevant hospitality product market

- 3.74 The Commission considers that self-supply via OPL is not in the same market as outsourced flat linen rental and maintenance services in the hospitality sector for the following reasons:
  - a) the fact that some potential purchasers of outsourced flat linen rental and maintenance services currently operate OPLs does not in itself indicate that self-supply is an effective substitute for customers that currently outsource flat linen rental and maintenance services. As noted in paragraph 3.56, currently the majority of customers in the hospitality sector prefer outsourcing to self-supply.

<sup>&</sup>lt;sup>149</sup> The largest of these former Kings Laundry customers (in terms of generated turnover) is [...]. Also see the discussion on Kings Laundry's diversions to OPL in paragraph 4.139 to 4.140 below where it is noted that these switching figures should be considered in the context where Kings Laundry, on average, had [...] hospitality customers in the period from 1 January 2014 to 30 June 2018.

<sup>&</sup>lt;sup>150</sup> In total, Kings Laundry listed [..] hospitality customers, but for [...], [...] customers closed/went into liquidation and for [...] of these customers it did not indicate whether customers switched to Kings Laundry's competitors or set up OPLs.

<sup>&</sup>lt;sup>151</sup> See [...] response to question 14 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>152</sup> The Commission's analysis of information set out in page 22 of the Merger Notification Form.

<sup>&</sup>lt;sup>153</sup> In paragraph 7 of the 1993 Decision, the Authority quotes the relevant parties' estimate of 75% for customers that self-supply textile rental and associated services.

<sup>&</sup>lt;sup>154</sup> Slide 16 of the Amárach Survey.

<sup>&</sup>lt;sup>155</sup> Berendsen's response to Question 10 of the Second RFI.

Moreover, having regard to information provided by the Customers, further closures of OPLs in the hospitality sector are likely;<sup>156</sup>

- b) the fact that a very limited set of customers switched to self-supply and did not revert back to linen laundries does not in itself merit the inclusion of self-supply in the relevant product market. The relevant question in assessing this issue is whether a sufficient number of customers currently purchasing outsourced flat linen rental and maintenance services would switch to self-supply, thus preventing the merged entity from imposing price increases. The Commission's market investigation revealed this not to be the case.<sup>157</sup> For example, Kings Laundry noted that [...] of its customers switched to self-supply since 2014.<sup>158</sup> The Commission also notes that 2 Self-suppliers in the hospitality sector that used to purchase flat linen rental and maintenance services in the past stated that the decision to open OPLs was made primarily due to insufficient level of service quality provided by linen laundries (e.g., delayed deliveries and/or linen cleanliness not meeting required standards) rather than high prices;<sup>159</sup>
- c) the existence of several suppliers of industrial/commercial laundry equipment (including second hand equipment) in the State does not necessarily contradict evidence that most customers do not consider self-supply to be a good substitute for the purchase of outsourced flat linen rental and maintenance services. Notwithstanding the views expressed by suppliers of laundry equipment, the Commission points to high rates of outsourcing flat linen rental. The Parties in the Report on OPLs state that they themselves routinely purchase second hand equipment. Thus, the existence of several suppliers of industrial/commercial laundry equipment does not in and of itself merit the inclusion of self-supply in the relevant product market;
- d) while the Parties did provide limited examples of hotels switching to self-supply, the Commission considers that customers are more likely to threaten to switch to their suppliers' competitors rather than setting up an OPL and this is supported by evidence provided by Kings Laundry;<sup>160</sup>
- e) the Commission does not agree with the Parties that the comparison of OPL setup costs with customers' revenue is an appropriate test in determining whether self-supply should be included in the relevant product market. In the Commission's view, the appropriate test is whether in response to a 5%-10% price increase a switch to OPL would be economically justifiable. In this regard, the Commission refers to the views of Customers summarised in paragraph 3.17 above. The Parties themselves list a wide range of costs (including lost opportunity costs) that customers must take into account when considering a switch to self-supply;<sup>161</sup>

<sup>&</sup>lt;sup>156</sup> See paragraph 10.26 below.

<sup>&</sup>lt;sup>157</sup> See paragraph 3.70 above.

<sup>&</sup>lt;sup>158</sup> Kings Laundry's response to question 29 of the First RFI. Furthermore, Kings Laundry did not provide evidence that [...]. <sup>159</sup> See paragraph 10.40 below.

<sup>&</sup>lt;sup>160</sup> Kings Laundry provided [...] in response to question 33 of the First RFI.

<sup>&</sup>lt;sup>161</sup> See <u>https://www.kingslaundry.com/what-is-linen-rental/</u> accessed on 5 December 2018.


- f) the Commission agrees with the Parties' view that it is easier for customers that already have OPL(s) to take further laundry in-house (at least in theory). However, as noted in paragraph 3.18, hotel groups currently operating OPLs for some of their hotels indicated that a full switch over to existing OPLs is not possible due to capacity constraints. Thus, such groups would have to incur costs associated with expanding OPL capacity (whether by expanding current OPLs or setting up additional OPLs in different locations); and
- g) the Commission does not consider that retaining staff through their redeployment within the hotel during months of low occupancy rates would be a strong reason for considering a switch to OPL. On the contrary, the demand for flat linen rental and maintenance services would also be quite seasonal for such hotels, thus further reducing justification for maintaining an OPL which will not operate at full capacity during months of low occupancy.
- 3.75 The Commission reiterates its views (summarised in paragraphs 3.56 to 3.57 above) in relation to Elis' financial statements and references to OPLs as main competitors of Elis.
- 3.76 To conclude, the Commission is of the view that self-supply of flat linen rental and maintenance services in the hospitality sector should not form a part of the relevant product market for the reasons set out in paragraphs 3.45 to 3.75 above.

## Assessment of self-supply in the analysis of competitive effects

- 3.77 The Commission does not dispute the Parties' argument that there is a precedent of NCAs considering self-supply either in the analysis of market definition or in the analysis of competitive effects. However, as noted above, the Commission did not observe any material evidence<sup>162</sup> that hotels or hospitals operating OPLs are competing with linen laundries that provide outsourced flat linen rental and maintenance services. Thus, for all of the reasons set out above, the Commission considers that OPLs do not form a part of the merchant market.
- 3.78 However, the Commission recognises that in both the healthcare and hospitality markets, certain customers view self-supply as a viable option, and this could potentially act as some level of constraint on suppliers of outsourced flat linen rental and maintenance services. In the Commission's view, the consideration of self-supply is more appropriate in the analysis of competitive effects when examining CBP rather than at a market definition stage, and it is discussed further in Chapter 4.

# Does the supply of different categories of flat linen services constitute distinct product markets?

3.79 As noted in paragraph 2.7 above, services provided by linen laundries can be grouped into three broad categories (maintenance only service, rental and maintenance service and managed service). The Parties (as well as their main competitors) can provide all three types of services (depending on a customer's requirements). However, the Commission's investigation revealed that flat linen rental and maintenance services (either as Rental Service or as Managed Service) is the main product offering of both

<sup>&</sup>lt;sup>162</sup> [...] indicated that it provides table linen rental and maintenance services to a restaurant located nearby.



Parties and their competitors. For example, both Berendsen and Kings Laundry stated that [...] their customers (i.e., hospitality and healthcare customers) availed of outsourced flat linen rental and maintenance services as of June 2018.<sup>163</sup>

## Healthcare market

- 3.80 All Customers from the healthcare sector that responded to the Questionnaire to Customers indicated that they opt for a service package that includes both the rental and maintenance of flat linen.<sup>164</sup> Customer contracts and ITTs submitted by the Parties also highlight customers' preference for a service package.
- 3.81 The Commission notes that healthcare customers that chose to purchase flat linen rental and maintenance services would not be likely to find the use of wash-only launderettes to be a good substitute, because wash-only launderettes could not meet the hygiene requirements specified by healthcare customers. For example, a wash-only launderette is highly unlikely to have a barrier wall that prevents cross-contamination.
- 3.82 Hence, for the reasons set out in paragraphs 3.80 to 3.81 above, the Commission is of the view that the supply of flat linen maintenance-only services should not be included in the same product market as the supply of outsourced flat linen rental and maintenance services to customers in the healthcare sector.

## Hospitality market

- 3.83 All of the Customers from the hospitality sector that responded to the Questionnaire to Customers indicated that they opt for a service package that includes both rental and maintenance of flat linen<sup>165</sup> with some Customers purchasing a mix of services (e.g., maintenance only service for bed linen but rental and maintenance service for towels). Customer contracts and ITTs submitted by the Parties also highlight customers' preference for a service package.
- 3.84 The Commission's investigation did not find linen processing in wash-only launderettes to be a good substitute for outsourced flat linen rental and maintenance services. For example, the majority of Customers that responded to the Questionnaire to Customers indicated that they would not consider purchasing linen and outsourcing its maintenance to launderettes in response to a 5%-10% price increase of flat linen rental and maintenance services.<sup>166</sup> [...] noted that while using wash-only launderettes is a potential option, there would be concerns whether wash-only launderettes would be able to ensure daily delivery of linen.<sup>167</sup>
- 3.85 Hence, for the reasons set out in paragraphs 3.83 to 3.84 above, the Commission is of the view that the supply of flat linen maintenance-only services should not be included in the same product market as the supply of outsourced flat linen rental and maintenance services to customers in the hospitality sector.

<sup>&</sup>lt;sup>163</sup> Please see Berendsen's response to question 20 of the First RFI and Kings Laundry's response to question 23 of the First RFI.

<sup>&</sup>lt;sup>164</sup> See also Customers' views on viability of using wash-only launderettes summarised in paragraph 3.23 above.

<sup>&</sup>lt;sup>165</sup> See also Customers' views on viability of using wash-only launderettes summarised in paragraph 3.23 above.

<sup>&</sup>lt;sup>166</sup> See paragraph 10.28 below.

<sup>&</sup>lt;sup>167</sup> [...] response to question 12 of the Questionnaire to Customers.



# Does the supply of rental and maintenance of workwear constitute a distinct product market from rental and maintenance of flat linen?

3.86 The Commission has considered whether the product market encompassing the rental and maintenance of flat linen items should be broadened to include other textile products such as workwear. Berendsen provides standard workwear rental and maintenance services<sup>168</sup> and protective workwear rental and maintenance services<sup>169</sup> while Kings Laundry provides standard workwear rental and maintenance services to customers in the healthcare sector.

## Healthcare market

- 3.87 The Parties noted that healthcare customers typically avail of a broad service package that includes flat linen, surgical scrub suits and other miscellaneous items. The Commission's investigation supported the Parties' views. For example, the majority of the Customers in the healthcare sector that responded to the Questionnaire to Customers noted that they purchase flat linen, surgical scrub suits and other miscellaneous items from the same supplier. ITTs issued by healthcare customers indicate that scrub suits are frequently included in the same lot as flat linen.
- 3.88 The Commission notes that linen laundries process scrub suits and other miscellaneous items in their facilities catering for flat linen. Some specific equipment is required for processing these items (e.g., ironing machines for scrub suits). In contrast, processing workwear items (both standard and protective workwear items) used in other sectors requires substantially different equipment and Berendsen has separate facilities catering for processing these items. In general, Competitors were of the view that rental and maintenance of workwear is a distinct business from rental and maintenance of flat linen due to the different equipment required for processing of items.
- 3.89 The Commission is of the view that the provision of workwear rental and maintenance services belongs to a distinct product market. The exception to this is scrub suits and other miscellaneous items provided by linen laundries to customers in the healthcare sector<sup>170</sup> that are generally purchased as a broader service package together with flat linen. Therefore, the Commission is of the view that the relevant product market includes rental and maintenance of scrub suits and other miscellaneous services provided by linen laundries to customers in addition to customers' flat linen requirements.

## Hospitality market

3.90 Only one Customer in the hospitality sector that responded to the Questionnaire to Customers ([...]) indicated that it purchases standard workwear (i.e., kitchen wear) rental and maintenance services from the supplier of outsourced flat linen rental and maintenance services. The Commission also notes that contracts with customers provided by the Parties indicate that the scope of contracted services typically includes

<sup>&</sup>lt;sup>168</sup> Standard workwear is supplied to customers in healthcare and hospitality sectors and includes items such as chefs wear, waiter wear, catering wear and surgical scrub suits. In the healthcare sector, Berendsen also processes other miscellaneous items (e.g., patient gowns and slings).

<sup>&</sup>lt;sup>169</sup> Protective workwear is supplied to industrial customers (e.g., food and beverage companies, utilities etc.) and includes textile products and technical products (protective glasses, gloves, helmets etc.).

<sup>&</sup>lt;sup>170</sup> Items such as patient gowns.



flat linen only. Customers from other sectors purchasing workwear rental and maintenance services (e.g., (Intel and Pfizer) do not require outsourced flat linen rental and maintenance services.

3.91 The Commission is of the view that the provision of workwear rental and maintenance services belongs to a distinct product market.

Overall conclusion on the scope of the relevant product markets

- 3.92 In order to assess and determine whether the Proposed Transaction might result in an SLC, the Commission has defined relevant markets as follows:
  - a) there are separate markets for the provision of outsourced flat linen rental and maintenance services to healthcare customers and to hospitality customers;
  - b) self-supply of laundry services via OPL is not included in the healthcare or the hospitality market for the provision of flat linen rental and maintenance services;
  - c) the supply of flat linen maintenance-only services is not included in the healthcare or the hospitality market for the provision of outsourced flat linen rental and maintenance services; and
  - d) the provision of workwear rental and maintenance services (with the exception of scrub suits and other miscellaneous items provided by linen laundries to healthcare customers) is not included in the healthcare or the hospitality market for the provision of outsourced flat linen rental and maintenance services.

## Relevant geographic markets

3.93 Taking into account the views expressed by the Parties, the Commission considered whether: (i) the geographic scope should be narrower than the State (discussed in paragraphs 3.94 to 3.103 below); and (ii) the geographic scope should be broader than the State (i.e., include Northern Ireland) (discussed in paragraphs 3.104 to 3.110 below) for each of the relevant product markets.

## Local/Regional scope

3.94 In relation to the healthcare market, the Commission notes that all 3 linen laundries that provide flat linen rental and maintenance services to healthcare customers on a significant scale (i.e., the Parties and [...]) provide a [...]. [...] told the Commission that it breaks geographic regions into separate lots in its ITTs.<sup>171</sup> The OGP does not have geographic differentiation in its RFT for admission to the Framework Contract,<sup>172</sup> but retains the option to run the follow-up mini-competitions of a geographical basis. The Commission observed that both Berendsen and [...] have tended to participate in the HSE's tenders irrespective of geographic location. Customer location maps provided by

<sup>&</sup>lt;sup>171</sup> See the Commission's minutes of meeting with [...].

<sup>&</sup>lt;sup>172</sup> See paragraph 2.12 for a description of the OGP Framework.

the Parties indicate that there are a significant number of healthcare customers located more than [...] kilometres away from the Parties' facilities catering for these customers.

- 3.95 In relation to the hospitality market, the Commission's investigation provided mixed evidence on whether the geographic scope of the relevant product market is regional or national in scope.
- 3.96 Some of the Customers that responded to the Questionnaire to Customers indicated that they purchase outsourced flat linen rental and maintenance services on a regional basis from several linen laundries (e.g. [...]).
- 3.97 The Commission notes that the activities of [...] are limited to areas that are in relative proximity to their facilities.<sup>173</sup> In general, the Commission observed evidence indicating that distribution costs do affect the competitiveness of linen laundries. In this regard, [...] noted that linen laundries that are able to refill their vehicles carrying flat linen more than once per day have a distinct economic advantage compared to linen companies that refill their vehicles once a day.<sup>174</sup> The Commission's investigation revealed that there are some variances in competitive conditions across the regions in the State with hospitality customers in some regions. However, the Commission considers that, in the supply of outsourced flat linen rental and maintenance services, regions within the State do not exhibit precise and stable boundaries in which conditions of competition are sufficiently homogenous and which can be distinguished from neighbouring areas because the conditions of competition are appreciably different.
- 3.98 There is considerable evidence indicating that the geographic market for the supply of outsourced flat linen rental and maintenance services to hospitality customers is national in scope.
- 3.99 First, the majority of Customers from the hospitality sector that responded to the Questionnaire to Customers (80%) indicated that the distance between their premises and the linen laundry's facility where flat linen is taken for maintenance is not a very important factor when selecting a supplier as long as the service is offered on a regular basis and the linen laundry is able to adapt quickly to the customers' needs.<sup>175</sup> In this regard, Kings Laundry indicated that [...] out of [...] ([...]%) hotel groups seek one supplier of outsourced flat linen rental and maintenance services across all of their sites (i.e., a national supplier).<sup>176</sup> While the Commission is of the view that this group of hospitality customers does not form a distinct product market, the Commission recognises that these customers may have a more limited set of options when compared to individual hotels.
- 3.100 Second, Berendsen, Kings Laundry and [...] service customers and participate in tenders for hospitality customers located across the State. Moreover, Kings Laundry and [...] were competing with Berendsen on a national basis at the time when each of them had facilities in only one geographic location. The Commission also notes that for long

<sup>&</sup>lt;sup>173</sup> Also see Table 7 below which sets out linen laundries' turnover in 2018 by geographic regions.

 $<sup>^{\</sup>rm 174}$  [...] response to question 9 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>175</sup> See paragraph 10.19 below.

<sup>&</sup>lt;sup>176</sup> Kings Laundry's response to question 20 of the First RFI.



distances, linen laundries may install depots across the State that serve as hubs to service their customer needs.<sup>177</sup>

- 3.101 Third, Berendsen noted that distribution costs accounted for [...]% of its totals costs of sales and did not consider it to be the most important factor in determining the price of flat linen rental and maintenance services<sup>178</sup>. Berendsen highlighted that [...] is the primary determinant of distribution costs due to the fact that the majority of transportation costs are related to [...]. Berendsen further noted that driving time is not necessarily reflective of the distance travelled as some customer locations (e.g., Dublin City centre) have high traffic volume, thus leading to a longer driving time than would be reflected by its distance from Berendsen's facilities.
- 3.102 Fourth, [...] confirmed that it has expanded its operations to [...], thus indicating that expanding activities without building an additional facility is possible.<sup>179</sup>
- 3.103 Having regard to the evidence presented in paragraphs 3.94 to 3.102 above, the Commission, on a conservative basis, considers that the scope of the relevant geographic markets is not narrower than the State. Whilst distribution costs do affect the competitiveness of suppliers over long distances, this effect is not strong enough to preclude linen companies from competing throughout the State. However, the location of linen laundries' facilities is further considered, where relevant, in the competitive assessment.

## All island scope

- 3.104 The Commission did not observe any significant evidence that would merit the inclusion of Northern Ireland in the scope of the relevant geographic market in either the healthcare or hospitality sector.
- 3.105 In the healthcare sector, the Commission did not observe any evidence that linen laundries based in Northern Ireland are providing services to healthcare customers in the State. In this regard, [...] noted that the last time it participated in any HSE tender was over ten years ago.<sup>180</sup>
- 3.106 In the hospitality sector, the Commission notes that both [...] provide services to hospitality customers in the State from their facilities in Northern Ireland. However, their activities in the State are relatively minor and in general limited only to Dublin and/or counties bordering Northern Ireland. Neither of these linen laundries indicated to the Commission that they intend to make significant expansion by, for example, opening facilities in the State. In this regard, the Commission notes that, according to [...] was invited to participate in its tender, but did not submit a bid.<sup>181</sup>

<sup>&</sup>lt;sup>177</sup> The Parties indicated that for longer distances, linen laundries can utilise a system known in the industry as 'trunking'. This means that textiles, collected by the vans in the depot catchment area, are transported in bulk over night to a laundry facility, handled there, and then trunked back via Heavy Goods Vehicle trucks to the depot for them to be distributed back to local customers.

 $<sup>^{\</sup>rm 178}$  Berendsen's response to Question 16 of the First RFI.

 $<sup>^{\</sup>rm 179}$  See the Commission's record of a phone call with [...] dated 16 October 2018.

<sup>&</sup>lt;sup>180</sup> See the Commission's record of a phone call with [...] dated 20 February 2019.

<sup>&</sup>lt;sup>181</sup> This was also confirmed by [...] during the Commission's telephone call with [...] on 20 February 2019.



- 3.107 The Amárach Survey indicated that just 1% of respondents from the hospitality sector purchase outsourced flat linen rental and maintenance services from linen laundries based in Northern Ireland.<sup>182</sup>
- 3.108 The Commission's market investigation also indicated that some of the customers having premises in Northern Ireland (e.g., [...]) purchase flat linen rental and maintenance services for these premises from linen laundries operating in Northern Ireland rather than linen laundries operating in the State. In instances where customers with premises in Northern Ireland purchase flat linen rental and maintenance services from a single linen laundry, the Commission observed evidence that services to premises located in Northern Ireland are outsourced to linen laundries that have a larger footprint in Northern Ireland. For example, in the Merger Notification Form Berendsen indicated that it [...]<sup>184</sup>
- 3.109 Finally, Berendsen in its internal documents noted that [...]<sup>185</sup> and illustrated that [...].
- 3.110 Thus, the Commission is of the view that the scope of the relevant geographic market should not be broadened to include Northern Ireland in either of the relevant product markets.

## Overall conclusion on the scope of the relevant geographic markets

3.111 Having regard to the evidence presented in paragraphs 3.94 to 3.110 above, the Commission considers that for the purpose of this Determination, the markets for the supply of flat linen rental and maintenance services to: (i) healthcare customers; and (ii) hospitality customers, are national in scope.

## Overall conclusion on relevant market definition

- 3.112 Having regard to the evidence available to it, the Commission considers that the relevant markets for the competitive assessment of the Proposed Transaction are:
  - the market for the outsourced supply of flat linen rental and maintenance services to healthcare customers<sup>186</sup> in the State; ("the healthcare market"); and
  - (ii) the market for the outsourced supply of flat linen rental and maintenance services to hospitality customers<sup>187</sup> in the State ("the hospitality market").

<sup>&</sup>lt;sup>182</sup> See slide 11 of the Amárach Survey.

<sup>&</sup>lt;sup>183</sup> Footnote 6 of the Merger Notification Form.

<sup>&</sup>lt;sup>184</sup> Footnote 9 of the Merger Notification Form.

<sup>&</sup>lt;sup>185</sup> See document titled [...] submitted by Berendsen in response to Question 9 of the First RFI.

<sup>&</sup>lt;sup>186</sup> As defined in paragraph 3.92.

<sup>&</sup>lt;sup>187</sup> As defined in paragraph 3.92.

## 4. COMPETITIVE ASSESSMENT – UNILATERAL EFFECTS

## Counterfactual

Coimisiún um Iomaíocht agus

Cosaint Tomhaltóirí

- 4.1 The counterfactual refers to the state of competition without the merger, and provides the point of comparison for assessing the competitive effects that arise from a merger. The Commission generally adopts the prevailing conditions of competition as the counterfactual against which it assesses the impact of the merger.<sup>188</sup> Identifying the relevant counterfactual is forward-looking and necessarily involves judgement on the part of the Commission.<sup>189</sup>
- 4.2 The counterfactual has to be plausible in light of available evidence. For example, the ECJ held that the available evidence must support the EC's conclusion that "..if such a [merger] decision were not adopted, the economic development envisaged by it [EC] would be plausible."<sup>190</sup> The competitive assessment then asks whether the merger compared to the counterfactual will lead to an SLC.
- 4.3 The Commission's counterfactual is that, absent the merger, Berendsen and Kings Laundry will continue to compete for the supply of outsourced flat linen rental and maintenance services to customers in both the hospitality and healthcare markets, and that Kings Laundry is likely to expand its activities in the healthcare market, including supply to public and private hospitals and to nursing homes. Under the prevailing conditions of competition, there are currently three main suppliers<sup>191</sup> of outsourced flat linen rental and maintenance services in the healthcare market, namely: Berendsen, Celtic Linen and Kings Laundry. The Commission notes that, in the period 2014-2018, Kings Laundry has increased its market share in both the healthcare market and the hospitality market.<sup>192</sup> The Commission's judgement is that, absent the merger, and based on evidence from its investigation, the most plausible counterfactual is that Kings Laundry will continue to expand in both the hospitality and healthcare markets.
- 4.4 The Parties, in their written and oral submissions, disagreed with the Commission's counterfactual insofar as it related to the Commission's finding that, absent the merger, Kings Laundry will continue to expand in the healthcare market, particularly by supplying flat linen rental and maintenance services to public hospitals.
- 4.5 Key arguments put forward by the Parties were:
  - a) Kings Laundry's alleged intention to expand in healthcare is unconvincing,<sup>193</sup> and

<sup>&</sup>lt;sup>188</sup> Paragraph 1.12 of "Guidelines for Merger Analysis" adopted by the Commission on 31 October 2014 (the "Commission's Merger Guidelines").

<sup>&</sup>lt;sup>189</sup> *Ibid*, paragraph 1.14.

<sup>&</sup>lt;sup>190</sup> Case C-12/03 P Commission v Tetra Laval EU:C:2005:87, paragraph 44.

<sup>&</sup>lt;sup>191</sup> Supply from other linen laundries accounts for less than 2% of the market

<sup>&</sup>lt;sup>192</sup> Kings Laundry's market share has increased between 2014 and 2018 from [0-10]% to [10-20]% in the healthcare market, and from [20-30]% to [30-40]% in the hospitality market.

<sup>&</sup>lt;sup>193</sup> Parties' Response, paragraph 4.2.5.1.



- b) The Commission fails to establish that Kings Laundry will overcome barriers to expansion in healthcare.<sup>194</sup>
- 4.6 The Commission has considered in detail each of these two lines of argument advanced by the Parties.

#### Kings Laundry's intention to expand in the healthcare market

- 4.7 The Parties' claimed that, in response to the RFIs issued by the Commission, and in further submissions, Kings Laundry has stated that it had "[...]".<sup>195</sup> In support of this argument, the Parties claimed that Kings Laundry's statement in its [...] was "loose talk"<sup>196</sup> and "simply not credible".<sup>197</sup>
- 4.8 The Parties noted that Kings Laundry stated in a meeting with the Commission that "[...]". In the same meeting, Kings Laundry stated that they had "[...]".  $^{198}$
- 4.9 The Commission has fully considered these points in the light of all the evidence obtained during the market investigation. The Commission recognises that Kings Laundry has made statements post-merger notification that it does not intend to expand in healthcare. As set out in the Commission's Merger Guidelines, the Commission expects the merging parties to substantiate "..any counterfactual they propose with objective evidence.....such evidence and analysis should obviously be consistent with the parties own internal <u>pre-merger</u> assessments of the likely counterfactual" (emphasis added).<sup>199</sup> Furthermore, the Commission notes that it "...will give <u>much greater weight to evidence that pre-dates the announcement of the merger</u> under review in comparison to post-merger announcement evidence, because the behaviour of the merging parties vis-a-vis each other and third parties (i.e. customers, competitors and suppliers) is likely to be heavily influenced by the announcement of the merger" (emphasis added).<sup>200</sup>
- 4.10 The Commission's approach to the treatment of evidence is in line with guidelines produced by the International Competition Network, which state that:

"Pre-existing documents that were prepared before the merger was under consideration are especially useful. Pre-existing documents are valued by agencies for the opportunity to see how the parties acted and how they viewed competition and markets before they had the merger in mind." <sup>201</sup>

"It may be helpful to request from undertakings involved in the merger (or indeed other firms active in the market) their commercial strategies and other internal documents such as internal communications, public statements, and studies on consumer preferences, market research, advertising plans, general marketing plans or business plans. These may indicate which products the

<sup>&</sup>lt;sup>194</sup> Parties' Response, paragraph 4.2.5.2.

<sup>&</sup>lt;sup>195</sup> Parties' Response, paragraph 4.2.8.

<sup>&</sup>lt;sup>196</sup> Parties' Response, paragraph 4.2.7.2.

<sup>&</sup>lt;sup>197</sup> Frontier Economics Report, Paragraph 3.8.b.

<sup>&</sup>lt;sup>198</sup> Parties' Response, paragraphs 4.2.11.1 to 4.2.11.2.

<sup>&</sup>lt;sup>199</sup> Paragraph 1.15 of the Commission's Merger Guidelines.

<sup>&</sup>lt;sup>200</sup> Paragraph 7.11 of the Commission's Merger Guidelines.

<sup>&</sup>lt;sup>201</sup> ICN (2005), *Investigative Techniques Handbook for Merger Review*, p.39. This may be accessed at <u>www.internationalcompetitionnetwork.org</u>.

undertakings believe to be the closest substitute to their own products and may also provide information on which companies they consider to be their competitors."  $^{\rm 202}$ 

- 4.11 The Commission considers that Kings Laundry's post-merger notification claims that it does not intend to expand in healthcare are contradicted by pre-merger notification statements made by Berendsen. The Commission's review of Berendsen's internal documents shows that Berendsen [...]. Berendsen articulated that [...] and Berendsen referred to the Proposed Acquisition as a "[...]"<sup>203</sup> According to the Parties' written submission, the Commission is incorrect to assume that this quote refers to the healthcare market. However, in another document Berendsen also noted that [...].<sup>204</sup> In their post-merger notification written and oral submissions, the Parties suggested that the Commission should not give weight to the comments in these internal documents and comments made by [...] in an email to [...]. According to the Parties, [...].<sup>205</sup> The Commission considers that the pre-merger notification statements in the internal documents of Berendsen and comments made by one of its senior officials are self-explanatory and cannot be dismissed by post-merger notification statements that attempt to give those statements different meaning or significance.
- 4.12 The Commission also considers that Kings Laundry's behaviour in the healthcare market and pre-merger statements concerning its healthcare business contradict its postmerger statements. Kings Laundry continues to grow its market share in the healthcare market. In addition, Kings Laundry stated in its response to the Commission's RFI that it [...].<sup>206</sup> The Commission recognises that this consideration was at a preliminary stage, but notes that Kings Laundry had gone as far as [...]. In addition, and as noted in the Parties' submission, in responding to an ITT issued by [...] in 2018, Kings Laundry stated that: [...].<sup>207</sup> [...] confirmed that, in its view, Kings Laundry planned to build a new laundry facility specifically for healthcare customers. In this regard, [...] noted that while they did not recall seeing specific plans, their view was that it was a genuine plan and not merely an aspiration.<sup>208</sup> When asked by the Commission during the Parties' oral submissions what it would have done had it won the tender for Mater Private Hospital, Kings Laundry responded that it would have entered into and fulfilled the contract. In the Commission's view, these examples of behaviour and pre-merger statements are clear indications of an intention to expand in healthcare.
- 4.13 The Commission considers that Kings Laundry's pre-merger notification statements and its performance, taken together with Berendsen's pre-merger notification statements and those of the [...], provide credible evidence to support the Commission's view that, absent the merger, the most plausible counterfactual is that Kings Laundry will continue to expand in both the healthcare and the hospitality markets. The Commission has seen no pre-merger notification evidence that would suggest that Kings Laundry had decided

<sup>&</sup>lt;sup>202</sup> ICN (2006), *Merger Guidelines Workbook*, Prepared for the Fifth Annual ICN Conference in Cape Town, April 2006, p.21 (emphasis in original).

<sup>&</sup>lt;sup>203</sup> Slide 24 of the Document titled [...], dated 23 March 2018 submitted by Berendsen in Appendix G of the Merger Notification Form.

<sup>&</sup>lt;sup>204</sup> See internal email between Berendsen and Elis titled [...], dated 2 February 2018 provided by Berendsen in response to Question 1 of the First RFI.

<sup>&</sup>lt;sup>205</sup> Parties' Response, paragraph 4.2.45.

<sup>&</sup>lt;sup>206</sup> Kings Laundry's response to Question 14 of the RFI. In responding to this question Kings Laundry noted that it [...].

<sup>&</sup>lt;sup>207</sup> This was provided by Kings Laundry to the Commission in response to Question 27 of the First RFI. Furthermore, on 27 February 2019 Kings Laundry indicated that it was the [...].

<sup>&</sup>lt;sup>208</sup> See email from [...] dated 9 May 2019.



to change from a growth strategy in healthcare to one where it was not interested in expanding.

#### Barriers to entry and expansion

- 4.14 The Commission notes at the outset that there is no disagreement between the Commission and the Parties that Kings Laundry is currently active in the healthcare market. In their written and oral submissions, the Parties claimed that the Commission's relevant counterfactual "does not adequately address how Kings Laundry will overcome the barriers to expansion in the healthcare market.... In particular, the Commission has not sufficiently demonstrated that Kings Laundry (i) would implement the necessary accreditations in a timely and sufficient manner; and (ii) has the requisite prior experience to expand in healthcare".<sup>209</sup> Furthermore, the Parties claimed that the Commission has not applied the same criteria as it applied to Kings Laundry to evaluate the potential entry or expansion of other competitors, notably OCL.<sup>210</sup>
- 4.15 The Commission recognises that there are a number of barriers to entry and expansion in the healthcare market. These include financial barriers, particularly for new entry to the healthcare market, and non-financial barriers such as accreditation and experience. Barriers to entry and expansion are discussed in more detail in the assessment of competitive constraints below<sup>211</sup>. For the purposes of establishing the counterfactual against which the impact of the merger can be measured, the relevant consideration is whether the Commission is correct to view Kings Laundry as an important competitor in the identified relevant markets.
- 4.16 While Kings Laundry currently services healthcare customers from its Dublin plant, it has stated that its facility in Cork can be upgraded to cater for private healthcare customers within [...] at an additional cost of €[...] to €[...].<sup>212</sup> In the Commission's view, the ability to upgrade within this timeframe, and with this level of investment would be considered as an ability to expand in a timely and sufficient manner.<sup>213</sup> Based on evidence from site visits and discussions with third parties, the Commission's understanding is that, once a laundry has put in place the infrastructure needed to process laundry from private hospital customers, there is no additional physical facility required to supply public hospitals. The additional requirements for public hospitals are those associated with meeting tender requirements for accreditation and being able to meet other tender requirements such as demonstrating relevant experience.
- 4.17 The Commission agrees that healthcare customers require suppliers of flat linen rental and maintenance services to adhere to: (i) hygiene standards that specify the level of quality<sup>214</sup> required for the processing of flat linen in terms of bio-contamination control;

<sup>212</sup> Kings Laundry's response to Question 27 of the Second RFI. The Commission notes that [...].

<sup>&</sup>lt;sup>209</sup> Parties' Response, paragraph 4.2.13.

<sup>&</sup>lt;sup>210</sup> Parties' Response, paragraph 4.2.22.

<sup>&</sup>lt;sup>211</sup> Paragraphs 4.188 to 4.195 (hospitality market) and 4.332 to 4.352 (healthcare market).

<sup>&</sup>lt;sup>213</sup> According to paragraph 6.5 of the Commission's Merger Guidelines, "..entry that is effective within two years is normally considered timely..". Paragraph 6.6 states that "The likelihood of entry post-merger generally depends on the profitability of entering the market.". Paragraph 6.8 notes that "For entry to be sufficient, it must be likely that incumbents would lose significant sales to new entrants."

<sup>&</sup>lt;sup>214</sup> The Commission understands that there are two commonly cited standards: (i) NHS Executive HSG(95)18. <u>"Hospital laundry arrangements for used and infected linen</u>", dated 21 April 1995; and (ii) and BS EN 14065:2002, <u>"Textiles – Laundry Processed Textiles – Biocontamination Control System</u>", dated 6 December 2002 (known in the industry as the risk analysis bio-contamination control ("RABC") standard). In 2016 this standard was superseded by BS EN 14065:2016.



and, (ii) certain linen<sup>215</sup> processing guidelines. For instance, the tender document for the supply of flat linen rental and maintenance services to healthcare customers in respect of St. James' Hospital stated the following in relation quality assurance:

"Tenderers must demonstrate satisfactory internal quality assurance mechanisms in place and accreditation attained. Tenderers must also demonstrate external quality assurance and accreditation – ISO, RABC Standard EN 14065 and HSG 95(18) or equivalent".<sup>216</sup>

- 4.18 Kings Laundry has experience of addressing hygiene standards and linen processing guidelines through its work for private hospitals. According to Kings Laundry: "[...]". <sup>217</sup>
- 4.19 The Commission recognises that public hospitals generally have more formal requirements than private hospitals, particularly in specifying quality standards such as the RABC. The Commission recognises that, to supply public hospitals, Kings Laundry would need to apply for additional formal accreditation. According to the NSAI, such accreditation usually takes 3-4 months from application to registration. The Commission notes also the views of a competitor that implementing these accreditations requires considerable input and effort.<sup>218</sup> The Commission is of the view that the administrative effort involved in applying for accreditation, and effort involved in implementing and maintaining the required standards are far from insurmountable for Kings Laundry. The Commission notes that Kings Laundry has stated that, in providing flat linen services to private hospitals, it is already meeting many of the requirements in the guidelines. Indeed, in a meeting with the Commission, Kings Laundry stated that: "The cost of RABC accreditation is not significant and can be obtained in a short timeframe. Obtaining ISO accreditation is significantly more costly and lengthy".<sup>219</sup> Given that, as noted in paragraph 4.18 above, King's Laundry has already obtained what it considers to be the more costly and lengthy ISO accreditation, and already has experience of implementing and adhering to hygiene standards, the Commission considers that it is reasonable to assume that, absent the merger, Kings Laundry will be able to overcome the effort required to comply with public hospitals' requirements for accreditation.
- 4.20 The Commission recognises the importance of previous experience in winning tenders for hospitals. For example, the ITT from St Vincent's University Hospital asks tenderers to demonstrate: "Your company's skills, efficiency, experience and reliability in delivering similar supplies and services in a comparable operational environment. Firm preference is for an acute hospital environment..... How the skills and experience gained from those contracts can be transferred to the Contracting Authority's contract."<sup>220</sup>
- 4.21 Similarly, when the HSE issued a tender for a Framework Contract (issued in late 2018, later withdrawn) it emphasised the requirement for bidders to have a "an experienced Senior Management Team capable of directing and managing the delivery of the

<sup>&</sup>lt;sup>215</sup> For example, see Health Service Executive's ("HSE") guidelines for managing laundry and linen in healthcare facilities which can be accessed at: <u>https://www.hse.ie/eng/about/who/healthwellbeing/infectcont/sth/gl/ipcc-guidelines-section-9.pdf</u>.

<sup>&</sup>lt;sup>216</sup> For more information, see section 12.3 "Selection criteria" of the St. James Hospital ITT document.

<sup>&</sup>lt;sup>217</sup> Kings Laundry's response to Question 21 of the First RFI.

<sup>&</sup>lt;sup>218</sup> Meeting Minutes with [...]. In respect of the RABC standard specifically, the NSAI informed the Commission that such accreditation would usually take about 3-4 months, i.e., from "application to registration" to obtain. The NSAI also noted that this depends on non-conformances which extend the timeline. See NSAI's email dated 8 February 2019.

<sup>&</sup>lt;sup>219</sup> Meeting minute 1 February 2019.

<sup>&</sup>lt;sup>220</sup> Appendix 8, St Vincent's University Hospital ITT, 3 May 2018.



Services" and stated that experience of providing "healthcare laundry linen services" is essential: "Tenderers must demonstrate that they have successfully delivered contracts of a similar nature to the HSE's requirements during the last three years by providing a list of its previous significant contracts delivered over the last three years.". This was to be supported by references.<sup>221</sup>

4.22 The OGP's tender for a Framework Agreement<sup>222</sup> (issued in June 2019) requires bidders to provide "summary details of three contracts in which they have successfully delivered Services of a similar nature to the Services required....within the last three years". Contracts must each have "...a minimum value of €144,000 per year of contracts", and one of the contracts "..must be in place or have been in place for more than 2 years." The Commission notes that the requirement for experience constitutes a barrier to entry for any linen laundry that is not already active in supplying hospitals, and that this would be a considerable barrier for all laundries with the exception of Berendsen, Celtic Linen and Kings Laundry, who all meet the qualification criteria for experience in delivering healthcare contracts.

## Potential market entry

- 4.23 In their written and oral submissions, the Parties claimed that, in developing its counterfactual, the Commission has not evaluated potential market entry and expansion by competitors other than Kings Laundry, and in particular has 'disregarded evidence of OCL's intention to expand in healthcare'.<sup>223</sup>
- 4.24 The Commission notes that the Parties' submission in respect of OCL is not relevant for the purposes of establishing the relevant counterfactual for the purpose of assessing the likely competitive effects of the Proposed Transaction. The question of whether or not OCL and other linen laundries active in the hospitality market are likely to enter the healthcare market is discussed in more detail in discussing unilateral effects in paragraphs 4.332 to 4.352 below.

## Conclusion on counterfactual

- 4.25 The Commission's considers that, pre-merger, Kings Laundry had the ability and intention to continue expanding in both the hospitality and healthcare markets, and had the potential to supply all types of healthcare customers, including public hospitals. The Commission bases this view on evidence provided by Kings Laundry's actions and statements pre-merger notification, on Berendsen's views expressed in internal documents, and notes that these views were supported by a healthcare customer. The Parties' submissions suggesting a counterfactual that Kings Laundry will change strategy and stop expanding in healthcare is not supported by evidence other than post-merger notification statements.
- 4.26 In light of the above, the Commission concludes that the most plausible and relevant counterfactual is that, absent the merger, Berendsen and Kings Laundry will continue to compete for the supply of flat linen rental and maintenance services to customers in

<sup>222</sup> The OGP, Request for Tender dated 4/06/19 to establish a Multi-Supplier Framework Agreement for the provision of and management of laundry and hire of linen, curtains, workwear and dust mat hire services.

<sup>221</sup> https://ted.europa.eu/udl?uri=TED:NOTICE:429143-2018:TEXT:EN:HTML&src=0

<sup>&</sup>lt;sup>223</sup> Parties' Response, paragraph 4.2.22.



both the hospitality and healthcare markets, and that Kings Laundry is likely to expand its activities in the healthcare market.

## The nature of competition

- 4.27 The Commission's approach to the assessment of horizontal mergers is set out in the Merger Guidelines.<sup>224</sup> The analysis of a proposed merger may cover unilateral and coordinated effects. The analysis considers whether product markets are homogenous or differentiated.
- 4.28 Before analysing each of the relevant healthcare and hospitality markets, the Commission sets out in this section its overall approach to considering the nature of competition in both markets.

## Bidding markets

- 4.29 The Commission notes that competition in the healthcare and hospitality markets takes place for contracts (which may be formally tendered or informally negotiated on the basis of requests for proposals) rather than via posted prices. In essence, both the healthcare and hospitality markets are bidding markets.
- 4.30 According to the Merger Guidelines,

"In some markets, particularly bidding markets, the number of possible suppliers can influence the intensity of competition. For example, in tendering processes the greater the number of firms able to tender for the supply of products or services, the more likely it is that there will be intense competition."<sup>225</sup>

- 4.31 The OECD states that: "Merger analysis is not significantly changed by the existence of a bidding process. Markets where bidding processes are used are subject to similar economic forces as those in other markets. As in any merger analysis, it is important to understand the competitive constraints to which the merging parties are subject and to ground the choice of economic model in an analysis of the factual circumstances."<sup>226</sup>
- 4.32 The Commission notes that both the Merger Guidelines and the OECD emphasise the need to fully and widely consider competitive effects in all merger analyses. This is further confirmed in work carried out for the CMA's predecessor the OFT, which states that: "...bidding processes are unlikely to create significantly different issues to those that may arise in a more traditional market. In particular, there is no 'bright-line' distinction between a 'bidding market' and a 'conventional market'; bidding processes are not all the same and case-specific analysis will always be needed to understand how competition works; and there is no 'ideal' bidding process that works well in all circumstances, especially not with few bidders."<sup>227</sup>
- 4.33 This sets the context within which the Commission has considered the healthcare and hospitality markets as bidding markets. Turning to specific aspects of the nature of

<sup>&</sup>lt;sup>224</sup> Merger Guidelines, paragraphs 4.1 to 4.40.

<sup>&</sup>lt;sup>225</sup> Merger Guidelines, paragraph 4.39.

<sup>&</sup>lt;sup>226</sup> OECD, Policy Roundtables, 'Competition in Bidding Markets' 2006.

<sup>&</sup>lt;sup>227</sup> DotEcon for the OFT, 'Markets with Bidding Processes', 2007.

bidding markets, Klemperer<sup>228</sup> notes that *"The existence of a bidding market is commonly cited as a reason to tolerate the creation and maintenance of highly concentrated markets."* In common with the Merger Guidelines and the OECD approach, Klemperer warns against the contention that, in bidding markets, ordinary competition concerns do not apply. In particular, he cites a number of fallacies often put forward to justify the contention that in bidding markets, market share does not imply market power, and that the existence of two firms is enough to imply perfect competition (or even that one firm may be enough).

- 4.34 The OECD<sup>229</sup> notes that "ideal" bidding markets would typically exhibit the following characteristics:
  - a) competition is "winner takes all," so each supplier either wins all or none of the contract;
  - b) competition for contracts is "lumpy," so that each contest is large (and, therefore, important) for a supplier relative to its total sales over a period;
  - c) competition begins afresh for each contract (and for each customer), such that when tenders are repeated, the outcome of a previous tender does not impact on the outcome of another tender (this broadly corresponds to tenders taking place infrequently); and
  - d) market entry is easy.
- 4.35 In an "ideal" bidding market exhibiting characteristics (a) to (d), two identical firms would fit the standard Bertrand model of competition in a homogenous product market with many consumers. In this model, assuming constant marginal costs and no capacity constraints, two identical firms would theoretically be sufficient to achieve a perfectly competitive outcome (i.e., prices being equal to marginal cost), and historic market shares which would essentially reflect the random choice of customers amongst identical suppliers would not imply future market success or indeed market power.

## Views of the Parties

4.36 In the Parties' written and oral submissions, the Parties argued that both hospitality and healthcare markets are bidding markets, and exhibit the following characteristics:

"a bidding process is used by customers to choose a supplier (either by formal tendering or the informal gathering of quotes);

the winner of each tender serves the entirety of the customer's needs which are up for award. Therefore, there is no smooth trade-off between the price offered and the quantity sold. Any customer losses are not marginal declines in turnover, but rather losses of entire contracts.

<sup>&</sup>lt;sup>228</sup> P. Klemperer (2005), "Bidding Markets", Competition Commission discussion paper.

<sup>&</sup>lt;sup>229</sup> OECD, Policy Roundtables, ibid.

customers are not locked into relationships with suppliers, so the outcome of one contest does not determine the outcome of a subsequent contest.

Because each tender is a separate competition, the prices offered in one bid do not affect the prices offered in other bids".<sup>230</sup>

- 4.37 The Parties then proposed that "... in the context of bidding markets, a customer issues a contract for a specific set of goods or services, to which suppliers respond with the best price at which they can provide this product. The scope for differentiation with respect to any individual customer, is therefore limited. This reduction in the scope for firms to differentiate themselves intensifies the degree of competitive pressure they face. In fact, the standard economic result is that with no differentiation at all between firms, only two firms are required to produce competitive outcomes in the market".<sup>231</sup> And in a footnote: "Specifically, two symmetric firms competing in [sic] prices for a homogenous product will result in prices being set equal to marginal costs."<sup>232</sup>
- 4.38 The Parties defined particular features of such bidding markets and claimed that in markets that exhibit these features *"market shares are of limited, if any, or no relevance they simply show past success rather than whether firms are viable options in the future for customers. A high post-merger market share for the Parties does not equate to the possession of market power".*<sup>233</sup> The Parties argued that in 'bidding markets' even a limited number of rivals can produce highly competitive outcomes, and that there would only be a competition impact if the merging parties were first and second ranked by the buyer (e.g., on the basis of providing the offer with the lowest and second lowest price), as only in this case they would constrain each other.<sup>234</sup>

## Views of the Commission

- 4.39 The Commission considers that the dynamics of the healthcare and hospitality markets indicate that they are not "ideal" bidding markets. An "ideal" bidding market relies on suppliers selling undifferentiated products, and the customer choosing the lowest-priced bidder, whose bid is only constrained by the second-lowest bidder in the sense that the winning bid must, by definition, be lower than the second lowest bid. In this scenario, eliminating a competitor through a merger would not necessarily reduce the constraints faced by the remaining competitors except in the case where the merging parties are the first and second ranked bidder.
- 4.40 The Commission has considered the features of the healthcare and hospitality markets against each of the characteristics of an "ideal" bidding market as set out in paragraph 4.34 and has formed the view that these markets do not exhibit features of an "ideal" bidding market. The Commission notes the following:
  - a) winner takes all: in the healthcare market, this describes the nature of bidding, as typically, one bidder wins the contract. However, in the hospitality market, there is

<sup>&</sup>lt;sup>230</sup> Frontier Economics Report, paragraph 4.4.

<sup>&</sup>lt;sup>231</sup> Frontier Economics Report, paragraph 4.6.a.

<sup>&</sup>lt;sup>232</sup> Frontier Economics Report, paragraph 4.6 a.

<sup>&</sup>lt;sup>233</sup> Frontier Economics Report, paragraph 4.2 a.

<sup>&</sup>lt;sup>234</sup> Frontier Economics Report, paragraph 4.6 a and b.



an increasing distribution of contracts amongst multiple suppliers, and it is not necessarily the case that each bidder would win all or none of the contract;

- b) contracts are "lumpy": it is not the case that each contract is large and infrequent. No single contract, in itself, is large enough to account for a significant part of the healthcare or hospitality market or proportion of the total sales of a supplier. Although there is evidence in both the healthcare and hospitality markets of contracts being rolled over, the possibility of rollover is not known at the time of contract award, and the infrequency of bidding is not deliberate ;
- c) competition begins afresh for each contract: in the healthcare market, experience of previous relevant contracts is an explicit requirement for many customers (particularly public and private hospitals). The outcome of previous tenders therefore can impact on the outcome of other tenders;
- d) market entry is easy: there are significant barriers to new market entry, although switching barriers are low between existing suppliers to healthcare and hospitality customers. Barriers to new market entry to the healthcare market include financial and non-financial barriers, and notably the need for evidence of relevant previous experience.
- 4.41 The Commission's view is that, even at this high level, the healthcare and hospitality markets do not conform to the "ideal" bidding market pattern. However, in order to fully consider the operation of competition, the Commission has further examined features associated with bidding markets, in terms of:
  - Product homogeneity/differentiation;
  - Pricing incentives; and
  - Pricing transparency.

## Product homogeneity/differentiation

4.42 For the notion of an "ideal" bidding market to hold, the product market should be homogenous, with little or no differentiation between products. In the healthcare and hospitality markets, the requirements of purchasers are complex, involving not only price but a range of contract terms reflecting service quality, service levels etc. While all linen laundries have to meet minimum criteria set out in healthcare customers' ITTs, bidders compete with each other on numerous quality of service parameters (e.g., service delivery reliability, ability to provide dedicated linen stock, the level of rejected linen items, cleanliness of the facility from a bio contamination control perspective, contingency plans etc.). This is also reflected in ITTs issued by healthcare customers where various criteria are set out, sometimes with costs accounting for 50% or less of the overall marks.<sup>235</sup>

<sup>&</sup>lt;sup>235</sup> See section 12.4 "Award criteria" of the St. James Hospital ITT document. It should also be noted that [...] indicated that price accounted for only 20% of total marks in its 2018 ITT



4.43 The fact that some of the healthcare customers interviewed by the Commission<sup>236</sup> awarded contracts to bidders that did not submit the lowest price further indicates that linen laundries do not compete solely on pricing. In the Commission's view, evidence from the markets indicates that firms offer differentiated products, and this does not fit the "ideal' bidding market scenario.

## Pricing incentives

- 4.44 In the Commission's view, it is important to consider the strategic interaction between bidders engaged in bidding processes, and particularly their incentives to adopt certain pricing strategies. The notion of first and second ranked bidder is important, but cannot be the only factor considered when looking at competition for a contract, as the ranking will be the result of the terms proposed by the bidders in response to the tenderer's requirement. The bidder will take numerous factors into account, and variation in costs (for example according to capacity issues, location and so on) will feed into the eventual bid price. Each bidder's objective is to win the tender whilst offering the most profitable terms, taking into account that other bidders have the same objective. Thus, when bidders specify the terms of their offer, including the prices they will charge, they need to consider the effect that offering better conditions has on the probability of winning the tender.
- 4.45 Bidders will not typically want to offer a price at which they just break even, but seek to maximise the expected value of the contract, where quoting higher prices makes winning more valuable, but will also imply a lower likelihood of winning. In trading off greater profits from quoting higher prices against the lower probability of winning a tender, bidders will need to form expectations about other bidders their number, their relative strengths and their likely behaviour. Who is the lowest and who is the second lowest bidder will only be known *ex post*, when the bids have been made and the tenderer has ranked them. There is therefore no meaningful way of identifying a particular firm as systematically being the lowest or the second lowest priced bidder *ex ante* that could be used in any forward-looking assessment of competition in tender processes.
- 4.46 There is no meaningful way of identifying whether a particular firm would be the lowest or the second lowest bidder at the point at which the bidders decide to take part in a tender or determine the terms of their offers. If any particular bidder knew in advance that it would end up in second place (or even further down the ranking), it would either want to improve its offer enough to win or if it could not profitably do so not take part in the tender at all.
- 4.47 Economic literature suggests that in a context where there is uncertainty over competing bids, the pricing incentives of competing firms resemble those at work in ordinary markets with differentiated products.<sup>237</sup> For example, if there is uncertainty on

<sup>236 [...]</sup> 

<sup>&</sup>lt;sup>237</sup> P. Klemperer (2005), "Bidding Markets", Competition Commission discussion paper, C. Shapiro (2010), "The 2010 Horizontal Merger Guidelines: From Hedgehog to Fox in Forty Years", Antitrust Law Journal, Vol. 77, G. Werden and L. Froeb (2008), "Unilateral Competitive Effects of Horizontal Mergers", in L. Buccirossi (ed.), Handbook of Antitrust Economics, Cambridge, MA, MIT Press, and OFT (2007), "Markets with Bidding Processes", Economics discussion paper, May, all stress the close connection between sealedbid auctions and pricing of differentiated products. For example, Klemperer states *"Note that sealed-bidding corresponds to standard Bertrand price-setting. With perfect information, the sealed-bid process corresponds to Bertrand competition in a market in which all consumers make the same choice between firms. And, as noted above, with imperfect information about rivals' costs or* 



the required price level of the winning bid, each firm faces a trade-off between increasing the probability of winning the tender by offering a lower price, and the lower margin earned in case of winning from doing so.<sup>238</sup> A higher bid would reduce the probability of winning the tender, but would also increase the margin if the bid were successful.

- 4.48 This trade-off is equivalent to the standard trade-off between quantity sold and price in an ordinary differentiated goods market; the difference being that in the case of a tender it is the probability of winning rather than actual quantities sold that represents the trade-off. Each bidder, therefore, chooses its optimal bid in order to optimise the trade-off between expected sales and price and thereby maximises its expected profits. Pricing incentives and the related incentives to exploit market power in bidding markets are therefore analogous to those at work in standard pricing of differentiated products.
- 4.49 The Commission also refers to the EC *Case M.7278 GE/Alstom<sup>239</sup>* where the EC explained that facing more than one rival bidder typically increases the *ex ante* probability that the buyer will prefer a rival offer, and therefore increases the competitive constraint on any given bidder. Thus, it is not only the runner-up whose identity is not and cannot be known *ex ante* that represents a competitive constraint on the winning bidder, but the pool of remaining bidders, so that a decrease in the number of potential bidders due to the merger may result in a reduction of the competitive constraint faced by the merged entity.<sup>240</sup>

## Pricing transparency

4.50 In most cases, bidders do not know the price that their competitors are bidding. The Commission's investigation indicated that, particularly in the healthcare market,<sup>241</sup> bidders typically submit bids without observing the bids of their competitors.<sup>242</sup> For instance, St James's Hospital's ITT noted the following in respect of previous tender evaluation process:

"The Hospital is utilising the open procedure (OJEU). This is **a one stage process** whereby;

about the bid-taker's preferences, bidders making sealed bids face a trade-off between the price and their expected sales that is similar to the price-quantity trade-off firms face in standard differentiated products Bertrand competition" (page 14). Similarly, Froeb and Werden state that "the contrast between the effects of mergers in Bertrand industries and their effects in auctions is greater the more bidders know about customer-specific competitive conditions, and that contrast may be nil in sealed-bid auctions" (page 28). See footnote 241 for definition of sealed bid auction.

<sup>&</sup>lt;sup>238</sup> DotEcon Report, slides 5 and 6, 6, using a simple stylised model of bidding to illustrate this point. For the avoidance of doubt, the Commission does not consider that this model describes the reality of how bidding in the relevant markets works, but that it highlights the fundamental considerations that affect the pricing decisions of players in such markets.

<sup>&</sup>lt;sup>239</sup> The EC's decision in <u>Case M.7278 – GE/Alstom (2015)</u>, Annex I - The Commission's Economic Analysis of Bidding Data, recitals 7 et seq.

<sup>&</sup>lt;sup>240</sup> This is the 'participation effect'. See for example DotEcon for the OFT, 'Markets with Bidding Processes', 2007.

<sup>&</sup>lt;sup>241</sup> This is not always the case in the hospitality market, where the Commission has seen instances of competitors being asked to match prices.

<sup>&</sup>lt;sup>242</sup> This type of bidding process is known in economic literature as "first price" sealed bid auction. In contrast, "second price" sealed bid auctions typically refer to a category of auctions wherein some mechanism in the auction means that the price of the winning bid is determined by the level of the second-best bid. Second-price mechanisms, which are strategically similar to open bidding processes, are easy to implement where the only dimension that matters is price, but not where the offers made by bidders have multiple dimensions.

Competition and Consumer Protection Commission

The CA [Contracting Authority] invites Tenderer(s) to outline their proposal(s) for the Provision of Laundry, Linen and Miscellaneous Services. Evaluation will be undertaken by the LLMS Steering Group nominated by the CA having regard to the final award criteria set out in this ITT.....

The CA may, at its discretion, request meetings with individual Tenderer(s) during the evaluation period for the purposes of clarifying any aspect of the Tenderer(s) proposal. If appropriate or relevant, members of the Steering Group may be present and actively participate at such meetings. Such meetings will be strictly confidential and will not result in any material change to the original tender. No discussions regarding the progress of the evaluation or the Tenderer(s) performance will be entered into. All such meetings will be held at the convenience of the CA and the CA will not be responsible for any costs incurred by the Tenderer(s) (however incurred.)<sup>243</sup> (emphasis added).

## Conclusion

- 4.51 The Commission has set out its reasoning above as to why the healthcare and hospitality markets are not amenable to an idealised bidding market analysis. Therefore, market share analysis in the healthcare and hospitality markets is relevant. This significantly contributes to an understanding of: (i) the evolution of different linen laundries' market position; (ii) the market strength of different linen laundries; and (iii) the market dynamics.
- 4.52 The Commission has established that the healthcare and hospitality markets do not exhibit the features of an "ideal" bidding market, where two rivals may be sufficient to produce a competitive outcome.<sup>244</sup> Evidence from the healthcare and hospitality markets is not consistent with the parties' view of bidding markets, where competition concerns around a 3-to-2 merger can be suspended in a belief that even two competitors will provide a good competitive outcome. The Commission's view is that there is a direct impact on the level of competitive constraints faced by linen laundries (particularly in the healthcare market) of the number of competitors, especially because that number is already very small, and Post Transaction would be reduced still further.

## **Competitive Assessment**

4.53 Having set out its counterfactual, and its overall approach to considering the nature of competition in both markets, the Commission now assesses the impact of the Proposed Transaction on each of the hospitality and healthcare markets in turn.

## The hospitality market

## The nature of pre-merger competition

4.54 In their written and oral submissions, the Parties described the hospitality market as a bidding market, exhibiting the same general characteristics as the healthcare market.<sup>245</sup>

<sup>&</sup>lt;sup>243</sup> See section 6.15 "Evaluation Process" of the St. James Hospital ITT document.

<sup>&</sup>lt;sup>244</sup> Note that closeness of competition is considered in the analysis of unilateral effects.

<sup>&</sup>lt;sup>245</sup> Frontier Economics Report, paragraph 4.4.



As context for discussion below, the Commission refers back to the discussion of the Parties' view of bidding markets and the Commission's assessment of these views set out in paragraphs 4.39 to 4.50.

- 4.55 This section sets out the Commission's analysis of the hospitality market as follows:
  - market structure;
  - market concentration; and
  - competitive effects analysis.

## **Market Structure**

## Views of the Parties

- 4.56 The Parties submitted that the Commission exaggerated the role of Berendsen in the hospitality market and misjudged the closeness of competition between the Parties. In the Parties' view, the Commission has "..not produced any credible evidence to support the view that Berendsen will continue to be a viable option for hospitality customers".<sup>246</sup>
- 4.57 The Parties argued that market shares are of limited relevance, because price negotiations are bilateral in the hospitality market.<sup>247</sup>
- 4.58 The Parties also submitted that the Commission "..disregarded the competitive constraint posed by numerous viable competitors based on their geographic coverage".<sup>248</sup>
- 4.59 Finally, the Parties argued that the Commission overstated the significance of multi-site customers and understated the extent to which competitors currently serve multi-site customers. Furthermore, the Parties disagreed with the Commission's preliminary view set out in the Assessment that only the Parties and Celtic Linen are credible suppliers of flat linen rental and maintenance services to multi-site customers seeking a single supplier.<sup>249</sup>
- 4.60 These points are addressed in paragraphs 4.61 to 4.76 below.

## Views of the Commission

- 4.61 The Commission has set out its assessment of the nature of bidding markets in paragraphs 4.39 to 4.50 above. Based on this assessment, the Commission is of the view that it is relevant to consider market shares (and the evolution of market shares) when considering the structure of the hospitality market.
- 4.62 According to the Parties' estimates, on a national level, the Parties' combined market share in terms of revenue from the provision of outsourced flat linen rental and maintenance services to hospitality customers (having regard to the market definition

<sup>&</sup>lt;sup>246</sup> Parties' Response, paragraph 4.3.5.

<sup>&</sup>lt;sup>247</sup> Parties' Response, paragraph 4.3.4.

<sup>&</sup>lt;sup>248</sup> Parties' Response, paragraphs 4.3.34 to 4.3.36.

<sup>&</sup>lt;sup>249</sup> Parties' Response, paragraphs 4.3.50 to 4.3.64.



Competition and Consumer Protection Commission

adopted by the Commission) was [40-50]% with an increment<sup>250</sup> of [30-40]% as a result of the Proposed Transaction.

4.63 Table 2 below illustrates the Parties' estimated shares in the hospitality market in the period from 1 January 2017 to 31 December 2017.

Table 2: Revenue from the provision of flat linen rental and maintenance services to hospitality customers in the State, 2017 (the parties' estimates)<sup>251</sup> [ $\gg$  REDACTED]

	2017		
	€m	%	
Berendsen	[]	[0-10]	
Kings Laundry	[]	[30-40]	
Combined	[]	[40-50]	
Increase		[30-40]	
Celtic Linen	[]	[10-20]	
Linencare	[]	[0-10]	
OCL	[]	[0-10]	
Premier Linen	[]	[0-10]	
Other <sup>252</sup>	[]	[10-20]	
Total	[]	100	

Source: Information submitted by the Parties. Percentages may not add up due to rounding.

4.64 The Commission conducted a market share reconstruction and collected the actual sales volumes of all the main competitors in the State for the period from 1 January 2014 to 30 June 2018 which is illustrated by Table 3 below. On the basis of the Parties' and their main competitors' sales volume data, in the period from 1 January 2018 to 30 June 2018, the combined market share of the parties is [50-60]%, with an increment of [40-50]%. This combined market share is higher than the Parties' estimate of a combined market share of [40-50]%<sup>253</sup>.

<sup>&</sup>lt;sup>250</sup> For the purposes of this Phase 2 Determination, increment refers to incremental market share acquired by Berendsen

<sup>&</sup>lt;sup>251</sup> The Parties included potential revenue from customers currently operating OPLs. The inclusion of the potential revenue reduces the Parties' combined market share from [40-50]% to [40-50]%. Such potential revenue is not presented in Table 2 having regard to the relevant product market definition.

<sup>&</sup>lt;sup>252</sup> This group includes [...] and a range of other providers. See page 15 of the merger notification.

<sup>&</sup>lt;sup>253</sup> For a more direct comparison, in the period from 1 January 2017 to 31 December 2017, the Parties' combined market share was [50-60]%.

Coimisiún um Iomaíocht agus

	201	4	201	5	201	6	2017	,	H1 20	)18
	€000s	%	€000s	%	€000s	%	€000s	%	€000s	%
Berendsen	[]	[10- 20]	[]	[10- 20]	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]
Kings Laundry	[]	[20- 30]	[]	[30- 40]	[]	[30- 40]	[] <sup>254</sup>	[40- 50]	[]	[40- 50]
Combined	[]	[40- 50]	[]	[40- 50]	[]	[40- 50]	[]	[50- 60]	[]	[50- 60]
Increment								[40- 50]		[40- 50]
Celtic Linen <sup>255</sup>	[]	[20- 30]	[]	[20- 30]	[]	[10- 20]	[]	[10- 20]	[]	[10- 20]
OCL	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	[]	[10- 20]
Linencare	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]
Premier Linen	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]
Other <sup>256</sup>	[]	[10- 20]	[]	[10- 20]	[]	[10- 20]	[]	[10- 20]	[]	[0- 10]
Total	[]	100	[]	100	[]	100	[]	100	[]	100

## Table 3: Revenue from the provision of outsourced flat linen rental and maintenance services to hospitality customers in the State, 2014 - 30 June 2018 [ *REDACTED*]

Source: The Commission's analysis using information provided by the Parties and the Competitors.

- 4.65 Table 3 illustrates that Kings Laundry is the clear leader in the hospitality market with a [40-50]% market share at the end of June 2018. Kings Laundry's market share has grown very rapidly since 2014 (from [20-30]% in 2014 to [40-50]% in the first half of 2018) and it was bolstered by the acquisition of CWS-boco's hospitality customers in 2017. Berendsen's market share in this sector has been gradually declining and it had a market share of [0-10]% at the end of June 2018, down from a [10-20]% market share in 2014. Notably, Kings Laundry seems to have been increasing its share at the expense of both Berendsen and Celtic Linen.
- 4.66 The Parties are facing several competitors in the hospitality market with Celtic Linen ([10-20]% market share), OCL ([10-20]% market share), Linencare ([0-10]% market share) and Premier Linen ([0-10]%) being their largest competitors (in terms of market share) at the end of June 2018. Celtic Linen's market share declined significantly since 2014, but has increased in H1 2018, [...] by the acquisition of Millbrook Linen. [...] and [...] have been steadily increasing their market shares, while smaller competitors'

<sup>&</sup>lt;sup>254</sup> Kings Laundry noted that its turnover for 2017 provided in the Merger Notification Form was an approximate figure and its turnover provided in response to Question 11 of the First RFI provides an accurate figure. Thus, Table 3 reflects the most accurate turnover figures. See page 3 of Kings Laundry's response to the Commission's queries dated 16 November 2018.

<sup>&</sup>lt;sup>255</sup> Celtic Linen's revenue in H1 2018 includes revenue from customers of Millbrook Linen Limited ("Millbrook Linen"), which Celtic Linen acquired in February 2018.

<sup>&</sup>lt;sup>256</sup> This group includes revenue of [...].



(classified as 'other') combined market share declined from [10-20]% of the hospitality market in 2014 to [0-10]% in H1 2018.

- 4.67 The Commission also notes that there has been significant market consolidation in recent years:
  - a) in February 2017, Kings Laundry acquired CWS-boco's hospitality sector customers;<sup>257</sup> and
  - b) in February 2018, Celtic Linen acquired Millbrook Linen<sup>258</sup> a linen laundry that catered for hospitality customers.
- 4.68 While the Commission examined the effects of the Proposed Transaction on the structure of the market on the basis of a state-wide geographical scope (see Chapter 3 above), it is relevant to note that there are differences in competitive conditions across regions in the State.
- 4.69 The Commission's investigation reveals that there are three linen laundries that are currently supplying flat linen rental and maintenance services to hospitality customers on a national basis (the Parties and Celtic Linen).<sup>259</sup> The Commission notes that while the majority of [...] turnover is in [...], it is expanding its activities in [...] and as such is attempting to become a nationwide service provider. Other linen laundries are currently providing outsourced flat linen rental and maintenance services on a regional basis as summarised below:
  - a) [...];
  - b) [...];
  - c) [...];
  - d) [...]; and
  - e) [...].
- 4.70 Table 4 sets out linen laundries' market shares on a regional basis. It demonstrates that all linen laundries (apart from [...]) provide outsourced flat linen rental and maintenance services in Leinster. The Parties' combined market share in Leinster (region where approximately [50-60]% of total turnover in the hospitality market is generated) is [50-60]% with an increment of [40-50]%. In comparison, there are only five linen laundries providing flat linen rental and maintenance services in Munster (region where approximately [20-30]% of total turnover in the hospitality market is generated) and the parties' combined market share in this region is [70-80]% with an increment of [50-60]%.

<sup>&</sup>lt;sup>257</sup> See CWS-boco's <u>public announcement</u>.

<sup>&</sup>lt;sup>258</sup> See Celtic Linen's public announcement.

<sup>&</sup>lt;sup>259</sup> In this regard, Table 7 below sets out each linen laundries' turnover broken down by region where this turnover was generated.

Coimisiún um Iomaíocht agus

	Leinster	Munster	Connacht	Ulster <sup>261</sup>
Linen laundry	%	%	%	%
Berendsen	[0-10]	[10-20]	[0-10]	[0-10]
Kings Laundry	[40-50]	[50-60]	[30-40]	[0-10]
Combined	[50-60]	[70-80]	[40-50]	[0-10]
Increment	[40-50]	[50-60]	[30-40]	[0-10]
Celtic Linen	[10-20]	[10-20]	[10-20]	[0-10]
OCL	[0-10]	[0-10]	[40-50]	[0-10]
Linencare <sup>262</sup>	[10-20]	[0-10]	[0-10]	[20-30]
Premier Linen	[10-20]	[0-10]	[0-10]	[0-10]
Lilliput	[0-10]	[0-10]	[0-10]	[50-60]
Carraig Linen	[0-10]	[0-10]	[0-10]	[0-10]
Ashton Linen	[0-10]	[0-10]	[0-10]	[0-10]
Millbrook Linen	[10-20]	[0-10]	[0-10]	[0-10]
Total revenue (€000's)	[]	[]	[]	[]

Table 4: Revenue from the provision of outsourced flat linen rental and maintenance services in
the hospitality market by region, 2017 <sup>260</sup> [

Source: The Commission's analysis using information provided by the Parties and the Competitors.

- 4.71 Thus, market share variations on a regional basis are relatively high. The Commission also notes that some linen laundries are likely to subcontract work won in a region where they are not currently present. For example, [...] informed the Commission that it would likely subcontract outsourced flat linen rental and maintenance services in the Munster area to one of its competitors if it won a contract with a multi-site customer with a site in Munster.<sup>263</sup>
- 4.72 The Commission's view is that the regional variation in market shares does not reflect a constraint on linen laundries, but rather reflects the fact that linen laundries tend to be stronger in areas closer to their facilities. The available evidence indicates that linen laundries with intentions to expand (e.g., [...]) can and do supply customers throughout the State from a single facility.
- 4.73 The Commission has considered the Parties' views that their supply to multi-site customers is not significant and that weight should be given to the extent to which multi-site customers choose to use multiple flat linen suppliers. The Commission understands that some multi-site hospitality customers may choose to use a single national supplier, but that several multi-site customers (e.g., Dalata Hotel group) use multiple suppliers. In addition, the Commission notes that [...] and [...] have recently

<sup>&</sup>lt;sup>260</sup> Turnover of [...] is not included in this analysis, as they are no longer competing actively in the hospitality market.

<sup>&</sup>lt;sup>261</sup> Turnover generated in Ulster includes turnover generated in Northern Ireland, as some Competitors were unable to separate turnover figures for Ulster counties in the State.

<sup>&</sup>lt;sup>262</sup>Linencare indicated the counties where it provides outsourced flat linen rental and maintenance services, but did not provide turnover breakdown by region. Thus, the Commission made assumptions on Linencare's regional breakdown of turnover based on the regional breakdown of turnover of other linen laundries operating in Leinster in the hospitality market.

<sup>&</sup>lt;sup>263</sup> Please see phone call record with [...] dated 19 February 2019.



decided to diversify their purchase of outsourced flat linen rental and maintenance services by giving business to [...].<sup>264</sup>

4.74 In summary, the Commission notes that currently Kings Laundry is the sole supplier to [...] multi-site customers. Only one other linen laundry (i.e., Lilliput) is the sole supplier to any hotel group. However, the Commission recognises that linen laundries can and do subcontract to cover areas where it may be less economic for them to supply directly and that this reduces any requirement for a national presence.

## Conclusion on Market Structure

- 4.75 The Commission is of the view that market shares are relevant for an understanding of the hospitality market structure and market dynamics. The Commission's investigation revealed that following the implementation of the Proposed Transaction, the merged entity will enjoy a very strong position in the hospitality market with a market share of over [50-60]%. The Commission has taken into account the EC's Horizontal Merger Guidelines regarding the relationship between high market shares and market power.<sup>265</sup>
- 4.76 The Commission notes Berendsen's declining market share in the hospitality market and notes the Parties' view that "[...]".<sup>266</sup> The Commission recognises the fact that the merged entity's high market share is largely accounted for by Kings Laundry's growth. The Commission notes also that some linen laundries have increased their market share over the last 5 years.

## **Market Concentration**

4.77 Based on the market share estimates set out in Table 3 above, the Commission is of the view that the hospitality market is highly concentrated.<sup>267</sup> Table 5 illustrates that the HHI (see paragraph 4.223 for the description of HHI) following implementation of the Proposed Transaction would be over 3,000.<sup>268</sup> Furthermore, the change in the HHI would be approximately 609. As set out in paragraph 3.10 of the Commission's Merger Guidelines, means that the Commission could not conclude at the end of the Phase 1 Determination that the Proposed Transaction was unlikely to raise competition concerns in the hospitality market and thus, had to intensify its analysis of the likely competitive impact of the Proposed Transaction in this market.

	H1 2018
HHI pre-Transaction	2,674
HHI post-Transaction	3,283

<sup>&</sup>lt;sup>264</sup> See the Commission's meeting minutes with [...].

<sup>&</sup>lt;sup>265</sup> See paragraph 4.215 below.

<sup>&</sup>lt;sup>266</sup> Parties' Response, paragraph 4.3.7.

<sup>&</sup>lt;sup>267</sup> Market shares of individual competitors combined under *Other* category in Table 12 above were taken into account individually in order to calculate a more accurate estimate of the HHI.

<sup>&</sup>lt;sup>268</sup> The Commission notes that the Proposed Transaction leads to an even greater concentration of an already highly concentrated market.



Competition and
<b>Consumer Protection</b>
Commission

	H1 2018
HHI delta	609

Source: The Commission's analysis using information provided by the Parties and the Competitors.

## **Competitive Effects Analysis**

4.78 In paragraphs 4.79 to 4.208 below, the Commission sets out its assessment of other factors, including closeness of competition between the merging parties, before reaching a conclusion as to whether the Proposed Transaction would be likely to lead to an SLC in the hospitality market.

## Views of the Parties

## Pre-merger competitive pressure

- 4.79 The Parties submitted that they faced competition from a large number of competitors across a broad range of product/service offerings listed in paragraph 1.4 above. In this regard, the Parties state that their activities overlap only with respect to flat linen rental and maintenance services.<sup>269</sup>
- 4.80 The Parties argued that their activities are largely complementary. The Parties stated that Berendsen focuses on the supply of workwear rental and maintenance services and, to a lesser extent, the supply of flat linen rental and maintenance services to healthcare customers while Kings Laundry focuses on the supply of flat linen rental and maintenance services to hospitality customers. In support of this argument, the Parties referred to their revenue in each customer sector. The Parties claimed that the complementarity between the Parties' activities means that, following the implementation of the Proposed Transaction, the merged entity "...will be able to serve a broader range of customers and to a higher standard across the board."<sup>270</sup>

## Closeness of competition

The Parties' pre-merger documents

4.81 Berendsen's internal documents show that [...] in the hospitality market. In particular the Commission notes the following references:

"[...]"<sup>271</sup>

"[...]"<sup>272</sup>

<sup>&</sup>lt;sup>269</sup> See pages 23 to 24 of the Merger Notification Form.

<sup>&</sup>lt;sup>270</sup> Page 24 of the Merger Notification Form.

<sup>&</sup>lt;sup>271</sup> Page 1 of the of the Document titled [...] dated 11 October 2016 provided by Berendsen in response to question 9 of the First RFI. During the Commission's meeting with Berendsen on 30 January 2019, Berendsen confirmed that [...].

<sup>&</sup>lt;sup>272</sup> Page 4 of the of the Document titled [...] dated 11 October 2016 provided by Berendsen in response to question 9 of the First RFI. During the Commission's meeting with Berendsen on 30 January 2019, Berendsen confirmed that [...].



## "[...]"<sup>273</sup>

## The Parties' written and oral submissions

- 4.82 The Parties expressed a view that Berendsen is [...] in the hospitality market<sup>274</sup> and that the Parties are not close competitors as evidenced from tender participation analysis.<sup>275</sup> In this regard, the Parties argued that the relevant measure of the closeness of competition between the Parties is the probability that, for the contracts one of the merging parties' bids for and wins, the other merging party is the customer's second choice bidder.<sup>276</sup>
- 4.83 The Parties claimed that customer loss data demonstrates that Berendsen does not exercise any competitive constraint on Kings Laundry. During the relevant time period [...] of the customers lost by Kings Laundry was lost to Berendsen. In the Parties' view, while Berendsen lost [...]% of its hospitality customers to Kings Laundry between January 2014 and June 2018, this does not prove a direct measure of diversion. According to the Parties, for there to be any material loss of competition, whenever Berendsen lost to Kings Laundry, it would need to be established that Berendsen was the second choice bidder for that customer.<sup>277</sup>

## Constraints from competitors

- 4.84 The Parties submitted that the merged entity will face significant competition following the implementation of the Proposed Transaction. In this regard, the Parties referred to existing competitors and in particular, Celtic Linen, Linencare, OCL and Premier Linen that are larger than or the same size as Berendsen.<sup>278</sup> The Parties also considered wash-only laundrettes to pose a significant competitive constraint on them.<sup>279</sup> The Parties were of the view that, in its Assessment, the Commission downplayed the competitive constraint that Celtic Linen will exercise in the hospitality market post-merger.<sup>280</sup>
- 4.85 For an overview of the Parties' Report on Celtic Linen, see paragraph 4.239 below. In addition to the information outlined in paragraph 4.239, the Parties noted that Kings Laundry bid for and lost [...] hospitality contracts<sup>281</sup> to Celtic Linen since January 2017.<sup>282</sup> Separately, the Parties referred to Celtic Linen's recent win of the Center Parcs contract and considered that this increases Celtic Linen's competitive strength. The Parties noted that the contract for Center Parcs in Longford Forest, which accommodates up to 2,500 guests, is one of the largest hospitality contracts in the State.<sup>283</sup>
- 4.86 The Parties also argued that, in its Assessment, the Commission erred in its evaluation of the merged entity's incentive to increase price in the hospitality market by

<sup>&</sup>lt;sup>273</sup> Page 1 of the of the Document titled [...] dated 11 October 2016 provided by Berendsen in response to question 9 of the First RFI.

<sup>&</sup>lt;sup>274</sup> Parties' Response, paragraphs 4.3.7 to 4.3.8.

<sup>&</sup>lt;sup>275</sup> Parties' Response, paragraphs 4.8 to 4.24 and Frontier Economics Report, paragraph 4.2.c.

<sup>&</sup>lt;sup>276</sup> Frontier Economics Report, paragraph 4.6.b.

 $<sup>^{\</sup>rm 277}$  Frontier Economics Report, paragraphs 4.19 c to 4.22.

<sup>&</sup>lt;sup>278</sup> Page 22 of the Merger Notification Form.

<sup>&</sup>lt;sup>279</sup> Page 19 of the Merger Notification Form.

<sup>&</sup>lt;sup>280</sup> Parties' Response, paragraphs 4.3.24 to 4.3.33.

<sup>&</sup>lt;sup>281</sup> These [...] hospitality contracts are as follows: [...].

<sup>&</sup>lt;sup>282</sup> Paragraph 33 in the Report on Celtic Linen.

<sup>&</sup>lt;sup>283</sup> Paragraph 32 in the Report on Celtic Linen.



Competition and Consumer Protection Commission

*"incorrectly estimating the price differential and mistakenly assuming the mere existence of a price differential is proof of incentive to increase price"*<sup>284</sup> and that it failed to demonstrate that Celtic Linen would have both the incentive and the ability to follow an increase in price by the merged entity.

#### Likelihood of new entry

- 4.87 The Parties submitted in the Merger Notification Form that they face a credible threat of entry in the hospitality market.<sup>285</sup> The Parties referred specifically to wash-only laundrettes and noted that these laundrettes pose a credible threat of entry into the hospitality market by expanding their current operations. The Parties referred to Kings Laundry, Linencare and Carraig Linen and noted that these operators originally operated as wash-only laundrettes and have subsequently expanded into the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State.
- 4.88 During a meeting with the Commission, Berendsen provided its views on the importance of experience and reputation in the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State. Berendsen considered experience and brand reputation to be an important factor in this sector, yet noted that its importance can be superseded with a better price offer. In addition, Berendsen expressed its view that it is relatively straightforward to enter and grow in the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State. In this regard, Berendsen referenced Carraig Linen and noted that it was set up by a former employee of Berendsen. Furthermore, Berendsen was of the opinion that while reputation does matter, reputation can be achieved relatively quickly.
- 4.89 In addition, Berendsen provided the Commission with a list of potential competitors that Berendsen considered may enter the hospitality market in the next 2 to 3 years, including Johnsons Services Group; K-Bro Linen Inc.; CLEAN Limited; CWS-boco; Lindstrom; Shortridge Limited; The Brilliant Laundry Group Limited; Bardusch; Alsco Inc.; Salesianer; and Cintas Corporation.<sup>286</sup> Berendsen noted that such entry could occur through either the acquisition of an existing business or greenfield entry as, in Berendsen's view, both options represent viable opportunities for entry.<sup>287</sup> However, Berendsen did not provide any supporting documents or evidence in relation to the timeliness, likelihood or sufficiency of entry by any of those listed potential competitors.
- 4.90 Kings Laundry informed the Commission that it is not currently aware of any potential new entrant in the supply of flat linen rental and maintenance services to hospitality customers in the State.<sup>288</sup> However, Kings Laundry opined that entry is more likely in the supply of outsourced flat linen rental and maintenance services to hospitality customers than to healthcare customers in the State.<sup>289</sup>
- 4.91 In the Parties' view, there are numerous viable competitors in the hospitality market and several competitors pose more significant competitive constraints on Kings Laundry

<sup>&</sup>lt;sup>284</sup> Paragraphs 4.3.14 to 4.3.15 of the Parties' Response and paragraphs 4.2.d, and 4.25 to 4.42 of the Frontier Economics Report.

<sup>&</sup>lt;sup>285</sup> See section 5.3 of the Merger Notification Form.

<sup>&</sup>lt;sup>286</sup> See Berendsen's response to question 22 of the Second RFI.

<sup>&</sup>lt;sup>287</sup> See Berendsen's response to question 22 of the Second RFI.

<sup>&</sup>lt;sup>288</sup> See Kings Laundry's response to question 19 of the Second RFI.

<sup>&</sup>lt;sup>289</sup> See Kings Laundry's response to question 19 of the Second RFI.

than Berendsen does. The Parties argued that these competitors have varying degrees of spare capacity, and should be considered as a competitive constraint.<sup>290</sup>

- 4.92 The Parties submitted that competing linen laundries (specifically OCL) are not geographically limited. Furthermore, the Parties noted that facility location was not perceived to be an important factor by Amárach Survey respondents, Customers and Competitors.<sup>291</sup> The Parties argued that the available evidence indicates that one facility is sufficient for a linen laundry to provide outsourced flat linen rental and maintenance services on a national basis.<sup>292</sup>
- 4.93 The Parties argued that the Commission was wrong to assume that Kings Laundry's competitive strategy will change post-merger. According to the Parties, there were particular reasons why Elis increased prices in the UK and in Brazil following the acquisition of organisations providing outsourced flat linen rental and maintenance services. The Parties noted that Kings Laundry's cost advantage and spare capacity will remain post-merger.<sup>293</sup>
- 4.94 The Parties argued that the Commission ignored and/or failed to give sufficient weight to views expressed by hospitality customers and competitors. The Parties argued that third parties do not have concerns regarding the Proposed Transaction.<sup>294</sup>

#### **Countervailing Buyer Power**

Coimisiún um Iomaíocht agus

Cosaint Tomhaltóirí

- 4.95 The Parties considered that customers in the hospitality sector are sophisticated, price sensitive and service level conscious.<sup>295</sup> Thus, in the Parties' view, these customers can credibly avail of the following options in response to any attempt by the merged entity to increase prices:
  - a) switch to alternative suppliers of flat linen rental and maintenance services;
  - b) switch to self-supply by setting up OPLs; or
  - c) purchase linen items, and outsource maintenance to wash-only launderettes.
- 4.96 The Parties considered that exercising switching options is relatively easy, as, in their view, barriers to switching are relatively low. The Parties referred to the relatively [...] of their contracts with customers<sup>296</sup> and argued that the costs of switching suppliers at the end of the contract period are low.
- 4.97 In the Report on OPLs, the Parties noted the precedent of NCAs (including the Commission) considering the option of self-supply during the assessment of competition.<sup>297</sup> The Parties noted that a number of customers in the hospitality sector

<sup>&</sup>lt;sup>290</sup> Parties' Response, paragraphs 4.3.34 to 4.3.36.

<sup>&</sup>lt;sup>291</sup> Parties' Response, paragraphs 4.3.41 to 4.3.43.

<sup>&</sup>lt;sup>292</sup> Parties' Response, paragraphs 4.3.44 to 4.3.49.

<sup>&</sup>lt;sup>293</sup> Parties' Response, paragraphs 4.3.19 to 4.3.23.

<sup>&</sup>lt;sup>294</sup> Paragraphs 4.3.83 to 4.3.88 of the Parties' Response and paragraphs 1.3.e, 4.3, 7.4 and 7.5 of the Frontier Economics Report.

<sup>&</sup>lt;sup>295</sup> Page 25 of the Merger Notification Form.

<sup>&</sup>lt;sup>296</sup> The Parties also noted that [...] of their hospitality customers do not have contracts with them. In instances where contracts exist,

the Parties noted that the duration of these contracts is typically between [...]. See page 12 of the Merger Notification Form.

<sup>&</sup>lt;sup>297</sup> Paragraphs 4 to 9 of the Report on OPLs.



(e.g., B&Bs) currently operate OPLs.<sup>298</sup> Similarly, the Parties reiterated that several hotels opted to switch to self-supply in the past.<sup>299</sup> In relation to customers that do not have OPLs, the Parties reiterated that the costs associated with setting up an OPL are relatively small.<sup>300</sup> The Parties noted that customers can avail of second hand machinery, citing providers and purchasers of such machinery in the State.<sup>301</sup> The possibility to source equipment from suppliers located outside the State was also highlighted.<sup>302</sup> Thus, the Parties were of the view that customers in the hospitality sector that are outsourcing flat linen rental and maintenance services can exercise significant CBP by threatening to switch to self-supply.

4.98 Finally, according to the Parties, hospitality customers' CBP is driven by consolidation in the hotel industry; the strategy of maintaining a diverse flat linen supplier base; and the potential to switch to OPL. The Parties noted the difficulties faced by linen laundries in attempting to impose a price increase and argued that the existence of rebates demonstrates the existence of CBP.<sup>303</sup>

## Views of third parties

## Hospitality Competitors

- 4.99 The Commission notes that when responding to the Commission's query regarding the competitive effects of the Proposed Transaction, Competitors did not segment their response in respect of the healthcare market and the hospitality market. As such, the views of those which currently supply flat linen rental and maintenance services to healthcare customers in the State (i.e., [...]) as well as to hospitality customers summarised in paragraph 4.255 above are also summarised below.
- 4.100 Seven out of the nine Competitors<sup>304</sup> contacted by the Commission which currently provide flat linen rental and maintenance services to hospitality customers in the State responded to this question in the First Competitor Questionnaire. <sup>305</sup>
- 4.101 Two<sup>306</sup> out of seven Competitors that provided views on this query indicated that the Proposed Transaction would result in a price increase, yet claimed that this price increase is *"much needed"*<sup>307</sup> as *"prices are currently slightly lower than what they should be anyway"*<sup>308</sup>. One<sup>309</sup> Competitor noted that the Proposed Transaction would

<sup>&</sup>lt;sup>298</sup> Paragraph 15 of the Report on OPLs.

<sup>&</sup>lt;sup>299</sup> Paragraph 31 of the Report on OPLs.

<sup>&</sup>lt;sup>300</sup> Paragraph 16 of the Report on OPLs.

 <sup>&</sup>lt;sup>301</sup> Paragraphs 33 to 34 of the Report on OPLs.
<sup>302</sup> Paragraph 35 of the Report on OPLs.

<sup>&</sup>lt;sup>303</sup> Parties' Response, paragraphs 4.3.65 to 4.3.75.

<sup>&</sup>lt;sup>304</sup> These 9 Competitors are as follows: [...]. The 7 competitors which responded to this query are as following: [...]. For completeness, the Commission notes that it also contacted [...] however, as previously mentioned, these parties are not currently active in the supply of Flat Linen Rental and Maintenance Services to hospitality customers in the State. While [...] did not provide views on the competitive effects of the Proposed Transaction, [...] did. For more information, please see [...] response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>305</sup> The Commission notes that such competitors provided a response in respect of the competitive effects of the Proposed Transaction in relation to the supply of outsourced flat linen rental and maintenance services in the State and did not segment their answer in respect of healthcare and hospitality. Please see paragraph 4.255 below for the [...] response to this query.

<sup>&</sup>lt;sup>306</sup> Please see [...] response to the First Competitor Questionnaire and [...] response to question 29 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>307</sup> See [...] response to question 29 of the First Competitor Questionnaire.

 $<sup>^{\</sup>rm 308}$  See [...] response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>309</sup> See [...] response to question 29 of the First Competitor Questionnaire.



have no impact on either their own business or customers of Berendsen and Kings Laundry. Another Competitor noted that they "don't believe the market will be affected" in respect of the Proposed Transaction citing the fact that Kings Laundry mainly serves the hospitality sector and that Berendsen mainly serves the healthcare sector and workwear.<sup>310</sup> One Competitor noted that the impact of the Proposed Transaction on customers would be positive and noted that the "resultant synergies should benefit the customer".<sup>311</sup> One Competitor noted that the Proposed Transaction would reduce competition and choice for customers.<sup>312</sup> Finally, the remaining Competitor which provided views on this issue noted that the impact of the Proposed Transaction was unforeseeable.<sup>313</sup>

## Customers and Self-suppliers in the hospitality sector

- 4.102 A number of Customers from the hospitality sector that responded to the Questionnaire to Customers provided views on the parties' closest competitors and the impact of the Proposed Transaction in the hospitality market.<sup>314</sup>
- 4.103 In relation to closeness of competition, Celtic Linen was consistently identified as the closest competitor to the Parties with all five Customers who responded in relation to this question identifying Celtic Linen as the Parties' main competitor. OCL, and to a lesser extent CWS-boco and Sovereign Linen, were also mentioned.
- 4.104 In relation to the impact of the Proposed Transaction, on their ability to purchase their linen services, Customers from the hospitality sector expressed less of a concern than those in healthcare, with some saying it would have no impact and others saying it could be a negative or positive impact.<sup>315</sup>
- 4.105 In relation to the question of the impact of the Proposed Transaction on the price of outsourced flat linen rental and maintenance services, of the three hospitality Customers who responded, two said that they would expect prices to increase and one said that they would expect no change.<sup>316</sup>
- 4.106 Three Self-suppliers from the hospitality sector that expressed views on the impact of the Proposed Transaction were of the view that the Transaction will have no impact in the hospitality market.<sup>317</sup> One Self-supplier from the hospitality sector did not have strong views, but noted that the reduced number of competitors could lead to an increase in prices in the hospitality market.<sup>318</sup>
- 4.107 The Amárach Survey indicated that nearly two thirds (64%) of respondents from the hospitality sector indicated they thought that the Proposed Transaction would have a

<sup>&</sup>lt;sup>310</sup> See [...] response to question 29 of the First Competitor Questionnaire.

 $<sup>^{\</sup>rm 311}$  See [...] response to question 29 and 30 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>312</sup> See [...] response to question 29 and 30 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>313</sup> See [...] response to question 30 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>314</sup> See paragraphs 10.29 to 10.30 below.

 $<sup>^{\</sup>rm 315}$  See paragraphs 10.31 to 10.33 below.

 $<sup>^{\</sup>scriptscriptstyle 316}$  See paragraphs 10.34 to 10.35 below.

<sup>&</sup>lt;sup>317</sup> See paragraphs 10.43 to 10.44 below.

<sup>&</sup>lt;sup>318</sup> See the Commission's phone call record with [...] dated 24 January 2019.

neutral impact, with 16% saying a positive impact and 9% saying it would have a negative impact.  $^{\rm 319}$ 

## Views of the Commission

Coimisiún um

Iomaíocht agus

Cosaint Tomhaltóirí

4.108 The Commission focused on two likely theories of harm as part of its assessment of the likely competitive impact of the Proposed Transaction in the hospitality market: (i) unilateral effects; and (ii) coordinated effects. The impact of coordinated effects is discussed in 4.379 to 4.386 below.

## Unilateral Effects

- 4.109 In considering the likelihood of unilateral effects occurring following implementation of the Proposed Transaction, the Commission has considered the factors identified in the Commission's Merger Guidelines as outlined in paragraph 4.27 and assesses a number of factors:
  - a) competitive pressure exerted by Kings Laundry on Berendsen (and other competitors) before the Proposed Transaction (discussed in paragraphs 4.110 to 4.119 below);
  - b) closeness of competition between the Parties (discussed in paragraphs 4.120 to 4.148 below)
  - c) constraints from remaining competitors of the Parties (discussed in paragraphs 4.149 to 4.186 below);
  - d) likelihood of new entry (discussed in paragraphs 4.187 to 4.195 below); and
  - e) countervailing buyer power (discussed in paragraphs 4.196 to 4.208 below).

## Pre-merger competitive pressure exerted by Kings Laundry on Berendsen

- 4.110 Since Kings Laundry commenced operating as a provider of outsourced flat linen rental and maintenance services, it has been competing vigorously in order to grow its customer base in the hospitality sector. As noted in paragraph 4.65 above, Kings Laundry's market share has grown very rapidly since 2014 and it currently provides services to major hotel groups in the State including [...]<sup>320</sup>. Furthermore, a significant portion of Kings Laundry's largest hospitality customers (in terms of turnover) was gained at the expense of both Berendsen and [...]<sup>321</sup>.
- 4.111 Kings Laundry's competitive strategy in the hospitality sector appears to be focussed on providing high quality service at relatively low prices. In paragraph 4.266 below, the Commission refers to Kings Laundry's dedicated linen system which Kings Laundry uses

<sup>&</sup>lt;sup>319</sup> See slide 59 of the Amárach Survey.

<sup>&</sup>lt;sup>320</sup> Also see Table 8 below.

<sup>&</sup>lt;sup>321</sup> See paragraph 4.65 above.

to differentiate itself from its competitors. Kings Laundry notes that it is  $[...]^{322}$  [...]. Nevertheless, Kings Laundry considers that  $[...]^{323}$ 

4.112 Kings Laundry publicly promotes itself as an important competitor and an innovative player in the hospitality market. For example, in an article published in "Hotel & Catering Review" magazine issued in February 2016<sup>324</sup> Kings Laundry stated the following:

"We reduced our costs during the recession";

"We took a very old system, modernised it, and put it into our plant in Dublin. We set it up and dedicated it to specific hotels. We introduced our dedicated linen rental service in Tallaght in 2008 and growth in demand for our services has really taken off since then – and it has reduced the average price of laundry by 20% over the last seven years. With the dedicated system, the customer is kept fully stocked"; and

"So our competitors are suffering badly because our product selection, distribution systems and customer focus provides a level of service that other commercial laundries can't match".

- 4.113 Kings Laundry confirmed that it was expanding during the recession at the time when other linen laundries were losing revenue, but noted that the aforementioned article should be read in a context where Kings Laundry was [...].<sup>325</sup> The Commission considers that Kings Laundry's ability to expand during a time when the overall demand for flat linen rental and maintenance services in the hospitality sector was in decline is illustrative of Kings Laundry's competitive acumen.
- 4.114 In Table 6, the Commission sets out the average price per linen item<sup>326</sup> charged by competitors in the hospitality market in the State. The Commission notes that pricing of outsourced flat linen rental and maintenance services will depend on many factors including customer size, customer location, frequency of deliveries etc. It is for this reason that the Commission has not attempted to estimate the average price of each individual linen laundry. Instead the Commission asked the Parties and Competitors to provide their own estimate of their average prices charged to: (i) healthcare; and (ii) hospitality customers.
- 4.115 In this regard, the Commission notes that in Berendsen's internal documents Berendsen itself [...] (see Figure 6 below). Thus, the Commission found it appropriate to compare the average prices cited by all linen laundries.

Figure 6: Extract from Berendsen's Internal Document [*X* REDACTED]

<sup>&</sup>lt;sup>322</sup> Stated by Kings Laundry during the Commission's meeting with Kings Laundry on 1 February 2019.

<sup>&</sup>lt;sup>323</sup> Ibid.

<sup>&</sup>lt;sup>324</sup> See pages 22 and 23. Accessible at <u>https://issuu.com/ashvillemedia/docs/h c feb 2016 online</u>.

<sup>&</sup>lt;sup>325</sup> Stated by Kings Laundry during the Commission's meeting with Kings Laundry on 1 February 2019.

<sup>&</sup>lt;sup>326</sup> The Commission understands that this metric is derived by dividing total turnover in a given period by the number of linen items processed by linen laundry in the same period.



4.116 Table 6 indicates that Kings Laundry's average price in 2018 was [...] ([...] price was [...]). In comparison, Berendsen's and Celtic Linen's prices were [10-20]% to [10-20]% [...] than Kings Laundry's price respectively.

Linen Laundry	Average price per linen item in the hospitality market
Premier Linen	€[]
Kings Laundry	€[]
Linencare	€[]
OCL	€[]
Carraig Linen	€[]
Lilliput	€[]
Berendsen	€[]
Celtic Linen	€[]
Ashton Linen	€[]

Table 6: Average price of flat linen items in the hospitality market in 2018 [*REDACTED*]

Source: Information provided by the Parties and the Competitors.

4.117 Kings Laundry also demonstrated willingness to invest in modern equipment which allowed it to achieve greater efficiencies.<sup>327</sup> In this regard, the Commission notes that in 2016 Kings Laundry opened its new facility in Cork having invested €[...] in total.<sup>328</sup> Berendsen, in its internal documents that describe [...] notes the following in relation to [...]:

"[...]"<sup>329</sup>

- 4.118 The importance of Kings Laundry as a strong competitive force in the hospitality market is further supported by evidence from Berendsen's internal documents as set out below:
  - "[...]"<sup>330</sup>
  - "[...]"<sup>331</sup>

<sup>&</sup>lt;sup>327</sup> During the Commission's meeting with Kings Laundry on 1 February 2019, Kings Laundry noted that its Cork facility is [...].

<sup>&</sup>lt;sup>328</sup> In response to question 13 of the First RFI, Kings Laundry noted that [...].

<sup>&</sup>lt;sup>329</sup> Internal email between Berendsen and Elis titled [...], dated 2 February 2018 provided by Berendsen in response to question 1 of the first RFI.

<sup>&</sup>lt;sup>330</sup> Internal email between Berendsen and Elis titled [...], dated 2 February 2018 provided by Berendsen in response to question 1 of the first RFI.

<sup>&</sup>lt;sup>331</sup> Internal email between Berendsen and Elis titled [...], dated 19 March 2018 provided by Berendsen in response to question 1 of the first RFI.



- "[...]"<sup>332</sup>
- "[...]"<sup>333</sup>
- "[...]
- [...]"<sup>334</sup>
- 4.119 In essence, the Proposed Transaction would see Kings Laundry, with a high market share of the hospitality market, being incorporated into Berendsen, which has a relatively limited and declining market share in the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State.<sup>335</sup>

## **Closeness of competition**

- 4.120 Paragraph 4.19 of the Commission's Merger Guidelines states that "All things being equal, a merger between close competitors (i.e., competitors engaged in intense competition)<sup>336</sup> will remove a strong competitive constraint and hence be more likely to raise competition concerns than a merger between distant competitors."
- 4.121 In assessing the closeness of competition between Berendsen and Kings Laundry the Commission has examined the following evidence:
  - a) the views of the Parties (summarised in paragraphs 4.81 to 4.83 above);
  - b) the Parties' pre-merger documents (discussed in paragraph 4.81 above);
  - c) the views of third parties (summarised in paragraphs 4.99 to 4.107 above);
  - d) the analysis of geographic areas and customer segments that linen laundries are catering for (discussed in paragraphs 4.122 to 4.129 below);
  - e) analysis of tendering data (discussed in paragraphs 4.130 to 4.134 below);
  - f) analysis of the Parties' customer loss data (discussed in paragraphs 4.135 to 4.141 below); and
  - g) the Amárach Survey (discussed in paragraphs 4.142 to 4.145 below).

The analysis of geographic areas and customer segments that linen laundries are catering for

4.122 The Commission examined whether the Parties target the same geographic areas and/or customer segments. In this regard, the Commission notes that currently the Parties and Celtic Linen provide nationwide services, with [...] also attempting to become a national supplier. Other linen laundries tend to provide services regionally

<sup>&</sup>lt;sup>332</sup> Page 1 of the of the Document titled [...] provided by Berendsen in response to question 9 of the First RFI.

<sup>&</sup>lt;sup>333</sup> Page 3 of the of the Document titled [...] provided by Berendsen in response to question 9 of the First RFI.

<sup>&</sup>lt;sup>334</sup> Page 4 of the of the Document titled [...] dated 11 October 2016 provided by Berendsen in response to question 9 of the First RFI. During the Commission's meeting with Berendsen on 30 January 2019, Berendsen confirmed that [...].

<sup>&</sup>lt;sup>335</sup> According to Berendsen's internal document [...] provided by Berendsen in response to Question 9 of the first RFI.

<sup>&</sup>lt;sup>336</sup> Footnote 9 of the Commission's Merger Guidelines states that *"In a differentiated product market the most intense competition will occur between the most substitutable products or services.".*


Competition and Consumer Protection Commission

(i.e., to customers that are relatively close to their facilities). This is illustrated by examining turnover by region, and is shown in Table 7 below.



		Bere	endsen	Kings La	aundry	Celtic	Linen	0	CL	Line	ncare		mier Ien	Lilli	iput		rraig nen
	Region	€	%	€	%	€	%	€	%	€	%	€	%	€	%	€	%
In	Leinster	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
000s	Munster	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Connacht	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Ulster	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Total	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%

# Table 7: Linen laundries' turnover in the hospitality market by region, 1 January 2017 - 31 December 2017 [ 🖉 REDACTED]

Source: The Commission's analysis of information provided by linen laundries



Competition and Consumer Protection Commission

- 4.123 In addition, the Commission asked linen laundries to indicate whether they provide flat linen rental and maintenance services to customers that have sites located in 3 or more regions of the State. According to the Parties, Berendsen supplies outsourced flat linen rental and maintenance services to [...] customer with sites in two regions of the State, but does not supply any hospitality customers with sites in more than two regions.<sup>337</sup> Kings Laundry serves purchasing groups, rather than individual customers with multiple sites and in 2018 it supplied [...] purchasing groups with sites in 3 or more regions.<sup>338</sup> [...] reported revenue from customers with sites in 3 or more regions, but no other linen laundry indicated any revenue from customers with multi-regional sites.
- 4.124 In order to further examine the significance of multi-site customers, the Commission assessed current and previous supply to hotel groups or hospitality management/purchasing groups.<sup>339</sup> The Commission's analysis is set out in Table 8.

<sup>&</sup>lt;sup>337</sup> Frontier Economics Report, paragraph 2.14 a.

<sup>&</sup>lt;sup>338</sup> Frontier Economics Report, paragraph 2.14 a.

<sup>&</sup>lt;sup>339</sup> Hospitality management/purchasing groups such as [...] act as outsourced procurement partners on behalf of independent hotels. The list of hotel and hospitality management/purchasing groups was provided by Kings Laundry in response to Question 20 of the First RFI. It should not be interpreted as exhaustive list of all multi-site hospitality customers in the State, but it provides indication on which linen laundries are catering for these type of customers.



Table 8: Multi-site customer analysis [ *SREDACTED*]

			[]	[.	]	[.	]	[,	]	[]	340	[	.]	[]	]	[	.]
iD	Hotel group	Supply now (SN)	Supplied previously (SP)	SN	SP	SN	SP	SN	SP	SN	SP	SN	SP	SN	SP	SN	SP
1	[]	YES	-	YES	-	YES	-	YES	-	NO	NO	YES	-	YES	-	NO	NO
2	[]	NO	NO	YES	-	NO	YES	YES	NO	NO	NO	NO	NO	NO	NO	NO	NO
3	[]	NO	NO	YES	-	YES	-	YES	-	YES	-	NO	NO	NO	NO	NO	NO
4	[]	NO	NO	YES	-	NO	NO	NO	NO	YES	-	NO	NO	NO	YES	NO	NO
5	[]	NO	NO	YES	-	NO	NO	YES	-	NO	NO	NO	NO	NO	NO	NO	NO
6	[]	NO	YES	YES	-	NO	NO	NO	YES	NO	NO	NO	NO	NO	NO	NO	NO
7	[]	NO	NO	YES	-	NO	NO	NO	NO								
8	[]	NO	NO	YES	-	NO	YES	NO	NO	NO	NO						
9	[]	NO	YES	YES	-	NO	YES	NO	YES	NO	NO						
10	[]	NO	NO	YES	-	NO	YES	NO	YES	NO	NO						
11	[]	YES	-	YES	-	NO	YES	NO	NO	NO	NO	NO	NO	YES <sup>341</sup>	-	NO	NO
12	[]	NO	NO	YES	-	NO	YES	NO	NO	NO	NO						
13	[]	NO	YES	YES	-	NO	NO	NO	NO								
14	[]	NO	NO	YES	-	NO	NO	NO	NO								
15	[]	NO	NO	YES	-	NO	YES	NO	NO	NO	NO						
16	[]	NO	NO	YES	-	NO	NO	NO	NO								
17	[]	NO	NO	YES	-	NO	NO	NO	NO								
18	[]	NO	YES	YES	-	NO	YES	NO	NO	NO	NO						
19	[]	NO	YES	YES	-	NO	YES	NO	NO	NO	NO	NO	YES	NO	NO	NO	NO
20	[]	NO	YES	YES	-	NO	YES	NO	NO	NO	NO						
21	[]	NO	YES	YES	-	NO	NO	NO	NO								

<sup>&</sup>lt;sup>340</sup> [...] did not provide response to this question. The Commission, used information provided by Kings Laundry and the Customers in order to identify which multi-site hospitality customers are currently supplied or were supplied previously by [...]. In any case, the Commission notes that [...] does not provide outsourced flat linen rental and maintenance services in [...].

<sup>&</sup>lt;sup>341</sup> It is the Commission's understanding that [...].



			[]	[.	]	[.	]	[.	]	[]	340	[	.]	[]	]	[	.]
iD	Hotel group	Supply now (SN)	Supplied previously (SP)	SN	SP	SN	SP	SN	SP	SN	SP	SN	SP	SN	SP	SN	SP
22	[]	YES	-	YES	-	YES	-	NO	YES	NO	NO	NO	NO	NO	NO	NO	NO
23	[]	NO	NO	YES	-	NO	NO	NO	NO	NO	NO	NO	NO	NO	NO	NO	NO
24	[]	YES	-	YES	-	YES	-	YES	-	YES	-	NO	NO	NO	NO	NO	NO
25	[]	NO	NO	NO	NO	NO	NO	NO	NO	NO	NO	NO	NO	YES	-	NO	NO

Source: The Commission's analysis of information provided by linen laundries



- Competition and Consumer Protection Commission
- 4.125 Table 8 illustrates the following:
  - the majority of multi-site hospitality customers (17 out of 25) currently have a single supplier of outsourced flat linen rental and maintenance services.
  - Kings Laundry currently supplies outsourced flat linen rental and maintenance services to the vast majority of multi-site hospitality customers ([...] out of 25);
  - Berendsen and Celtic Linen supply (or used to supply) outsourced flat linen rental and maintenance services to a significant number of multi-site hospitality customers ([...] and [...] respectively) – those multi-site hospitality customers that were previously serviced by these two linen laundries are currently purchasing outsourced flat linen rental and maintenance services from Kings Laundry;
  - Only Kings Laundry and one competitor ([...]) are the sole suppliers of outsourced flat linen rental and maintenance services to multi-site hospitality customers. All other linen laundries (including Berendsen and Celtic Linen) [...] supply outsourced flat linen rental and maintenance services to those multi-site hospitality customers that prefer to have several suppliers (as opposed to a single national supplier); and
  - to date, [...] has not supplied outsourced flat linen rental and maintenance services to multi-site customers.
- 4.126 The Commission also compared linen laundries by examining their customers' sizes (in terms of turnover).<sup>342</sup> The Commission notes that Kings Laundry has [...]. However, the Commission notes that no definitive overall conclusion can be drawn, because information was not available from all linen laundries.
- 4.127 Finally, the Commission considered whether having facilities in more than one location (as is the case with the Parties and Celtic Linen) accorded an advantage over linen laundries having only one facility. For example, [...] is of the view that not having a second facility in a different location was one of the reasons why they were unsuccessful in bidding for [...] contract.<sup>343</sup>
- 4.128 The Commission considers that having a facility in more than one location may be an advantage in terms of having better contingency planning. Contingency planning is an important criterion for organisations in the hospitality sector when selecting providers of flat linen rental and maintenance services. Also, the Amárach Survey indicates that overall over 4 in 5 hospitality customers said that good contingency plans were either essential or very important.<sup>344</sup> However, the use of multiple suppliers to some extent provides an alternative means of contingency planning for hospitality customers.
- 4.129 The Commission's view is that, in the hospitality market, both customers and competitors have largely overcome any disadvantages associated with geographic

<sup>&</sup>lt;sup>342</sup> Only the Parties and Celtic Linen were able to provide the requested information.

<sup>&</sup>lt;sup>343</sup> Stated by [...] during the Commission's telephone call with [...] on 20 February 2019.

<sup>&</sup>lt;sup>344</sup> Slide 36 of the Amárach Survey. The Commission notes that this slide refers to responses from all respondents rather than just hospitality customers.



location. Some customers have chosen to use multiple suppliers of flat linen, so minimising dependence on any one facility. There is evidence of competitors subcontracting work in areas which are less financially attractive for them to supply, which extends their geographic reach.

## Tender participation data

- 4.130 The Commission's investigation revealed that customers in the hospitality market tend to issue RFQs or engage in bilateral negotiations rather than issue formal tenders. The exception to this is higher education institutions and other public sector organisations that require outsourced flat linen rental and maintenance services and are obliged to abide by tendering rules (e.g., the Houses of the Oireachtas Commission). Thus, information on tender participation is very limited.
- 4.131 Nevertheless, according to information provided by Berendsen,<sup>345</sup> in recent years the Parties competed against each other for contracts of two major customers in the hospitality market (in terms of potential annual turnover):
  - a) [...] in 2017; and
  - b) [...] in 2018.
- 4.132 In this regard, during the meeting with the Commission, Berendsen stated that it would always expect to compete for large hospitality contracts.<sup>346</sup>
- 4.133 In general, the Commission observes that more recently, Berendsen's attempts to win [...] have not been successful. For example, Berendsen was not successful in its bids made to [...] (won by Kings Laundry) or [...] (won by [...]). Berendsen's declining turnover in the hospitality sector also reflects Berendsen's market position. The Commission's view is supported by a comment of one hospitality customer<sup>347</sup> which stated that Berendsen is not well known for serving customers in the hospitality sector. Thus, this customer noted that it might [...].
- 4.134 In summary, the Parties tend to participate in ITTs/RFQs issued by organisations in the hospitality sector. However, to date, Berendsen's participation in ITTs/RFQs appears to have little impact on Kings Laundry's probability of winning contracts.

## Analysis of the Parties' customer loss data

4.135 The Commission notes that the Parties provided a relatively limited number of instances where their hospitality customers switched to another supplier in the period from 1 January 2014 to 30 June 2018.<sup>348</sup> The analysis of customer loss data is therefore based on a small dataset.

<sup>&</sup>lt;sup>345</sup> Information provided by Berendsen in response to Question 22 of the First RFI.

<sup>&</sup>lt;sup>346</sup> The Commission's meeting with Berendsen on 30 January 2019.

<sup>&</sup>lt;sup>347</sup> See the Commission's record of a phone call with [...] dated 15 January 2019.

<sup>&</sup>lt;sup>348</sup> While in response to question 25 of the First RFI, Kings Laundry provided numerous examples where it attempted to win hospitality customers, the Commission was unable to unambiguously verify the identities of participating competitors that also attempted to win cited hospitality customers.



- 4.136 The extent to which the Parties have won or lost customers between each other provides a useful indicator of the closeness of competition between them.<sup>349</sup> The Commission explored switching patterns over time using information provided by the Parties. Both Parties provided a list of customers that switched from them in the period from 1 January 2014 to 30 June 2018. The Commission estimated switching ratios or diversion ratios between the Parties in terms of: (i) the number of switching customers ("diversion ratio")<sup>350</sup> and (ii) lost contract value of switching customers ("weighted diversion ratio").<sup>351</sup>
- 4.137 In performing this analysis, the Commission included only customers where the parties were able to identify who those customers switched to (e.g., name of competitor or OPL). The Commission also excluded customers that had become bankrupt (i.e., no longer required outsourced flat linen rental and maintenance services) and customers who received service cancellation letters (as opposed to switching on their own initiative).
- 4.138 Based on the analysis of Berendsen's hospitality customers' loss data for outsourced flat linen rental and maintenance services, the Commission found that Kings Laundry was the closest competitor to Berendsen by a significant margin. In particular, Table 9 indicates that Kings Laundry won [...]% of the customers lost by Berendsen during the examined period and gained [...]% of turnover lost by Berendsen in the same period. In comparison, Celtic Linen won [...]% of customers and [...]% of turnover lost by Berendsen. The Commission also notes that Berendsen used to provide outsourced flat linen rental and maintenance services to several hotel groups and hospitality procurement groups such as [...]<sup>352</sup>. All these customers are currently purchasing services from Kings Laundry.<sup>353</sup>

		20	)14	20	015	20	016	20	17	2016 2017 Jan-Jun 2018		Total	
	Linen Laundr Y	Nr.	%	Nr.	%	Nr.	%	Nr.	%	Nr.	%	Nr.	%
In terms of custom	Kings Laundr Y	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
er number	Celtic Linen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
S	Other	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	OPL	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%

Table 9: Berendsen's diversion ratios in the hospitality market, 2014 - 30 June 2018 [ $\gg$  REDACTED]

<sup>&</sup>lt;sup>349</sup> In this regard see paragraph 4.20 of the Commission's Merger Guidelines.

<sup>&</sup>lt;sup>350</sup> Diversion ratios were calculated by considering, on the numerator, the total number of a party's (e.g. Berendsen) lost contracts to a certain competitor (e.g., Kings Laundry) within a certain year (e.g. 2017) and on the denominator, the total number of a party's contracts lost within the same year.

<sup>&</sup>lt;sup>351</sup> Weighted diversion ratios were calculated by considering, on the numerator, the total number of a party's (e.g. Berendsen) lost contracts to a certain competitor (e.g. Kings Laundry) times their total value within a certain year (e.g. 2017) and on the denominator, the total number of a party's contracts lost times their total value for the same year.

<sup>&</sup>lt;sup>352</sup> Berendsen's response dated 20 March 2019 to the Commission's informal query dated 19 March 2019.

<sup>&</sup>lt;sup>353</sup> Kings Laundry's response to Question 23 of the Second RFI.



	Total	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%
		€	%	€	%	€	%	€	%	€	%	€	%
In terms	Kings												
of	Laundr	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
contract	У												
value	Celtic	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
(in	Linen	[]	[],	[]	[],•	[]	[],•	[]	[], •	[]	[], •	[]	[],,,
€000s)	Other	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	OPL	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Total	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%

Source: The Commission's analysis of Berendsen's data on lost hospitality customers

- 4.139 The same analysis was performed using information provided by Kings Laundry. The Commission found that Berendsen exerted a minor competitive constraint on Kings Laundry in the hospitality sector. In particular, Table 10 indicates that Berendsen did not win any of the hospitality customers lost by Kings Laundry during the examined period. In comparison, Premier Linen won [...]% of customers who switched and gained [...]% of turnover lost by Kings Laundry. The Commission notes that [...]% of customers and [...]% of turnover lost by Kings Laundry. The Commission notes that [...]% of customers that left Kings Laundry switched to self-supply, but notes that weighted diversion ratio to OPLs is significantly affected by one former customer of Kings Laundry (i.e., Merrion Hotel).
- 4.140 The Commission notes that Kings Laundry's relatively high customer diversion ratio to OPLs should be considered in the context of low numbers of customers overall that have left Kings Laundry. In the period from 1 January 2014 to 30 June 2018 Kings Laundry, on average, had [...] hospitality customers and €[...] of turnover. [...] hospitality customers that switched to OPLs represent just [...]% of Kings Laundry's customer base and the turnover lost from these customers represents just [...]% of Kings laundry's average turnover. Thus, Kings Laundry's diversion ratio to OPLs does not mitigate the Commission's view that for most customers a switch to OPL is not a viable option. The competitive pressure associated with customers' ability to switch to self-supply is discussed in paragraphs 4.201 to 4.202 below.<sup>354</sup>

		2	014	2(	015	2	016	2	017		n-Jun 018	Тс	otal
	Linen Laundry	Nr.	%	Nr.	%								
In terms	Berendsen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
of customer	Celtic Linen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
numbers	Premier Linen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%

Table 10: Kings Laundry's diversion ratios in the hospitality market, 2014 - 30 June 2018 [ $\gg$  REDACTED]

<sup>&</sup>lt;sup>354</sup> Please also see Appendix 4 which sets out organisations with OPLs.

	Carraig Linen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Linencare	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Other	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	OPL	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Total	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%
		€	%	€	%	€	%	€	%	€	%	€	%
In terms	Berendsen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
of contract	Celtic Linen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
value (in €000s)	Premier Linen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Carraig Linen	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Linencare	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Other	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	OPL	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%	[]	[]%
	Total	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%	[]	100%

Source: The Commission's analysis of Kings Laundry's data on lost hospitality customers

4.141 In summary, the switching data shows that the large majority of hospitality customers and revenue lost by Berendsen was lost to Kings Laundry. On the other hand, the majority of hospitality customers and revenue lost by Kings Laundry was lost to [...].<sup>355</sup> The Commission notes the Parties' view that diversion can only be considered where "..whenever Berendsen lost to Kings Laundry, it was the second choice bidder for that customer".<sup>356</sup> As the Commission has set out in paragraphs 4.39 to 4.52, this simplified description of bidding processes ignores the strategic interaction between bidders and the role played by the number of competitors in such processes. Therefore the ranking of bidders is not the sole factor in determining the competitive constraint posed by competing bidders. Rather, it is also important to establish how their presence as independent bidders affects the bids of all bidders.<sup>357</sup> In any event, based on all of the available evidence, the Commission is satisfied that, to date, Berendsen has posed a very limited constraint on Kings Laundry in the hospitality market and it is unlikely that Berendsen's participation or expected participation in ITT's/RFQs would have a major effect on Kings Laundry's bids.

#### The Amárach Survey

- 4.142 The Amárach Survey included questions designed to examine the closeness of competition between linen laundries in the hospitality market.
- 4.143 When asked about the suppliers who they considered in tenders and negotiations, hospitality sector respondents indicated that Berendsen and Kings Laundry were closely competing with each other. In particular the Amárach Survey results show that:

<sup>&</sup>lt;sup>355</sup> When OPL is not taken into account.

<sup>&</sup>lt;sup>356</sup> Frontier Economics Report, paragraph 4.21.

<sup>&</sup>lt;sup>357</sup> This view is confirmed in the DotEcon report, slide 9



- All hospitality respondents who switched from Berendsen switched to Kings Laundry while no hospitality respondents who switched from Kings Laundry, switched to Berendsen;<sup>358</sup>
- b) Hospitality respondents were asked what they would do if their supplier of outsourced flat linen rental and maintenance services raised its prices for the service by between 5% and 10% and everything else remained the same (price diversion). 39% of the hospitality respondents said they would use another linen laundry provider instead.<sup>359</sup> 18% said they would 'shop around' and 17% said they would stay with their current linen laundry provider. 6% said they would re-tender and just 2% said they would consider self-supply. However, of those who said they would switch, when asked who they would switch to 88% said they didn't know.<sup>360</sup>
- 4.144 Moreover, most hospitality respondents said that they have not become aware of any new linen laundries since they last procured outsourced flat linen rental and maintenance services.<sup>361</sup>
- 4.145 To conclude, although sample sizes are too small to allow the Commission to put significant weight on individual figures, the following broad findings can be drawn from the results of the Amárach Survey:
  - a) Kings Laundry holds an influential position in the hospitality market;
  - b) Kings Laundry and Celtic Linen are seen as the two main providers of outsourced flat linen rental and maintenance services to hospitality customers with other credible alternatives being OCL, Linencare, Premier Linen and to a lesser extent Berendsen; and
  - c) there is evidence to indicate that linen laundries other than Berendsen and Kings Laundry are widely known in the hospitality sector.

## Conclusion on closeness of competition

- 4.146 The evidence presented in paragraphs 4.120 to 4.145 above, in itself, does not enable the Commission to conclude that the Proposed Transaction is unlikely to lead to an SLC in the hospitality market. The Commission notes that the Proposed Transaction will result in the merged entity having a high market share in the already highly concentrated hospitality market. Kings Laundry's rapid growth was mainly at the expense of Berendsen and Celtic Linen. Thus, the Proposed Transaction will eliminate an aggressive competitor which exerts a significant competitive constraint on both of these linen laundries as well as other linen laundries.
- 4.147 The nearest competitor (Celtic Linen) will be approximately [...] of the size of the merged entity on the basis of H1 2018 market shares in the hospitality market.<sup>362</sup> While there are several other providers in the hospitality market, they are geographically focused,

<sup>&</sup>lt;sup>358</sup> See slide 31 of the Amárach Survey.

<sup>&</sup>lt;sup>359</sup> See slide 53 of the Amárach Survey.

<sup>&</sup>lt;sup>360</sup> See slide 54 of the Amárach Survey.

<sup>&</sup>lt;sup>361</sup> See slide 47 of the Amárach Survey.

<sup>&</sup>lt;sup>362</sup> See Table 3 above.



and tend to engage in smaller contracts that are of lesser interest to the larger linen laundries.

4.148 While Berendsen does not appear in itself to be an important and significant competitor, given the strength of the merged entity in the hospitality market, the Commission has gone on to consider whether existing competitors would act as an effective competitive constraint following the implementation of the Proposed Transaction.

### Constraints from competitors following the implementation of the Proposed Transaction

- 4.149 In order to assess the strength of competitive constraints from the remaining competitors on the merged entity, the Commission has followed the CCPC's Merger Guidelines<sup>363</sup> and considered the following:
  - a) whether there is substantial competition from remaining competitors;
  - whether remaining competitors have sufficient production capacity to increase output to respond to the exercise of market power by the merged entity; and
  - c) whether remaining competitors have a strong incentive to compete in order to exert a sufficient competitive constraint on the merged entity.<sup>364</sup>

### Constraints exerted by Celtic Linen

- 4.150 Celtic Linen is active in the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State.<sup>365</sup> On the basis of the Commission's market share calculations,<sup>366</sup> Celtic Linen accounted for [10-20]% of the hospitality market in H1 2018. In addition, on the basis of information obtained by the Commission, Celtic Linen's average price per linen item in the hospitality market in 2018 was € [...] average price.<sup>367</sup>
- 4.151 In general, the Commission broadly agrees with some of the arguments raised by the Parties in the Report<sup>368</sup> on Celtic Linen. For example, based on information obtained from Celtic Linen, it is evident that Celtic Linen has benefited from [...], the acquisition of Millbrook Linen, [...] and it has gained hospitality customers<sup>369</sup> since exiting examinership. In addition, the Commission broadly agrees that Celtic Linen's financial position has been improving since coming out of examinership.<sup>370</sup> However, the Commission notes that Celtic Linen still made a post-tax loss of approximately €1.6 million in 2017.<sup>371</sup> While the Commission considers that Celtic Linen will continue to compete against the merged entity in the hospitality market post-transaction, the

<sup>&</sup>lt;sup>363</sup> Paragraph 4.11 of the Commission's Merger Guidelines

<sup>&</sup>lt;sup>364</sup> See paragraph 4.11 of the Commission's Merger Guidelines.

<sup>&</sup>lt;sup>365</sup> See Celtic Linen's Response to the First Competitor Questionnaire

<sup>&</sup>lt;sup>366</sup> See Table 3 above.

<sup>&</sup>lt;sup>367</sup> See Table 6 above

<sup>&</sup>lt;sup>368</sup> See paragraph 4.239 below for a summary of the Report on Celtic Linen.

<sup>&</sup>lt;sup>369</sup> An example of such hospitality customers includes the following: [...]. The Commission notes that Celtic Linen has also [...]. Please see meeting minutes with Celtic Linen [...].

<sup>&</sup>lt;sup>370</sup> See meeting minutes with Celtic Linen [...]; phone call record with Celtic Linen [...]; and, Celtic Linen's response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>371</sup> Celtic Linen Directors' Report and Financial Statements for year to 31 December 2017, page 7.

Commission is concerned that Celtic Linen by itself would not exert an effective competitive constraint that would preclude the merged entity from increasing prices and/or reducing service quality.

4.152 Information provided by Celtic Linen indicates that Celtic Linen currently supplies nationwide outsourced flat linen rental and maintenance services to hospitality customers.<sup>372</sup> Also, Celtic Linen currently supplies outsourced flat linen rental and maintenance services to [...] hotel groups in the State out of the 25 listed in Table 8 (and previously supplied [...]). Finally, Celtic Linen informed the Commission that it has [...] capacity, as outlined in Figure 7 and Figure 8 below.<sup>373</sup>

Figure 7: Celtic Linen's Production Capacity of its Wexford facility [*REDACTED*]

Source: The Commission based on information provided by Celtic Linen in response to the Commission's First Competitor Questionnaire.

Figure 8: Celtic Linen's Production Capacity of its Naas facility [× REDACTED]

Source: The Commission based on information provided by Celtic Linen in response to the Commission's First Competitor Questionnaire.

4.153 The Commission notes that the evidence presented above indicates that Celtic Linen is likely to continue to compete in the hospitality market and that it has spare production capacity to be able to respond to a post-merger price increase by the merged entity.

Constraints exerted by OCL

- 4.154 OCL is a private limited company incorporated in 2000 and registered in the State at 1 Abbey Street, Ballinrobe, Co. Mayo. OCL has one laundry facility in the State located in Ballinrobe, Co. Mayo. OCL [...]. OCL [...]<sup>374</sup>, ...].<sup>375</sup>
- 4.155 On the basis of the Commission's market share calculations<sup>376</sup>, OCL accounted for [10-20]% of the hospitality market in H1 2018. On the basis of information obtained by the Commission, OCL's average price per linen item in the hospitality market in 2018 was € [...] average price.<sup>377</sup>
- 4.156 OCL's hospitality customers in the State include both hotels and restaurants located in counties [...]. In 2018, more than [...]% of OCL's hospitality customers in the supply of

<sup>&</sup>lt;sup>372</sup> Celtic Linen's response to question 5 in the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>373</sup> Meeting minutes with Celtic Linen [...]; phone call record with Celtic Linen [...]; and Celtic Linen's response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>374</sup> ISO 9001 is a quality management standard. According to Berendsen, *"It sets out the basis for how a business' quality systems operate and what should be included in those systems."* Berendsen's clarifications of its response to question 19 of the First RFI set out in page 5 of the letter dated 30 November 2018.

<sup>&</sup>lt;sup>375</sup> See OCL's Response to the First Competitor Questionnaire and phone call record with OCL [...].

<sup>&</sup>lt;sup>376</sup> See Table 3 above.

<sup>&</sup>lt;sup>377</sup> See Table 6 above.

outsourced flat linen rental and maintenance services in the State were located in [...].<sup>378</sup> In addition, OCL informed the Commission that [...].<sup>379</sup>

- 4.157 OCL informed the Commission that it currently has [...].<sup>380</sup> There are [...].<sup>381</sup> OCL also noted that [...].<sup>382</sup> Furthermore, OCL indicated that recently [...] and [...] gave business to OCL.<sup>383</sup>
- 4.158 The Commission's view is that OCL is likely to continue to be a viable competitor in the hospitality market. The Commission recognises that OCL has sufficient capacity to respond to any price increases implemented by the merged entity and that it is currently seeking to expand its activities in the hospitality market.

## Constraints exerted by Linencare

- 4.159 Linencare is a private limited company incorporated in 1998 and registered in the State at Currabeg Business Park, Ardee, Co. Louth. Linencare has one laundry facility in the State located in Ardee, Co. Louth. Linencare [...]. Linencare [...].<sup>384</sup>
- 4.160 On the basis of the Commission's market share calculations,<sup>385</sup> Linencare accounted for [0-10]% of the hospitality market in H1 2018. On the basis of information obtained by the Commission, Linencare's average price per linen item in the hospitality market in 2018 ranged from €[...] to €[...], ranking it [...] average price.<sup>386</sup>
- 4.161 Linencare's hospitality customers in the State include both hotels and restaurants located in counties [...]. Linencare informed the Commission that [...].<sup>387</sup>
- 4.162 Linencare indicated to the Commission that its facility is currently operating at approximately [...]% of its total capacity. There are currently [...] in respect of Linencare's laundry facility, i.e., [...]. In addition, [...].<sup>388</sup>
- 4.163 Finally, Linencare informed the Commission that Linencare is achieving an average of [...]% growth per annum and that it is intended that Linencare's turnover per annum increases from its current €[...] to €[...].<sup>389</sup>
- 4.164 The Commission's view is that Linencare will continue to be a viable competitor in the hospitality market. The Commission notes that Linencare has some production capacity that would allow it to respond to some extent to a price increase implemented by the merged entity and that it currently seeks to expand its activities in the hospitality market.

<sup>&</sup>lt;sup>378</sup> See response to question 1 in the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>379</sup> See phone call record with OCL [...].

<sup>&</sup>lt;sup>380</sup> See OCL's response to question 5 and 7 of the Competitor Questionnaire.

<sup>&</sup>lt;sup>381</sup> See phone call record with OCL [...].

<sup>&</sup>lt;sup>382</sup> See phone call record with OCL [...].

<sup>&</sup>lt;sup>383</sup> See the Commission's meeting minutes with OCL [...].

<sup>&</sup>lt;sup>384</sup> See Linencare's Response to the First Competitor Questionnaire and phone call record with Linencare [...].

<sup>&</sup>lt;sup>385</sup> See Table 3 above.

<sup>&</sup>lt;sup>386</sup> See Table 6 above.

<sup>&</sup>lt;sup>387</sup> Furthermore, Linencare noted that [...]. See Linencare's response to the First Competitor Questionnaire and phone call record with Linencare [...].

<sup>&</sup>lt;sup>388</sup> See phone call record with Linencare [...].

<sup>&</sup>lt;sup>389</sup> See phone call record with Linencare [...].



#### Constraints exerted by Premier Linen

- 4.165 Premier Linen was registered in 1999 and has a registered address at 62/63 Boyne Road, Dublin Industrial Estate, Glasnevin, Dublin 11. Premier Linen has only one laundry facility in the State located at 62/63 Boyne Road, Dublin Industrial Estate, Glasnevin. Premier Linen does not [...].<sup>390</sup>
- 4.166 On the basis of the Commission's market share calculations,<sup>391</sup> Premier Linen accounted for [0-10]% of the hospitality market in H1 2018. On the basis of information obtained by the Commission, Premier Linen's average price per linen item in the hospitality market in 2018 was €[...] average price.<sup>392</sup>
- 4.167 Premier Linen's hospitality customers in the State include hotels, B&Bs and restaurants predominantly located in [...]. However, Premier Linen also supplies outsourced flat linen rental and maintenance services albeit to a minimal extent to hospitality customers located in [...]. Premier Linen also informed the Commission that it used to supply the following hotel chains, i.e., [...], as outlined in Table 8 above.<sup>393</sup>
- 4.168 Premier Linen indicated to the Commission that its facility is operating at [...] capacity. For instance, while Premier Linen [...], it is [...]. There is currently [...] in respect of Premier Linen's laundry facility, [...]. There is [...].
- 4.169 The Commission's view is that Premier Linen is likely to continue to be a viable competitor in the hospitality market. However, the Commission notes that Premier Linen has [...]. This would limit the extent to which Premier Linen would be able to respond to some extent to a price increase imposed by the merged entity.

## Constraints exerted by Carraig Linen

- 4.170 Carraig Linen is a private limited company incorporated in 2010 and registered in the State at Unit 23a, Cookstown Industrial Estate Tallaght, Dublin 24. Carraig Linen has one laundry facility in the State located in Unit 23a, Cookstown Industrial Estate, Tallaght, Dublin 24. Carraig Linen [...]. Carraig Linen [...].<sup>394</sup>
- 4.171 On the basis of the Commission's market share calculations,<sup>395</sup> Carraig Linen accounted for [0-10]% of the hospitality market in H1 2018. On the basis of information obtained by the Commission, Carraig Linen's average price per linen item in the hospitality market in 2018 was € [...] average price.<sup>396</sup>
- 4.172 Carraig Linen's hospitality customers in the State include restaurants, golf clubs, student accommodation, small hotels, all located in [...]. All of Carraig Linen's hospitality

<sup>&</sup>lt;sup>390</sup> See Premier Linen Response to the First Competitor Questionnaire and phone call record with Premier Linen [...].

<sup>&</sup>lt;sup>391</sup> See Table 3 above.

<sup>&</sup>lt;sup>392</sup> See Table 6 above.

<sup>&</sup>lt;sup>393</sup> See email from [...].

<sup>&</sup>lt;sup>394</sup> See Carraig Linen's response to the First Competitor Questionnaire and phone call record [...].

<sup>&</sup>lt;sup>395</sup> See Table 3 above. As noted in Table 3 above, Carraig Linen's market share [...].

<sup>&</sup>lt;sup>396</sup> See Table 6 above.



customers in respect of the supply of flat linen rental and maintenance services in the State are located in [...].<sup>397</sup> In addition, Carraig Linen [...].<sup>398</sup>

- 4.173 In addition, information provided by Kings Laundry indicates that it [...].<sup>399</sup> This information suggests that Kings Laundry [...]
- 4.174 Carraig Linen indicated to the Commission that it [...]. The facility is currently operating [...]. In addition, [...].<sup>400</sup> Finally, Carraig Linen informed the Commission that its turnover from the supply of outsourced flat linen rental and maintenance services to hospitality customers has been growing annually from 2014 to date.<sup>401</sup>
- 4.175 The Commission considers that Carraig Linen is unlikely to exert an effective competitive constraint on the merged entity in the hospitality market post-transaction due to Carraig Linen's different customer segment focus.<sup>402</sup> The Commission notes that while Carraig Linen appears to have [...], its ability to respond to a price increase implemented by the merged entity is likely to be limited to a particular customer segment which is not currently primarily served by the merged entity.

## Constraints exerted by Lilliput

- 4.176 Lilliput is a private limited company incorporated in 1939 and registered in the United Kingdom at Suite 5, Ormeau House, 91-97 Ormeau Road, Belfast. Lilliput is active in the supply of flat linen rental and maintenance services to hospitality customers primarily in [...] and to a limited extent in [...]. Lilliput has one laundry facility located at 33 City Business Park, Dunmurry, Belfast. Lilliput [...]. Lilliput currently [...].<sup>403</sup>
- 4.177 On the basis of the Commission's market share calculations<sup>404</sup>, Lilliput accounted for just [0-10]% of the hospitality market in H1 2018. On the basis of information obtained by the Commission, Lilliput's average price per linen item in the hospitality market in 2018 was €[...] average price.<sup>405</sup>
- 4.178 In general, Lilliput's hospitality customers in the State are primarily located in counties [...]. Lilliput informed the Commission that it currently supplies outsourced flat linen rental and maintenance services to [...] hotel groups out of the 25 listed in Table 8. It previously supplied an additional [...] of these hotel groups.<sup>406</sup>
- 4.179 Lilliput indicated to the Commission that it has [...] its production capacity, [...]. In addition, Lilliput informed the Commission that it [...].<sup>407</sup>
- 4.180 The majority of Lilliput's turnover in respect of the supply of outsourced flat linen rental and maintenance services to hospitality customers is generated from customers located

<sup>&</sup>lt;sup>397</sup> See Carraig Linen's response to question 1 of the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>398</sup> See Carraig Linen's response to question 3 of the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>399</sup> Kings Laundry's response to question 30 of the first RFI.

<sup>&</sup>lt;sup>400</sup> See Carraig Linen's response to the First Competitor Questionnaire and phone call record with Carraig Linen [...].

<sup>&</sup>lt;sup>401</sup> See Carraig Linen's response to First Competitor Questionnaire.

<sup>&</sup>lt;sup>402</sup> In this regard, as noted in paragraph 4.173 above, the Commission notes that Kings Laundry [...].

<sup>&</sup>lt;sup>403</sup> See Lilliput's response to the First Competitor Questionnaire and phone call record with Lilliput [...].

<sup>&</sup>lt;sup>404</sup> See Table 3 above.

 $<sup>^{\</sup>rm 405}$  See Table 6 above.

<sup>&</sup>lt;sup>406</sup> See Lilliput's response to question 3 of the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>407</sup> See Lilliput's response to the First Competitor Questionnaire and phone call record with Lilliput [...].



in [...].<sup>408</sup> However, the Commission notes that Lilliput's turnover in respect of the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State has increased annually from 2014 to 2016 although it [...].<sup>409</sup>

4.181 The Commission's view is that Lilliput is likely to continue to be a viable competitor in the hospitality market. While Lilliput [...] that would allow it to respond to a price increase imposed by the merged entity, its presence in the State is very limited (see Table 7 above). The Commission notes that Lilliput's stated intention is to continue to focus on operations in Northern Ireland rather than the State, and that this would limit the extent to which it would act as a constraint on the merged entity.

## Constraints exerted by other remaining linen laundries

- 4.182 The Commission notes that out of the remaining 5<sup>410</sup> competitors contacted by the Commission, 3 indicated that they currently supply outsourced flat linen rental and maintenance services to hospitality customers in the State, i.e., [...].<sup>411</sup> [...] noted to the Commission that it plans on ceasing its linen rental operations shortly in the State.<sup>412</sup> [...] provides outsourced flat linen rental and maintenance services to hospitality customers in the State from its facilities in [...]. [...] activities in the State are relatively minor and in general limited to counties [...].<sup>413</sup> Finally, [...] informed the Commission that it provides outsourced flat linen rental and maintenance services to hospitality customers to just [...] in the State and generated a minimal turnover of less than €[...] in the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State in 2017.<sup>414</sup>
- 4.183 One of the remaining two competitors contacted by the Commission, [...] informed the Commission that it provides [...]<sup>415</sup> on a local basis<sup>416</sup> in [...]. The remaining competitor, [...] informed the Commission that it does not currently supply outsourced flat linen rental and maintenance services to hospitality customers in the State.<sup>417</sup>
- 4.184 The Commission does not consider that these remaining competitors would have the ability to exert a sufficient competitive constraint on the merged entity and thus, are unlikely to credibly constrain the merged entity's ability and incentive to raise prices/reduce service quality.

## Conclusion on constraints from competitors

4.185 The Parties are facing a number of competitors in the hospitality market. On the basis of the most recent market share figures obtained by the Commission in respect of the hospitality sector, i.e., for the period January 2018 to June 2018 the largest competitors

<sup>&</sup>lt;sup>408</sup> See Lilliput's response to question 1 in the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>409</sup> See Lilliput response to First Competitor Questionnaire.

<sup>&</sup>lt;sup>410</sup> These competitors are as follows: [...].

<sup>&</sup>lt;sup>411</sup> See [...] response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>412</sup> In addition, [...] informed the Commission that they are currently only renting to 2 small customers with a view to utilising the remaining stock. Please see [...] response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>413</sup> See [...] response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>414</sup> See [...] response to question 1 and 9 in the First Competitor Questionnaire.

<sup>&</sup>lt;sup>415</sup> See [...] response to question 10 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>416</sup> [...] informed the Commission that its operation currently serves customers located [...]. Please see [...] response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>417</sup> See [...] response to the First Competitor Questionnaire.

(in terms of market share) are as follows: Celtic Linen ([10-20]% market share), OCL ([10-20]% market share), Linencare's ([0-10]% market share) and Premier Linen ([0-10]%). Celtic Linen's market share declined significantly over the period 2014 to 2017, but has increased in H1 2018, [...] by its acquisition of Millbrook Linen's customers.

4.186 Notwithstanding the fact that the largest competitor in the hospitality market, i.e., Celtic Linen has increased its market share from just over [10-20]% to [10-20]% between 2017 and H1 2018 (which [...] reflects its acquisition of Millbrook Linen), Celtic Linen's market share remains well below the merged entity's [50-60]% share in the hospitality market. However, the Commission's market investigation indicated that following the implementation of the Proposed Transaction, in addition to Celtic Linen, the merged entity will face other linen laundries that have grown their market share in recent years such as OCL and Linencare. These linen laundries have expressed their intention to continue growing their turnover. The Commission considers that, while no individual linen laundry is likely to exercise a sufficient constraint, the Parties' competitors collectively would be likely to exert an effective competitive constraint on the merged entity. In paragraphs 4.187 to 4.208 below, the Commission examines whether there are any countervailing factors that would also prevent the merged entity from profitably increasing prices/reducing service quality in the hospitality market.

## Likelihood of new entry

4.187 In this section, the Commission assesses whether market entry or expansion in the hospitality market might prevent an SLC and as such, considers whether such entry or expansion would be timely, likely and sufficient.

## Barriers to Entry and Expansion

- 4.188 The Commission's investigation identified some barriers to entry and expansion in the hospitality market.
- 4.189 In relation to greenfield new entry, the Commission notes that a considerable financial investment<sup>418</sup> (e.g., capital costs for premises, linen, equipment, (e.g., CBW), civil engineering services (drainage, steam, water and ventilation systems) etc.) is required for a new entrant to enter the hospitality market. For example, [...] noted that its current laundry facility cost circa €[...] to set up and also noted that a new laundry facility would cost approximately €[...] to set up.<sup>419</sup> In comparison, [...] estimated that it would cost circa €[...] to set up a new laundry facility.<sup>420</sup> Smaller competitors quoted set up cost estimates of between €150,000<sup>421</sup> (on a small scale) and €300,000.<sup>422</sup> Finally, the

<sup>&</sup>lt;sup>418</sup> See [...] response to question 24 of the First Competitor Questionnaire [...] response to question 24 of the First Competitor Questionnaire and phone call record with [...].

<sup>&</sup>lt;sup>419</sup> See phone call record with [...].

<sup>&</sup>lt;sup>420</sup> See [...] response to question 26 of the First Competitor Questionnaire. The Commission notes that this estimation excludes the cost of leasing/purchasing the relevant premises as well as transport costs. In addition, this estimation was based on the cost of second hand machinery.

<sup>&</sup>lt;sup>421</sup> See [...] response to question 24 of the First Competitor Questionnaire.

<sup>&</sup>lt;sup>422</sup> See [...] response to question 24 of the First Competitor Questionnaire and [...] response to question 25 of the First Competitor Questionnaire, respectively. The former estimation by [...] concerned an initial outlay and the latter estimation by [...] involved second hand equipment and noted that all delivery vehicles and most equipment could be leased or financed.

Commission notes that Kings Laundry's Cork facility required an initial investment of approximately  $\in$ [...] with an additional  $\in$ [...] spent in [...].<sup>423</sup>

- 4.190 The Commission notes that prior experience/reputation is important for a new entrant when entering the hospitality market.<sup>424</sup> For instance, competitors in the hospitality market noted that reputation and prior experience is a key factor for customers at the following stages: when deciding whether the potential supplier is in a position to service them; when deciding to move from one supplier to another; and when they award contracts for "*new build*" projects.<sup>425</sup> However, [...] noted that a number of new entrants in the last 10 years were able to grow very quickly from a small base.<sup>426</sup>
- 4.191 Finally, the Commission notes that economies of scale are important in the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State, in particular when attempting to win larger hospitality customers in the State.<sup>427</sup> For instance, [...] noted that the linen rental business requires volume to deliver margin and as such, it is essential to have a sufficient customer base.<sup>428</sup> As a result, such scale economies may limit entry to small scale entry which would be unlikely to act as a timely and sufficiently effective competitive constraint on the merged entity.

### Evidence of Market Entry or Expansion

Coimisiún um Iomaíocht agus

Cosaint Tomhaltóirí

- 4.192 The Commission notes that the ability for wash-only laundrettes/dry cleaners to expand into the hospitality market is plausible. For instance, Kings Laundry originally operated as a wash-only laundrette before entering the hospitality market. In addition, the owners of both [...] first operated dry cleaning businesses before later expanding into the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State.<sup>429</sup>
- 4.193 The Commission notes that there is also evidence of market entry from operators active in other sectors in the State. For example, [...] previously operated in the clothing manufacturing business before moving into the hospitality market when it closed the clothing manufacturing business.<sup>430</sup> [...] opened its facility in the same location as the clothing manufacturing business.
- 4.194 The Commission notes that entry into the hospitality market in the last five years has been small scale and primarily on a regional/local basis. For instance, [...] (incorporated in [...]) does not currently supply outsourced flat linen rental and maintenance services to hospitality customers outside of the Greater Dublin Area and held a minimal [0-10]% market share in the hospitality market in H1 2018.<sup>431</sup>

<sup>&</sup>lt;sup>423</sup> See Kings Laundry's response to question 13 of the First RFI.

<sup>&</sup>lt;sup>424</sup> See [...] response to question 24 in the First Competitor Questionnaire; [...] response to question 24 in the First Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire; [...] response to question 10 in the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>425</sup> See [...] response to question 10 in the Second Competitor Questionnaire and [...] response to question 10 in the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>426</sup> See [...] response to question 15 in the Second Competitor Questionnaire.

<sup>&</sup>lt;sup>427</sup> See phone call record with [...].

 $<sup>^{\</sup>rm 428}$  See [...] response to question 24 in the First Competitor Questionnaire.

<sup>&</sup>lt;sup>429</sup> See [...] response to question 24 in the First Competitor Questionnaire and phone call record with [...].

<sup>&</sup>lt;sup>430</sup> See phone call record with [...].

<sup>&</sup>lt;sup>431</sup> See [...] response to question 1 of the Second Competitor Questionnaire and please see Table 3 above for market share analysis.



### Conclusion on likelihood of market entry

4.195 The Commission notes that barriers to entry and expansion in the hospitality market include both financial barriers and non-financial barriers, as discussed above. While there has not been any significant market entry in the last 5 years, the available evidence indicates that entry into the hospitality market by wash-only laundrettes is plausible. The Commission's view is that barriers to entry appear to be relatively modest and surmountable, and that this should be taken into account when assessing the risk of SLC.

#### Countervailing buyer power

- 4.196 The Commission recognises that, in some circumstances, a customer may, because of its position in the market, be able to successfully resist supplier price increases, and may possess sufficient negotiating strength to enable it to constrain the behaviour of a supplier.<sup>432</sup> The assessment of CBP considers the extent to which buyer power postmerger would be present and sufficiently effective to prevent an SLC.
- 4.197 The Commission's Merger Guidelines set out a non-exhaustive list of types of issues to be considered, and types of evidence to be evaluated.<sup>433</sup> For the purposes of this Determination, the Commission considers that the following issues are the most relevant:
  - a) a customer's ability to switch away from the merging Parties and/or switch to alternative suppliers (discussed in paragraphs 4.198 to 4.199 below);
  - b) the number, nature and viability of alternative options available (discussed in paragraphs 4.200 to 4.202 below); and
  - c) the size and significance of individual customers (discussed in paragraphs 4.203 to 4.207 below).

## The ability of customers to switch supplier

4.198 The Commission's investigation revealed that customers in the hospitality sector are able to switch supplier of outsourced flat linen rental and maintenance services relatively easily. In this regard, the majority of Customers interviewed by the Commission noted that it was relatively easy to switch. Also, the Amárach Survey indicated that 62% of hospitality respondents thought that switching suppliers was easy.<sup>434</sup> The Commission also observed a number of switching examples with several organisations in the hospitality sector switching between linen laundries. However, despite the claims of low barriers, approximately 29% of respondents indicated that they had not procured laundry services in the past five years.<sup>435</sup>

<sup>&</sup>lt;sup>432</sup> See paragraph 7.1 of the Commission's Merger Guidelines.

<sup>&</sup>lt;sup>433</sup> See paragraph 7.10 of the Commission's Merger Guidelines.

<sup>&</sup>lt;sup>434</sup> Slide 46 of the Amárach Survey.

<sup>&</sup>lt;sup>435</sup> Slide 26 of the Amárach Survey.



4.199 Overall, the Commission is of the view that switching barriers between the customers of linen laundries in the hospitality sector are relatively low.

## The number, nature and viability of alternative options available

- 4.200 As noted in paragraph 4.77, the Commission found that the hospitality market is highly concentrated. Kings Laundry serves the majority of the largest hospitality customers in the State. Celtic Linen, OCL, Berendsen, Linencare and Premier Linen are also providing outsourced flat linen rental and maintenance services and can be considered as viable options. Hence, following the implementation of the Proposed Transaction, hospitality customers currently outsourcing their flat linen rental and maintenance requirements could: a) choose between the remaining linen laundries; or b) in a limited number of cases set up an OPL and self-supply their flat linen requirements.
- 4.201 The Commission's consideration of the Parties' arguments in relation to the viability of self-supply as an alternative is summarised in paragraph 4.97. The Commission reiterates its view that a relatively limited set of examples whereby former customers of the Parties switched to OPLs does not in itself indicate that self-supply is a significant competitive constraint that would prevent the merged entity from increasing prices following the implementation of the Proposed Transaction.
- 4.202 Nevertheless, the Commission acknowledges that the option of switching to self-supply is a viable option for a limited number of customers as demonstrated by examples of hotels that have switched to OPLs. For example, hospitality customers with sufficient space for OPL at their premises could threaten the merged entity in response to a price increase/service quality reduction.

## The size and significance of individual customers

- 4.203 The Commission found that the purchase of outsourced flat linen rental and maintenance services in the hospitality market is typically characterised by RFQs and/or bilateral negotiations.
- 4.204 The Commission notes that the top 5 customers of the merged entity would collectively account for around [...]% of its total turnover from the provision of outsourced flat linen rental and maintenance services to hospitality customers in the State. The largest customer is [...] which would account for [...]% of the merged entity's turnover. The merged entity is therefore not likely to be overly dependent on any individual customer which, in turn, weakens the negotiating power of individual customers.
- 4.205 The Commission observed that large hotel groups are able to exert some pressure on linen laundries over prices they pay for outsourced flat linen rental and maintenance services and also has discretion on how their purchases of outsourced flat linen rental and maintenance services are distributed between linen laundries. In this regard, the Commission notes that Kings Laundry provided evidence indicating its customers' reluctance to [...].<sup>436</sup> In summary, this evidence highlights that:
  - a) [...];

<sup>&</sup>lt;sup>436</sup> Kings Laundry's response to Question 33 of the First RFI.



- b) [...];<sup>437</sup>
- c) [...]; and
- d) [...].
- 4.206 The Commission understands that ultimately, [...]. Information provided by Kings Laundry indicates that at the end of 2018 it had [...] customers compared to [...] at the end of 2017<sup>438</sup> while its EBIT<sup>439</sup> [...] from [...]% in 2017 to [...]% in 2018<sup>440</sup>.The Commission did not find any evidence that large hotel groups have threatened to open OPLs, sponsored market entry, or have threatened to do so in response to [...].
- 4.207 Finally, the Commission notes that [...] and [...] have recently decided to diversify their purchase of outsourced flat linen rental and maintenance services by giving business to [...].<sup>441</sup> This evidence indicates that multi-site hospitality customers are able to secure the purchase of outsourced flat linen rental and maintenance services from multiple sources and are willing to deal with several linen laundries.

## Conclusion on CBP

4.208 The Commission notes that customers in the hospitality sector can choose between several linen laundries with the hospitality market exhibiting relatively low barriers to switching. Some hospitality customers can also credibly set up OPLs. The available evidence indicates that some of the larger hospitality customers appear to be able to negotiate favourable deals with linen laundries. Furthermore, multi-site hospitality customers demonstrated willingness and ability to diversify and multi-source their purchase of outsourced flat linen rental and maintenance services. For these reasons, the merged entity would likely be constrained, to a certain degree, by the reaction of its customers to any unilateral price increase and/or reduction of service quality.

## Overall conclusion on Unilateral Effects in the hospitality market

- 4.209 For the reasons set out above and considering all of the evidence described above, the Commission considers that the Proposed Transaction is not expected to result in significant unilateral effects in relation to the hospitality market. This is based on the following findings:
  - a) to date, Berendsen has posed a very limited competitive constraint on Kings Laundry in the hospitality market;
  - b) the merged entity will continue to face a number of competitors who have grown their turnover in recent years and have excess capacity to be able to constrain the activities of the merged entity;
  - c) the hospitality market is growing, and this will encourage market entry;

<sup>&</sup>lt;sup>437</sup> The Commission interprets this notice as the customer's belief that it will obtain lower prices via tender.

<sup>&</sup>lt;sup>438</sup> Kings Laundry's response to Question 3 of the Second RFI.

<sup>&</sup>lt;sup>439</sup> Earnings before interest and taxes.

<sup>&</sup>lt;sup>440</sup> Kings Laundry's response to Question 1 of the Second RFI.

<sup>&</sup>lt;sup>441</sup> See the Commissions meeting minutes with [...].



- d) barriers to entry and expansion in the hospitality market are relatively low. While there has been limited market entry in the last 5 years, the available evidence indicates that entry into the hospitality market by wash-only laundrettes is plausible;
- customers in the hospitality sector can choose between several linen laundries. The hospitality market exhibits relatively low barriers to switching. There has been evidence of an increasing number of hospitality customers switching and using multiple linen laundries;
- f) the option of self-supply is also available to a limited number of hospitality customers.

# The healthcare market

## The nature of pre-merger competition

- 4.210 In their written and oral submissions, the Parties described the healthcare market as a bidding market, exhibiting the same general characteristics as the hospitality market.<sup>442</sup> As context for discussion below, the Commission refers back to the discussion on the Parties' view of bidding markets and the Commission's assessment of these views set out in paragraphs 4.39 to 4.50.
- 4.211 This section sets out the Commission's analysis of the healthcare market as follows:
  - market structure;
  - market concentration; and
  - competitive effects analysis.

## **Market Structure**

## Views of the Parties

4.212 The Parties noted that, in their view, their market shares do not reflect the importance of within market differences in the healthcare market, thus overstating the degree of overlap and closeness of competition between the Parties.<sup>443</sup>

## Views of the Commission

- 4.213 The Commission does not agree that market share analysis is not relevant, for the reasons set out in paragraphs 4.39 to 4.52 above. The Commission notes that, post transaction, the merged entity would have a very large market share in an already highly concentrated market.
- 4.214 According to the Merger Guidelines:

<sup>&</sup>lt;sup>442</sup> Frontier Economics Report, paragraph 4.4.

<sup>&</sup>lt;sup>443</sup> Parties' Response, paragraph 4.2.56 and Frontier Economics Report, paragraphs 2.6 to 2.8.

"The combined market share of the merging parties, when compared with their respective shares pre-merger, can provide an indication of the change in market power resulting from the merger. Competition concerns are more likely to arise when the merger creates a merged entity with a large market share."<sup>444</sup>

4.215 The Commission has also taken into account guidance from the EC Horizontal Merger Guidelines,<sup>445</sup> which state that:

"Generally, a merger giving rise to such non-coordinated effects would significantly impede effective competition by creating or strengthening the dominant position of a single firm, one which, typically, would have an appreciably larger market share than the next competitor post-merger. Furthermore, mergers in oligopolistic markets involving the elimination of important competitive constraints that the merging parties previously exerted upon each other together with a reduction of competitive pressure on the remaining competitors may, even where there is little likelihood of coordination between the members of the oligopoly, also result in a significant impediment to competition. The Merger Regulation clarifies that all mergers giving rise to such non-coordinated effects shall also be declared incompatible with the common market."

4.216 According to the Parties' estimates, on a national level, the Parties' combined market share measured in terms of revenue from the supply of outsourced flat linen rental and maintenance services (having regard to the market definition adopted by the Commission) was [50-60]% with an increment<sup>446</sup> of [10-20]% as a result of the acquisition of Kings Laundry. Table 11 below illustrates the Parties' estimated shares<sup>447</sup> in the healthcare market in the period from 1 January 2017 to 31 December 2017.

	20	17
	€m	%
Berendsen	[]	[40-50]
Kings Laundry	[]	[10-20]
Combined	[]	[50-60]
Increment		[10-20]
Celtic Linen	[]	[40-50]

Table 11: Revenue from the provision of flat linen rental and maintenance services to
healthcare customers in the State, 2017 (the parties' estimates) [ 🎾 REDACTED]

<sup>&</sup>lt;sup>444</sup> Merger Guidelines, paragraph 3.4.

<sup>&</sup>lt;sup>445</sup> Paragraph 25 of the EC Horizontal Merger Guidelines.

<sup>&</sup>lt;sup>446</sup> For the purposes of this Phase 2 Determination, increment refers to incremental market share acquired by Berendsen

<sup>&</sup>lt;sup>447</sup> The Parties included potential revenue from customers currently operating OPLs. The inclusion of potential revenue reduces the Parties' combined market share from [50-60]% to [30-40]%. Such potential revenue is not presented in Table 11, having regard to the relevant product market definition.

Competition and Consumer Protection Commission

	20	17
Total	[]	100

Source: Information submitted by the parties. Percentages may not add up due to rounding.

4.217 The Commission conducted a market share reconstruction and collected the actual sales volumes of all main competitors in the State for the period from 1 January 2014 to 30 June 2018, which is illustrated by Table 12 below. On the basis of the sales volume data of the Parties and their main competitors, in the period from 1 January 2018 to 30 June 2018, the combined market share of the Parties is [60-70]%, with an increment of [10-20]%. This is higher than the Parties' estimate of their combined market share of [50-60]%.<sup>448</sup>

Table 12: Revenue from the provision of flat linen rental and maintenance services to healthcare customers in the State, 2014 - 30 June 2018 [ $\gg$  REDACTED]

	201	4	201	5	201	6	201	7	H1 2018		
	€000s	%	€000s	%	€000s	%	€000s	%	€000s	%	
Berendsen	[]	[40- 50]	[]	[40- 50]	[]	[40- 50]	[]	[50- 60]	[]	[50- 60]	
Kings Laundry	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	[]	[10- 20]	[]	[10- 20]	
Combined	[]	[50- 60]	[]	[50- 60]	[]	[50- 60]	[]	[60- 70]	[]	[60- 70]	
Increment		[0- 10]		[0- 10]		[0- 10]		[10- 20]		[10- 20]	
Celtic Linen	[]	[30- 40]	[]	[30- 40]	[]	[40- 50]	[]	[30- 40]	[]	[30- 40]	
Other <sup>449</sup>	[]	[10- 20]	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	[]	[0- 10]	
Total	[]	100	[]	100	[]	100	[]	100	[]	100	

Source: The Commission's analysis using information provided by the parties and the Competitors.

4.218 The combined market shares of the Parties, in addition to other factors assessed in paragraphs 4.226 to 4.377 below, indicate that, following the completion of the Proposed Transaction, the merged entity will enjoy a very strong position in the healthcare market. In this regard, it should be noted that Berendsen's market share remained over [40-50]% since at least 2014. Celtic Linen (the significant competitor to the merged entity), accounts for [30-40]% of the market. Thus, the Proposed Transaction will increase the gap between Berendsen's and Celtic Linen's market share. The Commission notes the Parties' argument that market shares do not fully reflect the fact that in 2017 and 2018 Berendsen lost contracts with [...] to Celtic Linen, which

<sup>&</sup>lt;sup>448</sup> In the period from 1 January 2017 to 31 December 2017, the Parties' combined market share was [60-70]% with an increment of [10-20]%. The Parties were also of the view that Linencare and OCL provide flat linen rental and maintenance services to nursing homes and CWS-boco supplies scrub suits, but could not provide estimates of their turnover. See page 15 and page 23 of the Merger Notification Form.

<sup>&</sup>lt;sup>449</sup> This group include revenue of [...] and [...].

according to the parties represented a total contract value of approximately  $\in$ [...].<sup>450</sup> In this regard, H1 2018 estimates partly reflect Celtic Linen's revenue from some of these customers. However, the Commission considers that even if it was to discount  $\in$ [...] contract value from Berendsen and add that same amount to Celtic Linen, the merged entity would still have a market share of [60-70]% while Celtic Linen would have a market share of [30-40]%.

- 4.219 Table 12 indicates that both Berendsen and Kings Laundry increased their market share over the relevant period.<sup>451</sup> In fact, Kings Laundry has been steadily increasing its market share during the examined period (from [0-10]% in 2014 to [10-20]% in the first half of 2018). The Commission is of the view that the growth of Kings Laundry's market share, in combination with other factors discussed in paragraphs 4.265 to 4.274 below, indicates that Kings Laundry is an important and significant competitor in the healthcare market.
- 4.220 There is very limited participation in the healthcare market by other linen laundries. CWS-boco continues to provide scrub suit rental and maintenance services to hospitals after it transferred its healthcare flat linen business to Celtic Linen in 2015.<sup>452</sup> Currently, it participates only in those ITTs where scrub suit rental and maintenance services [...]. [...] indicated that it provides outsourced flat linen rental and maintenance services to three nursing home customers<sup>453</sup> in the State while [...] supplies outsourced flat linen rental and maintenance services to two nursing homes. It should also be noted that [...] informed the Commission that it intends to fully exit "rental" services within the next 2 years.<sup>454</sup>

## Conclusion on market shares

- 4.221 The Commission is of the view that market share analysis in the healthcare market is relevant, and significantly contributes to an understanding of the evolution of different linen laundries' market position, and to an understanding of market strength and market dynamics.
- 4.222 Evidence from the Commission's market investigation shows that there are 3 significant suppliers of outsourced flat linen rental and maintenance services to healthcare customers in the State accounting for over [90-100]% of the market. Following the completion of the Proposed Transaction, this will be reduced to 2 significant suppliers of outsourced flat linen rental and maintenance services and the merged entity will enjoy a very strong position in the healthcare market with a market share over [60-70]% and a market share increment of approximately [10-20]% as a result of the Proposed Transaction.

<sup>&</sup>lt;sup>450</sup> See page 23 of the Merger Notification Form.

<sup>&</sup>lt;sup>451</sup> King's Laundry increased its market share by [...]% between 2014 and 30 June 2018, while Berendsen increased its market share by [...]% during the same period.

<sup>&</sup>lt;sup>452</sup> See <u>https://www.cws-boco.ie/en-IE/cws-boco-acquires-parts-celtic-linen-Itd</u>.

<sup>&</sup>lt;sup>453</sup> In a meeting with the Commission, [...] stated that it expects one of these customer to expand. See the Commission's minutes of meeting with [...].

<sup>&</sup>lt;sup>454</sup> Please see [...] response to the Commission's First Competitor Questionnaire dated 10 October 2018.



#### Competition and Consumer Protection Commission

## **Market Concentration**

- 4.223 Market concentration refers to the degree to which production/supply in a particular market is concentrated in the hands of a few large firms. The most commonly used measure of concentration is the Herfindahl-Hirschman Index ("HHI"), which is defined as the sum of the squares of the market shares of all firms participating in the market. According to the Commission's Merger Guidelines,<sup>455</sup> any market with a post-merger HHI greater than 1,000 may be regarded as concentrated and any market with a post-merger HHI greater than 2,000 may be regarded as highly concentrated. In a concentrated market, a change in the pre-merger HHI compared to the post-merger HHI of less than 250 is *"unlikely to cause concern"*. In a highly concentrated market, a change in the post-merger HHI of less than 150 is *"unlikely to cause concern"*. Therefore, if the post-merger HHI is above 2,000 and the change in the HHI is greater than 150, this indicates that firms in that market may be able to exercise market power.
- 4.224 Based on the market share estimates set out in Table 12 above, the Commission is of the view that the healthcare market is highly concentrated with a pre-transaction HHI of [4,099]. Table 13 illustrates that the HHI following implementation of the Proposed Transaction would be over 5,300. Furthermore, the change in the HHI would be approximately 1,249, which, as set out in paragraph 3.10 of the Commission's Merger Guidelines, means that the Commission could not conclude that the Proposed Transaction is unlikely to cause concern and that it should intensify its analysis of the competitive impact of the Proposed Transaction.

	H1 2018
HHI pre-transaction	[4,099]
HHI post-transaction	5,348
HHI delta	1,249

Source: The Commission's analysis using information provided by the parties and parties' competitors.

4.225 In paragraphs 4.226 to 4.377 below, the Commission sets out its competitive effects analysis before reaching its conclusion as to whether the Proposed Transaction would be likely to lead to an SLC in the healthcare market.

## **Competitive Effects Analysis**

4.226 In this section, the Commission sets out its analysis of the likely competitive effects of the Proposed Transaction in the healthcare market. In particular, this section examines the likelihood of unilateral effects occurring following implementation of the Proposed Transaction. The likelihood of co-ordinated effects is considered in paragraphs 4.379 to 4.386 below.

<sup>&</sup>lt;sup>455</sup> Commission's Merger Guidelines, paragraph 3.10



- 4.227 Unilateral effects, as explained in paragraph 4.8 of the Commission's Merger Guidelines, occur when *"a merger results in the merged entity having the ability and the incentive to raise prices at its own initiative and without coordinating with its competitors."* The removal of a competitor of flat linen rental and maintenance services from the healthcare market could lead to consumer harm through higher prices and/or reduced service quality (e.g., loss of frequency of services), particularly in instances where the acquired competitor exerted a significant competitive constraint pre-transaction.
- 4.228 The Commission's Merger Guidelines note the following:

"Competitive constraints on a merged entity will be weaker to the extent that (i) there is an absence of substantial competition from other firms in the market or firms likely to enter in a timely manner, (ii) competitors have insufficient productive capacity to increase output, or (iii) competitors do not have a strong incentive to compete (for example, if they might also benefit from increased prices), also referred to as price accommodation."<sup>456</sup>

4.229 The potential responses of competing suppliers are also relevant in evaluating the merged entity's pricing incentives. Because a merger changes the market structure and eliminates a competitor from the market, it will reduce the constraints faced by both the merged entity and competing suppliers. Competing suppliers may respond to a price rise by the merged firm by raising their own prices. Therefore, the more concentrated a market is and the fewer effective competitors that remain to constrain one another, the more likely it is that a merger will enable the merged entity to raise prices.

## Views of the Parties

## Pre-merger competitive pressure

- 4.230 The Parties expressed the view that the Proposed Transaction will not lead to an SLC in the healthcare market.<sup>457</sup>
- 4.231 The Parties submitted that they face competition from a large number of competitors across a broad range of product/service offerings listed in paragraph 1.4 above. In this regard, the Parties stated that their activities overlap only with respect to flat linen rental and maintenance services.<sup>458</sup>
- 4.232 The Parties argued that their activities are largely complementary. The Parties stated that Berendsen focus on the supply of workwear rental and maintenance services and, to a lesser extent, the supply of outsourced flat linen rental and maintenance services to healthcare customers and Kings Laundry focus on [...]. In support of this argument, the Parties referred to their revenue in each customer sector. The Parties noted that complementarity between the Parties' activities means that, following the implementation of the Proposed Transaction, the merged entity "...will be able to serve a broader range of customers and to a higher standard across the board."<sup>459</sup>

<sup>&</sup>lt;sup>456</sup> See paragraph 4.11 of the Commission's Merger Guidelines.

<sup>&</sup>lt;sup>457</sup> The Parties' Letter, dated 28 May 2019, paragraph 3.

<sup>&</sup>lt;sup>458</sup> See pages 23 to 24 of the Merger Notification Form.

<sup>&</sup>lt;sup>459</sup> See page 24 of the Merger Notification Form.



- 4.233 The Parties claimed that the Commission has conflated Kings Laundry's approach in the hospitality market with its approach in the healthcare market.<sup>460</sup>
- 4.234 The Parties claimed that Kings Laundry's system of dedicated laundry is not innovative, and is not unique to Kings Laundry.<sup>461</sup> Kings Laundry has also stated that it is [...].<sup>462</sup>

### **Closeness of competition**

### The Parties' pre-merger documents<sup>463</sup>

4.235 Berendsen's internal pre-merger notification documents showed that Berendsen perceives [...]. In particular, the Commission observed that Berendsen itself considered this market to [...].<sup>464</sup> Berendsen also considered that [...] while [...].<sup>465</sup> As indicated in paragraph 4.272 below, Berendsen perceived [...].

### The Parties' written and oral submissions

- 4.236 In their written submission, the Parties claimed that the Proposed Transaction would be at least a 4-to-3 merger (including OCL and any other competitors supplying nursing homes) and, if OPL were to be included, a 5-to-4 merger.<sup>466</sup>
- 4.237 The Parties expressed a view that Kings Laundry is not an important player in the healthcare market<sup>467</sup> and that the Parties are not close competitors as evidenced from tender participation analysis.<sup>468</sup> In this regard, the Parties proposed that: "…when assessing the possible unilateral effects or upward pricing pressure emerging from a merger in a bidding market, the relevant measure of the closeness of competition between the Parties is the probability that, for the contracts one of the merging parties bids for and wins, the other merging party is the customer's second choice bidder".<sup>469470</sup>

## Constraints from remaining competitors

4.238 The Parties submitted that the merged entity will face significant competition following the implementation of the Proposed Transaction. In this regard, the Parties referred to existing competitors and, in particular, Celtic Linen which is the main competitor of the Parties in the healthcare sector. In support of this argument, the Parties referred to Berendsen's recent loss of some of its [...] healthcare customers to Celtic Linen.<sup>471</sup> The Parties were of the view that the Commission downplayed the competitive constraint that Celtic Linen will exercise in the healthcare market post-merger.<sup>472</sup>

<sup>&</sup>lt;sup>460</sup> The Parties' Response, paragraphs 4.2.67 to 4.2.68

<sup>&</sup>lt;sup>461</sup> The Parties' Response, paragraph 4.2.69

<sup>&</sup>lt;sup>462</sup> Minute of meeting, 1 February 2019

<sup>&</sup>lt;sup>463</sup> Kings Laundry indicated that it [...]. See Kings Laundry's response to question 10 of the First RFI.

<sup>&</sup>lt;sup>464</sup> See paragraph 4.273 below.

<sup>&</sup>lt;sup>465</sup> For example, see page 2 of the document titled [...] provided by Berendsen in response to question 9 of the First RFI.

<sup>&</sup>lt;sup>466</sup> The Parties' Letter 4 June 2019, paragraph 10.

<sup>&</sup>lt;sup>467</sup> The Parties' Response , paragraphs 4.2.57 to 4.2.63 and 4.267 to 4.2.69 and the Frontier Economics Report, paragraph 4.11

<sup>&</sup>lt;sup>468</sup> The Parties' Response, paragraphs 4.2.64 to 4.2.66, and Frontier Economics Report, paragraphs 4.2.c and 4.8 to 4.24

<sup>&</sup>lt;sup>469</sup> Frontier Economics Report, Paragraph 4.6 b

<sup>&</sup>lt;sup>470</sup> DotEcon was contracted to address specific points raised by the Parties in their written and oral submissions.

<sup>&</sup>lt;sup>471</sup> More specifically, Berendsen noted that in 2017 and 2018 it lost contracts with [...] to Celtic Linen.

<sup>&</sup>lt;sup>472</sup> Parties' Response, paragraphs 4.2.79 to 4.2.91.



- 4.239 In this regard, in the Report on Celtic Linen, the Parties argued that Celtic Linen acts as a strong competitive force in the supply of outsourced flat linen rental and maintenance services to healthcare customers in the State and included the following arguments:
  - the Parties noted that it is imperative to consider Celtic Linen's position in light of the fact that it successfully exited the examinership process. The Parties noted that *"it is inconsistent with the objective of the statutory regime and more specifically the process itself to consider that exiting examinership demonstrates a weakness in the financial position of a company"*,<sup>473</sup>
  - the Parties referenced the IER report and the initiatives identified which should positively impact the company's commercial prospects following examinership, e.g., a new experienced leadership team; acquisition of new healthcare contracts, the disposal of non-core services, among others;<sup>474</sup>
  - the Parties referred to Celtic Linen's investors Causeway Capital and noted that Celtic Linen ultimately has significant financial backing from extremely credible institutions;<sup>475</sup>
  - the Parties referred to Celtic Linen's latest financial statements and noted that Celtic Linen made a marked improvement on its trading performance compared to the previous financial year;<sup>476</sup>
  - the Parties referred to Celtic Linen's recent investments and restructuring efforts, e.g., the outsourcing of logistics, investments in equipment, textiles and systems, acquisition of Millbrook Linen, among others.<sup>477</sup> Finally, the Parties referred to [...] healthcare contracts<sup>478</sup> which Berendsen bid for and lost to Celtic Linen since June 2017 and noted that they expect that Berendsen was the [...] bidder in each of the tenders for those healthcare customer contracts.<sup>479</sup> Ultimately, the Parties indicated that their view is that it is clear that not only is Celtic Linen an effective competitor of Berendsen in providing services to healthcare customers, but it is also Berendsen's closest competitor in this customer segment and thus, would clearly exercise a significant competitive constraint on the merged entity post-transaction.

## Assessment of the likely effect on price and/or quality

4.240 The Parties claim that, post-merger, the Parties "..will only be incentivised to raise prices if there is a sufficient diversion between Kings Laundry and Berendsen." The Parties argue that this diversion is low, and so "..we consider that the Parties are unlikely to have any incentive to raise their prices post-merger."<sup>480</sup>

<sup>&</sup>lt;sup>473</sup> See paragraph 8 in the Report on Celtic Linen.

<sup>&</sup>lt;sup>474</sup> See Section B and specifically paragraph 11 in the Report on Celtic Linen.

<sup>&</sup>lt;sup>475</sup> See paragraph 14 to 19 in the Report on Celtic Linen.

<sup>&</sup>lt;sup>476</sup> See paragraph 20 and 21 in the Report on Celtic Linen.

<sup>&</sup>lt;sup>477</sup> See paragraph 23 to 25 in the Report on Celtic Linen.

<sup>&</sup>lt;sup>478</sup> These contracts are as follows: [...].

<sup>&</sup>lt;sup>479</sup> See paragraph 27 in the Report on Celtic Linen.

<sup>&</sup>lt;sup>480</sup> Frontier Economics Report, paragraph 4.41



4.241 The Parties argued that, in its Assessment, the Commission erred in its evaluation of the merged entity's incentive to increase prices in the healthcare market by *"incorrectly estimating the price differential and mistakenly assuming the mere existence of a price differential is proof of incentive to increase price,"*<sup>481</sup> and that it failed to demonstrate that Celtic Linen would have both the incentive and the ability to follow an increase in price by the merged entity.

### Likelihood of new entry

- 4.242 The Parties are of the view that the merged entity will face a credible threat of entry or expansion. In this regard, the Parties argued that entry or expansion from: (i) linen laundries currently having minimal operations in the healthcare sector and in particular, OCL; (ii) former providers of flat linen rental and maintenance services to healthcare customers (e.g., CWS-boco); (iii) linen laundries operating in the healthcare sector in Northern Ireland (e.g., Lilliput); (iv) linen laundries currently only servicing the hospitality sector in the State; and (v) laundries providing textile rental and maintenance services (except for flat linen) is credible.<sup>482</sup> In the Parties' view, these potential developments would pose a significant constraint on the merged entity's ability to increase prices following the implementation of the Proposed Transaction. The Parties also expressed a view that the Commission was inconsistent in its Assessment in its reliance on evidence and data by citing various barriers to entry and expansion that apply to competitors such as OCL, but not to Kings Laundry.<sup>483</sup>
- 4.243 The Parties argued that providers of flat linen rental and maintenance services to hospitality customers in the State could enter the healthcare market within a relatively short period of time and without significant investment.<sup>484</sup>
- 4.244 The Parties referred to Lilliput and noted that it was a credible potential competitor in the healthcare market due to its significant experience providing such services to the NHS in Northern Ireland (e.g., the Royal Victoria Hospital in Belfast).<sup>485</sup>
- 4.245 The Parties claimed that they faced a credible threat of entry and expansion from *"existing healthcare players who could expand their operations"*.<sup>486</sup> In this regard, the Parties specifically referenced OCL and Linencare and noted that these players already provide a small amount of flat linen to healthcare customers, i.e., nursing homes, and that these competitors could further expand their operations to hospital customers. The Parties also proposed that such competitors are not hindered by capacity constraints. In this regard, the Parties were of the view that the Commission has not placed sufficient weight on [...] expressed intention to expand in the healthcare market and [...] intention to serve nursing homes.<sup>487</sup> In addition, the parties suggested that CWS-boco which provides other services to healthcare customers (e.g., scrub suit maintenance) could easily expand into the healthcare market.

<sup>&</sup>lt;sup>481</sup> Parties' Response, paragraphs 4.2.70 to 4.2.78 and Frontier Economics Report, paragraphs 4.2.d, and 4.25 to 4.42

<sup>&</sup>lt;sup>482</sup> See page 27 of the Merger Notification Form and paragraphs 6.3 to 6.5 of the Frontier Economics Report.

<sup>&</sup>lt;sup>483</sup> Parties' Response, paragraphs 4.2.103 to 4.2.114

<sup>&</sup>lt;sup>484</sup> See section 5.3 of the Merger Notification Form.

<sup>&</sup>lt;sup>485</sup> See section 5.3 in the Merger Notification Form.

<sup>&</sup>lt;sup>486</sup> See section 5.3 in the Merger Notification Form.

<sup>&</sup>lt;sup>487</sup> Parties' Response, Paragraphs 4.211 to 4.214 and Frontier Economic Report, paragraph 6.5.



- 4.246 During a meeting with the Commission, Berendsen provided its views on the importance of experience and reputation in the supply of flat linen rental and maintenance services to healthcare customers in the State.<sup>488</sup> Berendsen considered that the healthcare customer sector is very different to hospitality and more challenging to enter without previous experience, reputation, etc. It was noted by Berendsen that references are quite important in respect of the healthcare sector and that large organisations want assurance that the laundry operations are safe and reputable.
- 4.247 In addition, Berendsen provided the Commission with a list of potential competitors or new entrants that Berendsen considered may start providing flat linen rental and maintenance services to healthcare customers in the State in the next 2 to 3 years, i.e., Johnsons Services Group; K-Bro Linen Inc.; CWS-boco; Lindstrom; Alsco Inc.; Salesianer; and Cintas Corporation.<sup>489</sup> Berendsen expressed a view that such entry could occur through either the acquisition of an existing business or greenfield entry as both options represent viable opportunities for entry. However, Berendsen did not provide any supporting documents or evidence in relation to the timeliness, likelihood or sufficiency of entry by any of these linen laundries.
- 4.248 In respect of the specific costs associated with the healthcare sector, Kings Laundry indicated<sup>490</sup> to the Commission that some level of investment is required when considering an expansion into healthcare sector including:
  - a) healthcare sector specific equipment (e.g., ironing machines for scrub suits);
  - b) protective clothing for the laundry's staff;
  - c) staff training; and
  - d) costs associated with obtaining health & safety management and cleanliness standards e.g. swab testing.
- 4.249 Kings Laundry informed the Commission that it is not currently aware of any potential new entrant in the supply of flat linen rental and maintenance services to healthcare customers in the State. In addition, Kings Laundry noted that such entry is likely to be facilitated by the acquisition of an existing supplier or by greenfield entry. Finally, Kings Laundry noted that entry is less likely in the supply of flat linen rental and maintenance services to healthcare customers than hospitality customers in the State due to the need for accreditations when supplying the healthcare customers.<sup>491</sup>

<sup>&</sup>lt;sup>488</sup> Commission's meeting with Berendsen on 30 January 2019, see page 4 of minutes of meeting with Berendsen of 30 January 2019.

<sup>&</sup>lt;sup>489</sup> See response to question 22 in Berendsen's response to the Second RFI.

<sup>&</sup>lt;sup>490</sup> See Kings Laundry clarifications in relation to its response to Question 21 of the First RFI.

<sup>&</sup>lt;sup>491</sup> See response to question 19 in Kings Laundry's response to the Second RFI.



#### Countervailing Buyer Power

- 4.250 The Parties considered that customers in the healthcare sector (e.g., the HSE) are highly sophisticated buyers and have significant buyer power and can credibly avail of the following options in response to the merged entity's attempt to increase prices:<sup>492</sup>
  - a) switch to alternative suppliers of flat linen rental and maintenance services; or
  - b) switch to self-supply via OPLs.
- 4.251 In the Report on OPLs, the Parties noted the precedent of NCAs (including the Commission) considering the option of self-supply during the assessment of competition.<sup>493</sup> The Parties noted that a number of customers in the healthcare sector currently self-supply flat linen rental and maintenance services. In relation to customers that do not have OPLs, the Parties reiterated that the costs associated with setting up an OPL are relatively small. The Parties noted that customers can avail of second hand machinery, citing providers and purchasers of such machinery in the State. The possibility to source equipment from suppliers located outside the State was also highlighted. Thus, the Parties were of the view that customers in the healthcare sector that are outsourcing flat linen rental and maintenance services can exercise significant CBP by threatening to switch to self-supply.
- 4.252 The Parties noted that barriers to switching supplier are relatively low. The Parties referred to the relatively short durations of their contracts with customers<sup>494</sup> and argued that the costs of switching suppliers at the end of the contract period are low.

## Other factors

- 4.253 The Parties alleged that, in its Assessment, the Commission ignored and/or failed to give sufficient weight to views expressed by healthcare customers and competitors. The Parties argued that third parties do not have concerns regarding the Proposed Transaction.<sup>495</sup>
- 4.254 Finally, the Parties stated that in its Assessment the Commission misunderstood the context and/or meaning of their internal documents.<sup>496</sup>

<sup>&</sup>lt;sup>492</sup> See page 27 of the Merger Notification Form, paragraphs 4.2.92 to 4.2.94 of the Parties' Response and paragraphs 6.8 to 6.9 of the Frontier Economics Report.

<sup>&</sup>lt;sup>493</sup> The Parties referred to paragraph 7.1 of the Commission's Merger Guidelines, the OFT's decision in case ME/3633/08 - <u>Fishers</u> <u>Services Ltd / The Sunlight Service Group</u> (the activities of the parties' in this transaction overlapped with respect of the supply of linen rental and laundry services in Scotland), paragraph 4.1 of the Authority's Decision in case <u>COM/107/02 Investigation into</u> <u>TicketMaster fees & exclusivity arrangements (2005)</u>, and the Commission's determination in case <u>M/15/026 Baxter Healthcare</u> / Fannin Compounding.

<sup>&</sup>lt;sup>494</sup> The Parties noted that contracts with healthcare customers are typically between one and three years in length.

<sup>&</sup>lt;sup>495</sup> Parties' Response, paragraphs 4.2.95 to 4.2.101 and Frontier Economics Report, paragraphs 1.3.e, 4.3, 7.4 and 7.5

<sup>&</sup>lt;sup>496</sup> The Parties' Response, paragraphs 4.2.32 to 4.2.53



Views of Competitors in the Healthcare Sector

4.255 Two<sup>497</sup> out of the three<sup>498</sup> competitors contacted by the Commission<sup>499</sup> which currently provide outsourced flat linen rental and maintenance services to healthcare customers in the State responded to the Commission's query regarding the competitive effects of the Proposed Transaction in response to the First Competitor Questionnaire.<sup>500</sup> One<sup>501</sup> Competitor indicated to the Commission that the merger will result in a price increase across the whole *"rental market"*, however they noted *"prices are currently slightly lower than what they should be anyway"*.<sup>502</sup> The remaining competitor noted that the Proposed Transaction will reduce competition for the merged entity's customers.<sup>503</sup>

### Views of Customers and Self-suppliers in the Healthcare Sector

- 4.256 Eight Customers from the healthcare sector which were contacted directly by the Commission provided views on the parties' closest competitors and the impact of the Proposed Transaction in the healthcare market.<sup>504</sup>
- 4.257 In relation to closeness of competition, Celtic Linen was consistently identified as the closest competitor to the Parties. [...] was of the view that Celtic Linen was the main competitor of Berendsen while [...] noted that in its last tender won by Kings Laundry, Berendsen came second and Celtic Linen third.
- In relation to the impact of the Proposed Transaction on their ability to purchase their 4.258 linen services, Customers from the healthcare sector indicated that the main concern relates to a reduction in competition and in particular for some healthcare customers the fact that the market is going from 3 to 2 service providers.<sup>505</sup> A public healthcare customer noted that Kings Laundry was the third largest player in the healthcare market and post-transaction there will just be two remaining players.<sup>506</sup> In relation to the question of the impact of the Proposed Transaction on the price of outsourced flat linen rental and maintenance services, of six healthcare Customers who responded, three Customers said they thought prices would go up and two mentioned there would be less competition. One Customer said that if the Parties were to try imposing higher prices, they would need to meet and negotiate terms and conditions of provided services and also look for potential alternative providers.<sup>507</sup> The Commission notes that the Customers it contacted directly were those identified by the parties as being of most significance to their business, and it would therefore emphasise the weight of their opinions.

<sup>&</sup>lt;sup>497</sup> Please see [...] response and [...] response to First Competitor Questionnaire.

<sup>&</sup>lt;sup>498</sup> The Commission notes that the remaining competitor which did not provide views on this issue noted that it had no knowledge of the likely impact of the Proposed Transaction on competition. Please see [...] response to First Competitor Questionnaire.

<sup>&</sup>lt;sup>499</sup> The Commission notes that it contacted all linen laundries offering services in the healthcare market, including those providing limited services to a limited customer group (e.g., providing service to a small number of nursing homes).

<sup>&</sup>lt;sup>500</sup> The Commission notes that such competitors provided a response in respect of the competitive effects of the Proposed Transaction in relation to the supply of flat linen rental and maintenance services in the State and did not segment their answer in respect of healthcare and hospitality.

<sup>&</sup>lt;sup>501</sup> The Commission notes that this competitor intends to cease its linen rental operations shortly.

<sup>&</sup>lt;sup>502</sup> See [...] response to First Competitor Questionnaire.

<sup>&</sup>lt;sup>503</sup> See [...] response to First Competitor Questionnaire.

<sup>&</sup>lt;sup>504</sup> See paragraphs 10.29 to 10.30 below.

 $<sup>^{\</sup>rm 505}$  See paragraphs 10.31 to 10.36 below.

 $<sup>^{\</sup>rm 506}$  See phone call record with [...] dated 21 January 2019.

<sup>&</sup>lt;sup>507</sup> See phone call record with [...] dated 4 October 2018.



- 4.259 A number of self-suppliers from the healthcare sector that expressed views on the impact of the Proposed Transaction were of the view that the Transaction will have no impact in the healthcare market.<sup>508</sup>
- 4.260 The Commission received no Third Party submissions. The lack of Third Party submissions does not in itself suggest that third parties have no concerns about the Proposed Transaction. In this regard, the Commission notes that only 22% of the Amárach Survey respondents were aware of the Proposed Transaction.<sup>509</sup>
- 4.261 The Amárach Survey found that over two thirds (67%) of customers from the healthcare sector indicated they thought that the Proposed Transaction would have a neutral impact, with 8% saying a positive impact and 7% saying it would have a negative impact.<sup>510</sup>

## Views of the Commission

- 4.262 The assessment of the competitive effects of the Proposed Transaction requires the identification of any relevant theories of harm (i.e., how the Proposed Transaction could result in an SLC) and an analysis of those theories of harm through an examination of the available evidence.
- 4.263 In this case, the Commission focused on two theories of harm as part of its assessment of the likely competitive impact of the Proposed Transaction in the healthcare market in the State: (i) unilateral effects; and (ii) coordinated effects (discussed in paragraphs 4.379 to 4.386 below).

## Unilateral Effects

- 4.264 In paragraphs 4.27 to 4.52, the Commission set out its overall approach to the assessment of competition, and this provides a context for further analysis. In considering the likelihood of unilateral effects occurring following the implementation of the Proposed Transaction, the Commission assessed a number of factors that are discussed in turn below:
  - a) competitive pressure exerted by Kings Laundry on Berendsen and Celtic Linen before the Proposed Transaction (discussed in paragraphs 4.265 to 4.274 below);
  - b) closeness of competition between the Parties (discussed in paragraphs 4.275 to 4.304 below)
  - c) constraints from remaining competitors of the parties (discussed in paragraphs 4.305 to 4.313 below);

<sup>&</sup>lt;sup>508</sup> See paragraphs 10.43 to 10.44 below.

<sup>&</sup>lt;sup>509</sup> Slide 56 of the Amárach Survey.

<sup>&</sup>lt;sup>510</sup> See slide 58 of the Amárach Survey. Please note that just 22% of healthcare respondents said that they were aware of the Proposed Transaction (slide 56 of the Amárach Survey)



- d) assessment of the likely effects of the Proposed Transaction on price and/or quality (discussed in paragraphs 4.314 to 4.331 below);
- e) likelihood of new entry and/or expansion (discussed in paragraphs 4.332 to 4.352 below); and
- f) countervailing buyer power (discussed in paragraphs 4.353 to 4.377 below).

# Pre-merger competitive pressure<sup>511</sup> exerted by Kings Laundry

Kings Laundry's importance as a competitive force in the healthcare market

- 4.265 The Commission's view is that Kings Laundry is an important and effective competitor in the healthcare market. First, since Kings Laundry commenced operating as a linen laundry, it has competed vigorously in order to grow its customer base. While the Parties claimed that Kings Laundry's primary focus has been the hospitality sector, the Commission observes that Kings Laundry has participated in tenders issued by some of the largest private healthcare organisations<sup>512</sup> and has provided services to private healthcare customers for over 10 years.<sup>513</sup> As noted in paragraph 4.218 above, Kings Laundry has been steadily increasing its market share and currently provides services to large private healthcare organisations including [...]. In this regard, information provided by Kings Laundry indicates that it was successful in [...] out of [...] attempts to win healthcare customers ([...]%)<sup>514</sup>. The Commission also points out that Kings Laundry managed to maintain some of its largest healthcare customers such as [...<sup>515</sup>] for over 5 years.<sup>516</sup> Furthermore, as noted in paragraph 4.26 above, evidence gathered during the Commission's investigation indicates that there is a realistic prospect of Kings Laundry making further expansion into the healthcare sector in line with its current growth trajectory. This evidence indicates that Kings Laundry is an important and significant competitive force in the healthcare market.
- 4.266 Kings Laundry's competitive strategy in the healthcare sector appears to be focussed on providing high service quality via a dedicated linen system which, according to Berendsen, "[...]".<sup>517</sup> The Commission observes that Kings Laundry attempts to differentiate itself from its competitors by referring to the dedicated linen system in [...].<sup>518</sup>
- 4.267 The Commission notes the following statements on Kings Laundry's website:

<sup>&</sup>lt;sup>511</sup> See paragraphs 4.7 to 4.13 for an explanation of how the Commission treats evidence pre- and post- merger notification.

<sup>&</sup>lt;sup>512</sup> In this regard, the Commission notes [...] view that at the time of the ITT issued by [...] (i.e., in 2016), Kings Laundry sought actively to expand into the healthcare sector.

<sup>&</sup>lt;sup>513</sup> During the Commission's meeting with Kings Laundry on 1 February 2019, Kings Laundry stated that it started to provide outsourced flat linen rental and maintenance services to [...] in 2007.

<sup>&</sup>lt;sup>514</sup> Kings Laundry's response to question 25 of the First RFI.

<sup>&</sup>lt;sup>515</sup> During the review period, the Commission understands that Kings Laundry lost its contract with [...]. This is further confirmation of the 3 player nature of the healthcare market.

<sup>&</sup>lt;sup>516</sup> Kings Laundry's response to question 22 of the Second RFI.

<sup>&</sup>lt;sup>517</sup> Stated by Berendsen during the Commission's meeting with Berendsen on 30 January 2019, see page 3 of minutes of meeting with Berendsen of 30 January 2019. The Commission notes that the accuracy of these minutes was agreed with the parties.

<sup>&</sup>lt;sup>518</sup> See [...]. This [...] was provided to the Commission in response to question 27 of the First RFI.
# *Figure 9: Kings Laundry's marketing of its outsourced flat linen rental and maintenance services to healthcare customers*<sup>519</sup>

We believe our solutions provide customers with the following benefits:

- An innovative textile product range which concentrates on promoting comfort, protection and cost effectiveness.
- Clear accountability for service delivery.
- Best value for money.
- Innovative services to add value.
- A proven track record based on over 10 year's service to the laundry industry and the healthcare sectors.
- Continually achieving high scores for laundry services.
- Financial stability for long term investment.
- Substantial service support.
- Committed to quality improvement and infection control.
- Unique contingency support.
- ISO 9001:2008 approved.
- All laundry processes meet the HSG (95) 18 Standard.

Figure 10: Kings Laundry's marketing of its dedicated linen system<sup>520</sup> [ *K* REDACTED]

- 4.268 The Commission accepts that Kings Laundry may be [...]. The Commission however notes that Kings Laundry's success in building market share in the healthcare market has been at least partly down to the way in which it has marketed itself. Customers perceive Kings Laundry as innovative and as providing high quality service.
- 4.269 The Commission's investigation revealed that Kings Laundry is known for providing outsourced flat linen rental and maintenance services to healthcare customers. In this regard, [...] noted that it invited Kings Laundry to participate in its ITT in 2018 because *"Kings Laundry are already working with around [...] private hospitals including the [...]* and [...]".<sup>521</sup> In the Commission's view, customers like [...] are likely to be well informed about alternative suppliers given that they have gone through the process of issuing ITTs/RFQs recently. This can be contrasted with the Amárach Survey's findings on supplier awareness, which were based on customers who had rolled over existing contracts without contacting alternative suppliers.
- 4.270 Kings Laundry has demonstrated a willingness to invest in modern equipment which allows it to achieve greater efficiencies.<sup>522</sup> In this regard, the Commission notes that in

<sup>&</sup>lt;sup>519</sup> See <u>https://www.kingslaundry.com/healthcare-linen-rental/</u>. Accessed on 27 May 2019.

<sup>&</sup>lt;sup>520</sup> Page 4 of the Kings Laundry's [...]. This [...] was provided to the Commission in response to question 27 of the First RFI. Furthermore, on 27 February 2018 Kings Laundry indicated that it was the [...] that Kings Laundry participated in a formal tender process for the [...].

<sup>&</sup>lt;sup>521</sup> See phone call record with [...] dated 7 May 2019.

<sup>&</sup>lt;sup>522</sup> During the Commission's meeting with Kings Laundry on 1 February 2019, Kings Laundry noted that its Cork facility is [...].



2016 Kings Laundry opened its new facility in Cork having invested  $\in$ [...] in total.<sup>523</sup> Berendsen, in its internal documents that describe [...], notes the following in relation to [...]:

"[....]"<sup>524</sup>

- 4.271 Whilst this facility predominantly services Kings Laundry's [...]<sup>525</sup>, [...<sup>526</sup>].
- 4.272 Finally, the importance of Kings Laundry as a competitive force in the healthcare market is further supported by evidence from Berendsen's internal documents. In particular, the Commission refers to the following statements and figures:
  - a) "[...]"<sup>527</sup>; and
  - b) *"[...]"*.<sup>528</sup>

Figure 11: Berendsen's internal document on [...]<sup>529</sup> [*X* REDACTED]

*Figure 12: Berendsen's internal document on* [...]<sup>530</sup> [ *KREDACTED*]

4.273 The Commission notes in its counterfactual that, absent the merger, the healthcare market is a market with 3 significant players. The Commission also observes that Berendsen itself, in pre-merger notification internal documents considers [...] as shown in Figure 13 and Figure 14.

*Figure 13: Berendsen's internal document on* [...] (I)<sup>531</sup> [*X REDACTED*]

*Figure 14: Berendsen's internal document on [...] (II)*<sup>532</sup> [*X REDACTED]* 

<sup>&</sup>lt;sup>523</sup> In response to question 13 of the First RFI, Kings Laundry noted that initial investment was approximately €[...] with additional €[...] spent in [...].

<sup>&</sup>lt;sup>524</sup> Internal email between Berendsen and Elis titled [...], dated 2 February 2018 provided by Berendsen in response to question 1 of the First RFI.

<sup>&</sup>lt;sup>525</sup> Kings Laundry noted that its Cork facility services [...]. See Kings Laundry's response to the Commission query dated 15 February 2019.

<sup>&</sup>lt;sup>526</sup> While Kings Laundry specifically referred to private healthcare customers, the Commission is of the view that Kings Laundry would be equally well placed to cater for public healthcare customers.

<sup>&</sup>lt;sup>527</sup> Internal email between Berendsen and Elis titled [...], dated 2 February 2018 provided by Berendsen in response to question 1 of the First RFI.

<sup>&</sup>lt;sup>528</sup> Page 1 of the Document titled [...] provided by Berendsen in response to question 9 of the First RFI.

<sup>&</sup>lt;sup>529</sup> Slide 8 of the Document titled [...], dated 23 March 2018 submitted by Berendsen in Appendix G of the Merger Notification Form.

<sup>&</sup>lt;sup>530</sup> Slide 24 of the Document titled [...], dated 23 March 2018 submitted by Berendsen in Appendix G of the Merger Notification Form. Redaction was made by the parties.

<sup>&</sup>lt;sup>531</sup> Slide 15 of the document titled [...] provided by Berendsen in response to question 9 of the First RFI. It should be noted that in slide 13 of this document, [...].

<sup>&</sup>lt;sup>532</sup> Slide 26 titled [...] provided by Berendsen in response to question 9 of the First RFI.



4.274 The Proposed Transaction would see Kings Laundry, currently an expanding innovative competitor in the healthcare market, being incorporated into Berendsen, the largest supplier. In the Commission's view, following the implementation of the Proposed Transaction, healthcare customers that can currently invite three bidders when tendering for services would no longer benefit from the competitive pressure exerted by Kings Laundry's presence as an independent and strong competitor in the market.

#### Closeness of competition between the Parties

- 4.275 Paragraph 4.19 of the Commission's Merger Guidelines states "[A]II things being equal, a merger between close competitors (i.e., competitors engaged in intense competition)<sup>533</sup> will remove a strong competitive constraint and hence be more likely to raise competition concerns than a merger between distant competitors."
- 4.276 The healthcare market is characterised by tenders in which bidders face significant uncertainty over the offers made by rival bidders and the risk of losing out to more attractive offers from competitors. In these circumstances, facing more rival bidders typically increases, for any particular bid, the probability that the buyer will receive a preferable rival offer, which means that bidders will tend to bid more aggressively (i.e. closer to cost). Thus, the question is not, as the parties argued, only whether the Parties were the first and second choice in the observed tenders in the past (although this is also an important piece of evidence in measuring closeness of competition). Rather, the relevant question is whether the Parties are perceived to be strong and important competitors (in terms of their competitive offering) to have a chance of winning, (i.e., they are considered to be good alternatives by customers) and thus, whether bid participation (including perceived participation) of Kings Laundry and/or Berendsen affects bidding incentives of other competitors.<sup>534</sup> As well as considering past performance, the Commission's analysis of the effects of the merger must also be forward-looking, and must take into account the effect of reducing the number of significant competitors in the healthcare market from 3 to 2.
- 4.277 In assessing the closeness of competition between Berendsen and Kings Laundry the Commission has examined the following evidence:
  - a) the views of the Parties (summarised in paragraph 4.235 to 4.237 above);
  - b) the Parties' internal documents (discussed in paragraph 4.235 above);
  - c) the Parties' submissions (discussed in paragraphs 4.236 to 4.237 above);
  - d) the views of third parties (summarised in paragraphs 4.255 to 4.261 above);
  - e) analysis of tendering data (discussed in paragraphs 4.278 to 4.287 below);
  - f) analysis of the Parties' customer loss data (discussed in paragraphs 4.288 to 4.297 below); and

<sup>&</sup>lt;sup>533</sup> Footnote 9 of the Commission's Merger Guidelines states that "In a differentiated product market the most intense competition will occur between the most substitutable products or services."

<sup>&</sup>lt;sup>534</sup> As confirmed by DotEcon, see slide 9 of the DotEcon Report.



g) the Amárach Survey (discussed in paragraphs 4.298 to 4.301 below).

#### Tender participation data

- 4.278 The Commission notes that the Parties and Competitors provided a relatively limited number of instances of ITTs issued by healthcare customers in the period from 1 January 2014 to 30 June 2018. Furthermore, tender participation should be considered in a context where the HSE has not issued any ITTs since 2014, but has instead rolled over existing contracts.<sup>535</sup> The analysis of past tender participation data is therefore based on a small dataset, and at best only gives an indication of the closeness of competition.<sup>536</sup>
- 4.279 It should also be noted that the Commission's analysis includes only tenders where: a) at least two competitors confirmed their participation; or b) the list of participants was provided by Customers interviewed by the Commission. Thus, analysis excludes healthcare customer procurement attempts where the Commission was unable to unambiguously verify the identities of participating competitors.
- 4.280 At the outset, the Commission notes that some of the current largest Kings Laundry's healthcare customers<sup>537</sup> (in terms of turnover) were gained by Kings Laundry prior to 1 January 2014 (i.e., the date from which tender participation data is available). Thus, Kings Laundry's participation and success in tenders does not fully reflect its position in the healthcare market. In this regard, information provided by Kings Laundry indicates that it was successful in [...] out of [...] healthcare tenders ([...]%).<sup>538</sup> As far as Kings Laundry is aware, they competed against Berendsen in [...] instances, winning [...]. Kings Laundry believes it competed against Celtic Linen in [...] instances, winning [...].
- 4.281 Table 14 below shows that Berendsen most frequently faced Celtic Linen in tenders for the supply of outsourced flat linen rental and maintenance services issued by healthcare customers. In the period from 1 January 2014 to 30 June 2018 Berendsen participated in [...] tenders, with Celtic Linen participating in [...] of those tenders, while Kings Laundry participated in [...] of those tenders (i.e., [...]%).

Linen laundry	Number of bids	Frequency of participation
Celtic Linen	[]	[]%
Kings Laundry	[]	[]%
Other	[]	[]%
Total	[]	-

Table 14: Bidders facing Berendsen – the healthcare market 2014 – 30 June 2018 [ $\gg$  REDACTED]

Source: The Commission's analysis using information provided by the parties and parties' competitors.

<sup>&</sup>lt;sup>535</sup> See phone call record with [...] dated 7 May 2019.

<sup>&</sup>lt;sup>536</sup> In this regard, the Commission notes that some healthcare customers tend to extend their existing contract with service providers. In addition, some of the contracts were acquired outside the Commission's reference period 2014 to 30 June 2018. <sup>537</sup> [...].

<sup>&</sup>lt;sup>538</sup> Kings Laundry's response to question 25 of the First RFI.



4.282 Similar analysis in respect of competitors facing Kings Laundry when it bids for the supply of outsourced flat linen rental and maintenance services in the healthcare market shows that Berendsen and Celtic Linen are its most frequent competitors. In the period from 1 January 2014 to 30 June 2018 Kings Laundry participated in [...] tenders, with Berendsen and Celtic Linen [...] of those tenders.

Table 15: Bidders facing Kings Laundry – the healthcare market, 2014 - 30 June 2018 [ 🎉
REDACTED]

Linen laundry	Number of bids	Frequency of participation
Berendsen	[]	[]%
Celtic Linen	[]	[]%
CWS-boco <sup>539</sup>	[]	[]%
Total	[]	-

Source: The Commission's analysis using information provided by the parties and parties' competitors.

- In sum, the analyses of the Parties' bids for tenders issued by healthcare customers 4.283 show that: (i) Kings Laundry has actively participated in tenders issued by private healthcare sector customers, in many of which Kings Laundry was competing on the understanding that it was bidding against Berendsen; (ii) Celtic Linen is Berendsen's most frequent competitor in ITTs; and (iii) those tenders where the parties compete (and will therefore suffer from a loss of competition) have a concentrated market structure, with all tenders having 3 or less bidders. Thus, in tenders where there were 3 bidders pre-transaction, the Proposed Transaction will have a very significant impact on alternatives available to healthcare sector customers. Although the Commission has unambiguously identified only two tenders that had 3 bidders during the examined period, it notes that many healthcare contracts during this period have been rolled-over rather than re-tendered. Having regard to the relevant counterfactual set out in paragraph 4.26 above that Kings Laundry would expand its activities in the healthcare market, the Commission would expect that, absent the Proposed Transaction, the number of tenders with 3 bidders is likely to increase.<sup>540</sup>
- 4.284 However, participation alone is an incomplete indicator of competitive constraint in tenders. The Commission considers that the importance of participation by each linen laundry should also be interpreted in light of its assumptions regarding who it is bidding against, and also in light of its wins. A firm that participates less often but wins frequently is perceived to be a stronger competitor who might be expected to participate in future tenders. The analysis of win rates therefore provides evidence of both the importance of the competitive constraints that the Parties exert on one another, and that is exercised by remaining competitors.
- 4.285 Table 16 and Table 17 below report win rates in tenders for outsourced flat linen rental and maintenance services in the healthcare sector that Berendsen and Kings Laundry participated in during the period from 1 January 2014 to 30 June 2018.

<sup>&</sup>lt;sup>539</sup> This bid was made before CWS-boco stopped providing outsourced flat linen rental and maintenance services.

<sup>&</sup>lt;sup>540</sup> For example, as noted in footnote 207, Kings Laundry participated in [...].



4.286 Table 16 illustrates that Berendsen won [...] out of [...] tenders in which it participated ([...]%). Celtic Linen systematically competed against Berendsen and won [...]% of tenders in which Berendsen also submitted bids while Kings Laundry won [...] of those tenders ([...]%). Table 17 illustrates that Kings Laundry won [...] out of [...] tenders in which it participated ([...]%) with Celtic Linen winning [...] tender ([...]%).

Table 16: Winners in tenders for flat linen rental and maintenance services in the healthcare market that Berendsen participated in, 2014 - 30 June 2018 [ $\gg$  REDACTED]

Linen laundry	Number of wins	Win rate
Berendsen	[]	[]%
Celtic Linen	[]	[]%
Kings Laundry	[]	[]%
Number of tenders	[]	-

Source: The Commission's analysis using information provided by the parties and parties' competitors.

Table 17: Winners in tenders for flat linen rental and maintenance services in the healthcare market that Kings Laundry participated in, 2014 - 30 June 2018 [ $\gg$  REDACTED]

Linen laundry	Number of wins	Win rate
Berendsen	[]	[]%
Celtic Linen	[]	[]%
Kings Laundry	[]	[]%
CWS-boco <sup>541</sup>	[]	[]%
Number of tenders	[]	-

Source: The Commission's analysis using information provided by the parties and parties' competitors.

4.287 In summary, the Parties are part of a small number of three linen laundries that also includes Celtic Linen competing against each other for business from healthcare customers. The competitive interactions in these tenders show that Berendsen won [...] of the tenders that it participated in, whilst Kings Laundry exercised significant competitive pressure on both Celtic and Berendsen by successfully competing and winning [...] out of [...] tenders in which all 3 linen laundries submitted bids. In this regard, as highlighted in paragraph 4.280 above, information provided by Kings Laundry indicates that it was successful in [...] out of [...] attempts to win healthcare customers ([...]%).<sup>542</sup> Finally, the Commission would note that jointly Berendsen and Kings Laundry accounted for a [...] share of won tenders ([...] out of [...] or [...]%) which further indicates the strength of the market power of the merged entity, and the increasing asymmetry in the market if the transaction goes ahead.

#### Analysis of the Parties' customer loss data

4.288 The Commission notes that the Parties provided a relatively limited number of instances where their healthcare customers switched to another supplier in the period from 1

<sup>&</sup>lt;sup>541</sup> This bid was made before CWS-boco stopped providing flat linen rental and maintenance services.

<sup>&</sup>lt;sup>542</sup> Kings Laundry's response to question 25 of the First RFI. For 10 of these attempts, the Commission could not verify whether there were any other participating competitors. Thus, they were excluded from the Commission's tendering analysis.



January 2014 to 30 June 2018. The analysis of customer loss data is based on a small dataset.

- 4.289 The extent to which the Parties have won or lost customers between each other provides a useful (if limited) indicator of the closeness of competition between them.<sup>543</sup> The Commission explored switching patterns over time using information provided by the Parties. Both Parties provided a list of customers that switched from them in the period from 1 January 2014 to 30 June 2018. The Commission estimated switching ratios or diversion ratios between the Parties in terms of: (i) the number of switching customers ("diversion ratio").<sup>544</sup> and (ii) lost contract value of switching customers ("weighted diversion ratio").<sup>545</sup>
- 4.290 In performing this analysis, the Commission included only customers for which the parties were able to identify to whom those customers switched (e.g., name of competitor or OPL). The Commission also excluded customers that had become bankrupt (i.e., no longer require flat linen rental and maintenance services) and customers who received service cancellation letters (as opposed to switching on their own initiative).
- 4.291 Based on the analysis of Berendsen's healthcare customer loss data for flat linen rental and maintenance services, the Commission found that Celtic Linen was the closest competitor to Berendsen. In particular, Table 18 indicates that Celtic Linen won [...] of the customers and gained [...] of the turnover lost by Berendsen during the examined period.

		20	14	20	15	20	16	201	L <b>7</b>		-Jun 18	То	tal
	Linen Laundr Y	Nr ·	%	Nr	%	Nr	%	Nr.	%	Nr.	%	Nr.	%
	Kings Laundr Y	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%
In terms of	Celtic Linen	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%
custome r	Other	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%
numbers	OPL	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%
	Total	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%

Table 18: Berendsen's diversion ratios in the healthcare market, 2014 - 30 June 2018 [ $\gg$  REDACTED]

<sup>543</sup> In this regard see paragraph 4.20 of the Commission's Merger Guidelines.

<sup>544</sup> Diversion ratios were calculated by considering, on the numerator, the total number of a party's (e.g. Berendsen) lost contracts to a certain competitor (e.g., Kings Laundry) within a certain year (e.g. 2017) and on the denominator, the total number of a party's contracts lost within the same year.

<sup>545</sup> Weighted diversion ratios were calculated by considering, on the numerator, the total number of a party's (e.g. Berendsen) lost contracts to a certain competitor (e.g. Kings Laundry) times their total value within a certain year (e.g. 2017) and on the denominator, the total number of a party's contracts lost times their total value for the same year.



		2014		4 2015		2016		2017		Jan-Jun 2018		Total	
		€	%	€	%	€	%	€	%	€	%	€	%
In terms of contract	Kings Laundr Y	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%
value (in €000s)	Celtic Linen	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%
	Other	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%
	OPL	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%
	Total	[ ]	[ ]%	[ ]	[ ]%	[ ]	[ ]%	[]	[] %	[]	[] %	[]	[]%

Source: The Commission's analysis of Berendsen's data on lost healthcare customers

4.292 The same analysis was performed using information provided by Kings Laundry. The Commission found that Berendsen exerted the strongest competitive constraint on Kings Laundry in the healthcare sector. In particular, Table 19 indicates that Berendsen won [...]% of the customers lost by Kings Laundry during the examined period and gained [...]% of turnover lost by Kings Laundry in the same period. In comparison, Celtic Linen won [...]% of customers and [...]% of turnover lost by Kings Laundry.

Table 19: Kings Laundry's diversion ratios in the healthcare market, 2014 - 30 June 2018 [🎉
REDACTED]

		2014		2015		2016		2017		Jan-Jun 2018		Total	
	Linen Laundry	Nr.	%	Nr.	%	Nr.	%	Nr.	%	Nr.	%	Nr.	%
	Berends en	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %
In terms	Celtic Linen	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %
of customer numbers	Other	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %
	OPL	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %
	Total	[]	[] %	[]	[] %	[]	[] %	[]	[	[]	[ 	[]	[] %



									] %		] %		
		€	%	€	%	€	%	€	%	€	%	€	%
	Berends en	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %
In terms	Celtic Linen	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %
of contract value (in €000s)	Other	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %
€000S)	OPL	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %
	Total	[]	[] %	[]	[] %	[]	[] %	[]	[  ] %	[]	[  ] %	[]	[] %

Source: The Commission's analysis of Kings Laundry's data on lost healthcare customers

- 4.293 The Commission acknowledges the value of examining the ranking performance in bids, and in calculating diversion between competitors. In the Commission's view, looking at the performance of the merging parties in past tenders provides an indication of their relative strengths, and so the extent to which their potential participation in a tender affects their bidding incentives and the bidding incentives of other bidders.
- 4.294 However, in the Commission's view, the analysis should not be limited to a question of who came first or second in previous tenders. If Berendsen and Kings Laundry were not to compete against one another in future tenders, then the competition effect of the merger would indeed be minimal. However, if they do compete, it cannot be determined beforehand who will be first, second or third, and this is why the number of participants matters, and why there is a participation effect. As the Commission has set out in its counterfactual, the most realistic scenario absent the merger is that Kings Laundry will continue to compete and will indeed expand its activities in the healthcare market, and so it is more rather than less likely that the parties will compete in the future. It is the Commission's view that its analysis of the likely effects of the merger should not be limited to an analysis of past behaviour.
- 4.295 The loss of a competitor would make a price increase significantly more risky, and thereby contribute to an upward price pressure. This risk would be mitigated only if that competitor is perceived to be too weak (in terms of its competitive offering) to have a chance of winning, or if there are so many strong remaining bidders that there is little incremental effect from an additional competitor. Neither of these conditions holds in the healthcare market.

Coimisiún um Iomaíocht agus Cosaint Tomhaltóirí

Competition and Consumer Protection Commission

- 4.296 To sum up, the switching data confirms that the healthcare market can be characterised as a market with 3 significant players. [...] customers and revenue lost by Berendsen was lost to Celtic Linen. On the other hand, the majority of healthcare customers and revenue lost by Kings Laundry was lost to [...]. The Parties' Response emphasised that Kings Laundry did not lose a customer to [...] since 2016.<sup>546</sup> However, Kings Laundry [...] since 2016 (i.e. from 1 January 2017 to 30 June 2018). Thus, this time period is not informative about Kings Laundry's diversion ratios in the healthcare market and the Commission considers that it is more useful to examine the whole period over which information on customer losses was collected.
- 4.297 Kings Laundry's presence in the healthcare market exerts a significant competitive constraint on both Berendsen and Celtic Linen. In particular, the Commission notes that Kings Laundry acquired the contract for [...] from [...] in 2016 and has attempted to win the contract with [...] (one of the [...] former healthcare customers of Berendsen) in 2018.<sup>547</sup> Other linen laundries with limited activities in the healthcare market have posed no competitive constraint on the Parties to date. As discussed in more detail in paragraphs 4.333 to 4.352 below, the Commission observed no evidence to suggest that this will change post transaction.

#### The Amárach Survey

- 4.298 The Amárach Survey examined the closeness of competition between linen laundries in the healthcare market.
- 4.299 When asked about the providers who they considered in tenders and negotiations, healthcare sector customers indicated that Celtic Linen, Berendsen and Kings Laundry were most likely to submit quotes or tenders. In particular the Amárach Survey results show that:
  - a) two healthcare sector respondents who said that Berendsen was their previous provider and who switched, indicated that they switched to Celtic Linen, while one healthcare sector respondent who switched from Kings Laundry switched to Berendsen;<sup>548</sup>
  - b) healthcare sector respondents were asked what they would do if their supplier of flat linen rental and maintenance services raised its prices for the service by between 5% and 10% and everything else remained the same (price diversion). Of those healthcare sector respondents, 42% said they would use another linen laundry instead, and 27% would stay with their current provider. None said that they would self-supply.<sup>549</sup> However, of those who said they would switch to another provider, when asked who they would switch to 73% said they didn't know.<sup>550</sup>

<sup>&</sup>lt;sup>546</sup> Paragraph 4.265.1 of the Parties Response and paragraph 4.19 b. of the Frontier Economics Report.

<sup>&</sup>lt;sup>547</sup> Furthermore, on 27 February 2018 Kings Laundry indicated that it was the [...]. It should also be noted that in this tender Kings Laundry was ranked [...] and Berendsen (the incumbent provider at the time) third. See phone call record with [...] dated 7 May 2019.

<sup>&</sup>lt;sup>548</sup> See slide 30 of the Amárach Survey.

<sup>549</sup> See slide 52 of the Amárach Survey

<sup>&</sup>lt;sup>550</sup> See slide 54 of the Amárach Survey.



- 4.300 Moreover, most healthcare sector respondents, (73%) said that they have not become aware of any new linen laundries since they last procured flat linen rental and maintenance services.<sup>551</sup>
- 4.301 To conclude, although sample sizes are too small to allow the Commission to put significant weight on individual figures, the following broad findings can be drawn from the results of the Amárach Survey:
  - a) Berendsen holds an influential position in the healthcare market;
  - b) Berendsen and Celtic Linen are the two main providers of flat linen rental and maintenance services to healthcare customers with the only other credible alternative being Kings Laundry; and
  - c) there is little evidence to indicate that other linen laundries are widely known in the healthcare sector.

## Conclusion on closeness of competition

- 4.302 The evidence presented in paragraphs 4.275 to 4.301 above suggests that the premerger healthcare market is a market with 3 significant players. Celtic Linen is the main competitive constraint on Berendsen's probability of winning a tender, and Kings Laundry is a constraint on Berendsen's probability of winning a tender in tenders where both Berendsen and Kings Laundry participated. Thus, the Proposed Transaction will result in the removal of an important and significant competitor to both Berendsen and Celtic Linen and reduce the number of important and significant competitors in the healthcare market from 3 to 2.
- 4.303 In this regard, it is important to consider the degree of the healthcare market concentration in addition to the diversion ratio between the Parties when assessing the likely merger impact. The Proposed Transaction would reduce the number of credible options to healthcare customers, in a concentrated market with little choice of alternative suppliers (the choice would be the merged entity or Celtic Linen).
- 4.304 Moreover, having regard to the relevant counterfactual as set out in paragraph 4.26 above, the Proposed Transaction will result in the elimination of the expanding competitor in the healthcare market, which was perceived by Berendsen as [...].<sup>552</sup>

# Constraints from competitors following the implementation of the Proposed Transaction

Constraints exerted by Celtic Linen

- 4.305 Celtic Linen is the only other linen laundry which actively competes in a significant way with Berendsen and Kings Laundry in the healthcare market.
- 4.306 Celtic Linen is a private limited company registered in the State with a registered address at Drinagh, Co. Wexford. Celtic Linen's primary plant is located in Drinagh, Wexford where it has two laundry facilities; one of which is currently accredited to serve

<sup>&</sup>lt;sup>551</sup> See slide 47 of the Amárach Survey.

<sup>&</sup>lt;sup>552</sup> See paragraph 4.272 above.

Coimisiún um Iomaíocht agus Cosaint Tomhaltóirí

healthcare customers.<sup>553</sup> Celtic Linen also has another laundry facility located in Naas.<sup>554</sup> Celtic Linen has three depots (for linen storage) located in Ballinasloe, Cork and Dublin.

- 4.307 The Commission notes the arguments raised by the Parties in the Report<sup>555</sup> on Celtic Linen. Based on information obtained from Celtic Linen, it is evident that Celtic Linen has benefited from [...], the acquisition of Millbrook Linen, [...] and [...] since exiting examinership.<sup>556</sup> However, the Commission notes that while Celtic Linen's financial position has been improving since its recent exit from examinership, it still made a posttax loss of approximately €1.6 million in 2017.<sup>557</sup> While the Commission considers that Celtic Linen will continue to compete against the merged entity in the healthcare market post-transaction, the Commission is concerned that Celtic Linen would have an ability and incentive to raise its prices in response to a price rise by the merged entity, as discussed in more detail below in paragraphs 4.320 to 4.321 below.
- 4.308 Specifically, the Commission notes that the removal of Kings Laundry as a competitive constraint on both Berendsen and Celtic Linen will significantly reduce competition in the healthcare market. Ultimately, the Proposed Transaction will result in a reduction of the pool of prospective suppliers of outsourced flat linen rental and maintenance services to healthcare customers from three to two, which must be expected to result in less competitive tenders with reduced pressure on bidders. While the Commission agrees with the Parties' view that Celtic Linen will continue to compete with the merged entity, Celtic Linen will not replace the loss of competitive pressure exerted by Kings Laundry pre-merger. In this regard, as discussed in paragraphs 4.320 to 4.321 below, the Commission is of the view that Celtic Linen would have an incentive to follow any price increases implemented by the merged entity, and would have an incentive to compete less aggressively, thereby limiting its competitive constraint on the merged entity post-transaction.

#### Constraints exerted by other linen laundries

4.309 The Commission notes that [...] and [...] provide very limited laundry services to certain healthcare customers.<sup>558</sup> For example, [...] currently provides such services to three nursing homes in the State and generates minimal turnover from such customers.<sup>559</sup> In addition, while [...] was asked to provide back-up services to [...],<sup>560</sup> it has not actively supplied any outsourced flat linen rental and maintenance services (including back up services) to [...]<sup>561</sup> For instance, [...] has not participated in any hospital tenders to date.

<sup>&</sup>lt;sup>553</sup> The Commission notes that Celtic Linen [...]. Please see meeting minutes with Celtic Linen [...].

<sup>&</sup>lt;sup>554</sup> Celtic Linen acquired Millbrook Linen Limited ("Millbrook Linen") located in Co. Kildare in February 2018. Celtic Linen informed the Commission that it intends [...]. Please see meeting minutes with Celtic Linen [...] and phone call record with Celtic Linen [...]. <sup>555</sup> Son paragraph 4 239 above for a summary of the Papert on Celtic Linen.

<sup>&</sup>lt;sup>555</sup> See paragraph 4.239 above for a summary of the Report on Celtic Linen.

<sup>&</sup>lt;sup>556</sup> See meeting minutes with Celtic Linen [...]; phone call record with Celtic Linen [...]; and, Celtic Linen's response to the First Competitor Questionnaire.

 <sup>&</sup>lt;sup>557</sup> Celtic Linen Directors' Report and Financial Statements for year to 31 December 2017, page 7. Please see meeting minutes with Celtic Linen [...]; phone call record with Celtic Linen dated [...]; and Celtic Linen's response to the First Competitor Questionnaire.
<sup>558</sup> The Commission notes that it contacted 11 businesses currently active in the linen/laundry sector in the State. Please see [...]

response to question 1 [...] to the First Competitor Questionnaire and please see [...] response [...] to the First Competitor

Questionnaire. For completeness, the Commission also refers to [...] response to the CCPC's First Competitor Questionnaire and its two nursing homes customers. However, [...] informed the Commission that it intends to fully exit "rental" within the next 2 years. Please see [...] response to the CCPC's First Competitor Questionnaire [...].

<sup>&</sup>lt;sup>559</sup> See [...] response [...] to question 5 of the First Competitor Questionnaire. [...] told the Commission that it expects to serve a few additional nursing homes as one of its existing nursing home customers is adding additional sites.

<sup>&</sup>lt;sup>560</sup> In this regard, [...] noted that they have not provided back-up services to [...] to date. Please see meeting minutes with [...]. <sup>561</sup> See phone call record with [...].



While [...] has stated an intention to supply hospital customers, it has not sought appropriate accreditation, nor contacted any potential customers. <sup>562</sup> The Commission also notes that [...] supplies outsourced flat linen rental and maintenance services to two nursing home customers. However, [...] informed the Commission that it intends to fully exit "rental" within the next 2 years. <sup>563</sup> [...] has experience of serving healthcare customers in [...], but has not participated in healthcare tenders in the State in the last 10 years and has not provided evidence to the Commission of an intention to build a presence in the healthcare market.

4.310 For completeness, the Commission notes that [...] only participates in tenders where scrub suit rental and maintenance services [...].<sup>564</sup> As such, the Commission considers that the competitive constraint exerted by [...] and [...] in the healthcare market is minimal and unlikely to credibly constrain the merged entity post-transaction. The Commission notes that no linen laundry other than Berendsen, Kings Laundry and Celtic Linen have tendered for any of the hospital contracts which have been issued in the last 5 years. The Commission notes that the OGP Framework can potentially facilitate linen laundries' entry and expansion in the healthcare market. However, at this point in time, it is uncertain which linen laundries will submit bids and which linen laundries will be admitted to the OGP's Framework. It is also important to note that admission to the OGP's Framework does not guarantee the award of any individual customer contracts.<sup>565</sup>

#### Conclusion on constraint from competitors

- 4.311 It is the Commission's view that the Proposed Transaction will result in a reduction in the number of significant competitors (from 3 to 2) in the healthcare market. In addition, as noted in paragraph 4.308, notwithstanding Celtic Linen's emergence from examinership and the subsequent investment in its facilities, the Commission considers that Celtic Linen will not replace the competitive pressure exerted by Kings Laundry premerger in the healthcare market.
- 4.312 The Commission notes in the Merger Guidelines that: "The number of suppliers may be particularly relevant to the intensity of competition if customers seek to have more than one supplier, e.g., a primary supplier and a secondary supplier, as might be the case if continuity of supply is important."<sup>566</sup> Healthcare customers have emphasised the importance of reliability and continuity of supply, and the HSE in particular has indicated that it needs to have more than one supplier, both to provide the volume of service it requires, and to ensure continuity and contingency of supply.<sup>567</sup>
- 4.313 The Commission also notes that constraint from other actual and potential competitors may change following the implementation of the OGP Framework. However, the impact of the OGP Framework on competition in the healthcare market is uncertain at this

<sup>&</sup>lt;sup>562</sup> See [...] to the First Competitor Questionnaire and the record of phone call with OCL [...].

 $<sup>^{\</sup>rm 563}$  See [...] response to the CCPC's to the First Competitor Questionnaire.

 $<sup>^{\</sup>rm 564}$  See [...] response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>565</sup> Paragraph 2.1.2 of the document titled "Request for Tenders dated 04/06/2019 to establish a Multi Supplier Framework Agreement for the provision of and Management of Laundry and Hire of Linen, Curtains, Workwear and Dust Mat Hire Services.". "The OGP's RFT".

 $<sup>^{\</sup>rm 566}$  The Commission's Merger Guidelines, paragraph 4.40

 $<sup>^{\</sup>rm 567}$  See phone call record with [...] dated 7 May 2019.



point in time and will depend on the number of linen laundries that will be admitted to the OGP Framework, and their eventual success in bidding for contracts.

Assessment of the likely effects of the Proposed Transaction on price and/or quality

- 4.314 The Commission's view is that the merged entity would have an incentive and ability to increase prices and/or to degrade quality.
- 4.315 The Commission's market investigation found that the healthcare market is highly concentrated, with the Parties and Celtic Linen supplying the majority of customers and jointly accounting for over [90-100]% of the market.
- 4.316 The evidence shows that Kings Laundry exerts a significant competitive constraint on both Berendsen and Celtic Linen before the Proposed Transaction, based on customer loss analysis, tendering analysis, internal documents and third-party submissions. For the reasons set out in paragraphs 4.305 to 4.308 above, the Commission expects that Celtic Linen will continue to operate as a competitor to the Parties post-transaction. However, in a concentrated market with limited credible alternatives available to customers (see the analysis of CBP in paragraphs 4.353 to 4.377 below), the elimination of one of the Parties from the market reduces the already limited set of options available to customers and is therefore likely to reduce the competitive constraint faced by both the merged entity and Celtic Linen.
- 4.317 The Commission finds that the remaining linen laundries providing outsourced flat linen rental and maintenance services to customers in the healthcare sector are unlikely to offset the loss of Kings Laundry's competitive constraint, because their activity in the market is extremely limited. The potential for market entry and/or expansion is discussed in paragraphs 4.332 to 4.352 below.

#### Customer perspective of the likely effects

4.318 The Commission recognises that third parties' views on the effects of the Proposed Transaction are inconclusive. Third party views came from the Commission's direct engagement with Customers identified by the parties as significant, and from the Amárach Survey. The Commission notes that, in the Parties' Response, the Parties mainly relied on the results of the Amárach Survey and the HSE's view on the Proposed Transaction's impact on HSE. The Parties did not consider that most of the largest healthcare Customers identified by the Parties and interviewed directly by the Commission expressed a significant level of concern, with 6 out of 8 Customers (75%) stating that the Proposed Transaction will have a negative impact.<sup>568</sup> As noted in paragraph 1.15 above, the Commission's interviews with the significant Customers allowed the Commission to explore Customers' purchasing behaviour and their views on the competitiveness of the market in more detail compared to responses in the Amárach Survey. Furthermore, the Commission notes that the Customers it contacted directly were those identified by the Parties as being of most significance to their business, and it would therefore emphasise the weight of their opinions.<sup>569</sup>

<sup>&</sup>lt;sup>568</sup> This is further discussed in paragraphs 10.31to 10.36 below.

<sup>&</sup>lt;sup>569</sup> Also see footnote 22 above where the Commission highlights the proportion of Parties' total turnover accounted for by the interviewed Customers.



4.319 In the Amárach Survey, looking at views expressed by customers, the Commission notes that 12% of respondents thought that the Proposed Transaction will have negative effects with the largest customers expressing a relatively higher level of concern as set out in Table 20 below.<sup>570</sup>

		Sector		Size				
	Total Public		Private	<10	10-24	25-49	50-100	100+
	<u>26</u>	<u>13</u>	<u>13</u>	<u>1</u>	<u>4</u>	<u>7</u>	<u>6</u>	<u>8</u>
Positive	12%	8%	15%	0	0	29%	0	12%
No Impact	46%	46%	46%	0	25%	43%	66%	50%
Negative	12%	15%	8%	0	0	14%	0	25%
Don't Know	31%	31%	31%	100%	75%	14%	33%	12%

Table 20: Healthcare respondents' views on the impact of the Proposed Transaction

(Source: The Amárach Survey)

#### Likely reaction of Celtic Linen

- 4.320 The Commission has set out its view in paragraphs 4.314 to 4.317 of the likelihood that the SLC in the healthcare market will lead to increased prices by the merged entity. The Commission notes that Celtic Linen also benefits from the reduced number of competitors in the healthcare market post transaction, because it faces the same tradeoff as the merged entity between the probability of winning the tender and the margin earned in case of winning the tender. Although Celtic Linen has informed the Commission that it intends to [...],<sup>571</sup> the Commission considers that this strategy would not be inconsistent with Celtic Linen's changed bidding incentives and less aggressive competition between Celtic Linen and the merged entity. The Commission considers that Celtic Linen would be likely to adjust its bids in light of the reduced number of competitors following the implementation of the Proposed Transaction. This strategy is all the more likely given Celtic Linen's recent exit from examinership and the fact that it made a post-tax loss of approximately €1.6 million in 2017.<sup>572</sup> While the Parties dismiss Celtic Linen's financial position as being irrelevant,<sup>573</sup> the Commission believes it provides a relevant context when considering whether a price increase is likely to be a viable and appealing strategy for Celtic Linen.
- 4.321 The Commission is therefore of the view that the Proposed Transaction is likely to enable the merged entity to raise price or reduce service quality profitably. The Commission is of the view that Celtic Linen would be likely to accommodate any price increase by the merged entity.

*The magnitude of competitive effects* 

4.322 The Commission sought to examine whether the magnitude of likely competitive effects of the Proposed Transaction could be estimated.

<sup>&</sup>lt;sup>570</sup> Responses to this question should be interpreted with caution due to small sample size.

 $<sup>^{\</sup>rm 571}$  Stated by Celtic Linen during the meeting with the Commission on [...].

<sup>&</sup>lt;sup>572</sup> Celtic Linen Directors' Report and Financial Statements for year to 31 December 2017, page 7.

<sup>&</sup>lt;sup>573</sup> Parties' Response, paragraph 4.2.86



- 4.323 The Parties have highlighted the difficulty of using price differential analysis as the basis for considering the magnitude of potential price increases post transaction, and have noted that estimates of price differential may not result in reliable evidence.<sup>574</sup> The Commission agrees that estimates of any price differential cannot be relied on for this purpose.
- 4.324 The Commission notes that the existence of a price gap between the Parties is not in itself sufficient to indicate that the merged entity would attempt to close the gap following the Proposed Transaction. However, having regard to the nature of competition in the healthcare market, a reduction in the number of independent, strong and important competitors would, in the Commission's view, give the merged entity an incentive to increase prices due to the increased probability of winning any given bid.<sup>575</sup> As noted in paragraph 4.47 above, there is a direct trade-off between the probability of winning the tender and the margin earned if the tender is won. The probability of winning a tender decreases as the number of strong bidders increases. Therefore, the elimination of Kings Laundry as a strong and important competitor decreases the competitive constraint on the merged entity following the implementation of the Proposed Transaction.
- 4.325 The Commission considered whether traditional tools that are widely used to model post-merger price changes such as the Indicative Price Rise test ("IPR Test")<sup>576</sup> or the gross upward price pressure index ("GUPPI")<sup>577</sup> could be used to estimate the likely magnitude of competitive effects in the healthcare market. In this regard, it should be noted that these tests are typically used in markets where firms use posted prices for all customers. In contrast, the healthcare market is characterised by bilateral negotiations and/or tendering, and prices are customised.
- 4.326 The Commission notes that the CMA in a recent case in a sector which exhibits a similar nature of competition to that of the healthcare sector (i.e., sector where suppliers submit bids or negotiate bilaterally), found that:

"GUPPI uses measures of diversion between the merging firms and does not take into account the responses of competitors. It may therefore understate the overall price impact in this Merger, in which we found that elimination of one of the Parties from the market reduces the already limited set of options available to customers and is therefore likely to reduce the competitive constraint faced by both the merged entity and PHS."<sup>578</sup>

4.327 Thus, while a GUPPI analysis was carried out and presented in the PMCA report,<sup>579</sup> the Commission has not relied on it due to the limitations acknowledged by the Commission.

<sup>&</sup>lt;sup>574</sup> The Parties' Response, paragraphs 4.28 to 4.37

<sup>&</sup>lt;sup>575</sup> This is confirmed in the DotEcon report, slide 6

<sup>&</sup>lt;sup>576</sup> See Shapiro, C. (1996), 'Mergers with Differentiated Products', Antitrust, Spring.

<sup>&</sup>lt;sup>577</sup> See Salop, S. and Moresi, S. (2009), 'Updating the Merger Guidelines: Comments'.

<sup>&</sup>lt;sup>578</sup> See paragraph 7.266 in the CMA's report on the completed acquisition by Rentokil Initial plc of Cannon Hygiene Limited dated 25 January 2019.

<sup>&</sup>lt;sup>579</sup> Slide 35 in the PMCA Report.



- 4.328 Instead, having regard to the lack of information that would allow a more precise estimation of the magnitude of competitive effects in the healthcare market,<sup>580</sup> the Commission notes that, according to accepted economic theory and literature, the participation effect of a reduction in the number of players from three to two in cases where bidders compete in first-price sealed bid auctions can be large.
- 4.329 For example, based on Claess Bengtsson studies,<sup>581</sup> in a symmetric case with 3 firms, a reduction from 3 to 2 firms can reduce consumer surplus by between roughly 9% and 32% (depending on the degree of differentiation between the bidders). In an asymmetric case with one market leader with 70% of the market pre-merger, and two smaller firms with 15% each, the merger between the two smaller firms leads to a reduction in surplus by between the bidders). While the Commission does not consider that these findings are directly applicable to the healthcare market, they support the view that the removal of one strong competitor from a pool of three prior to the transaction is likely to have a significant impact on bid levels.
- 4.330 The Commission concludes that there is insufficient data to calculate the precise magnitude of a price increase associated with the Proposed Transaction. However, economic literature confirms the principle that a reduction in the number of suppliers from three to two is likely to result in a material increase in prices.

#### Conclusion on the assessment of likely effects

4.331 The Commission is of the view that the Proposed Transaction is likely to enable the merged entity to profitably raise price or reduce service quality. The Commission is of the view that Celtic Linen would be likely to accommodate any price increase by the merged entity.

#### Likelihood of new entry and/or expansion

4.332 In this section, the Commission assesses whether market entry or expansion in the healthcare market might prevent an SLC and, as such, considers whether such entry or expansion would be timely, likely and sufficient.

# Barriers to Entry and Expansion

- 4.333 The Commission's market enquiries revealed a number of barriers to entry and expansion in the healthcare market.
- 4.334 First, the Commission notes that a considerable financial investment (e.g., capital for linen, equipment, facility, etc.) is required for a new entrant to enter the healthcare market.<sup>582</sup> This financial investment includes capital costs for premises, linen,

<sup>&</sup>lt;sup>580</sup> In this regard, the Commission notes that the Parties were not able to provide their expected margins in the bids that they took part in, but rather provided target margins in the healthcare market. See Berendsen's response to question 22 of the First RFI and Kings Laundry's response to question 25 of the First RFI.

<sup>&</sup>lt;sup>581</sup> See Claess Bengtsson, "Simulating the effect of Oracle's Takeover of PeopleSoft", (2005) in "Modelling European Mergers: Theory, Competition Policy and Case)" edited by Peter A.G. van Bergeijk and Erik. Kloosterhuis, Cheltenham: Edward Elgar Publishing, pp. 133-149.

<sup>&</sup>lt;sup>582</sup> In this regard, the Commission refers to the HSE's view that the likely costs of building a laundry compliant with HIQA/Infection control standards would be €6.8m. Please see <u>https://www.lmfm.ie/news/hse-says-laundry-services-under-review/</u>. Accessed on 1 March 2019.



equipment, (e.g., CBW), civil engineering services in respect of drainage, steam, water and ventilation systems etc.

4.335 Second, the Commission notes that certain healthcare customers require suppliers of outsourced flat linen rental and maintenance services to adhere to: (i) hygiene standards that specify the level of quality<sup>583</sup> required for the processing of flat linen in terms of bio-contamination control; and, (ii) certain linen<sup>584</sup> processing guidelines.<sup>585</sup> For instance, the tender document for the supply of outsourced flat linen rental and maintenance services to healthcare customers in respect of St. James' Hospital stated the following in relation quality assurance:

"Tenderers must demonstrate satisfactory internal quality assurance mechanisms in place and accreditation attained. Tenderers must also demonstrate external quality assurance and accreditation – ISO, RABC Standard EN 14065 and HSG 95(18) or equivalent".<sup>586</sup>

4.336 Similarly the OGP's RFT states that:

"Tenderers may initially confirm by way of a self-declaration that it will have an external quality assurance accreditation including EN 14065:2002 Risk Analysis and Bio contamination (RABC) Systems (and have quality assurance accreditation EN 14065: 2016 R.A.B.C by <u>**31st December 2019**</u> or equivalent and that this (these) accreditation(s) shall be made available for inspection upon request. It shall be a condition precedent to the award of Services Contracts for Suppliers to provide their external quality assurance accreditation(s)."<sup>587</sup>

- 4.337 The Commission recognises that, to supply public hospitals, linen laundries not currently supplying public hospitals would need to apply for formal accreditation. According to the NSAI, such accreditation usually takes 3-4 months from application to registration. The Commission notes also the views of a competitor that implementing these accreditations requires considerable input and effort.<sup>588</sup>
- 4.338 The Commission also recognises the importance of previous experience in winning tenders for hospitals. For example, the ITT from St Vincent's University Hospital asks tenderers to demonstrate: "Your company's skills, efficiency, experience and reliability in delivering similar supplies and services in a comparable operational environment. Firm preference is for an acute hospital environment..... How the skills and experience gained from those contracts can be transferred to the Contracting Authority's

<sup>&</sup>lt;sup>583</sup> The Commission understands that there are two commonly cited standards: (i) NHS Executive HSG(95)18. <u>"Hospital laundry arrangements for used and infected linen"</u>, dated 21 April 1995; and (ii) and BS EN 14065:2002, <u>"Textiles – Laundry Processed Textiles – Biocontamination Control System"</u>, dated 6 December 2002 (known in the industry as the risk analysis bio-contamination control ("RABC") standard). In 2016 this standard was superseded by BS EN 14065:2016.

<sup>&</sup>lt;sup>584</sup> For example, see the HSE's guidelines for managing laundry and linen in healthcare facilities which can be accessed at: https://www.hse.ie/eng/about/who/healthwellbeing/infectcont/sth/gl/ipcc-guidelines-section-9.pdf.

<sup>&</sup>lt;sup>585</sup> See paragraph 2.3 above which outlines the specific requirements under such guidelines.

 <sup>&</sup>lt;sup>586</sup> For more information, please see section 12.3 "Selection criteria" of the St. James Hospital invitation to tender document.
<sup>587</sup> Table 4 of the OGP's RFT.

<sup>&</sup>lt;sup>588</sup> See the Commission's minutes of meeting with [...]. In respect of the RABC standard specifically, the NSAI informed the Commission that such accreditation would usually take about 3-4 months, i.e., from "application to registration" to obtain. The NSAI also noted that this depends on non-conformances which extend the timeline. See NSAI's email dated 8 February 2019.



*contract.*<sup>"589</sup> Similarly, when the HSE issued a tender for a Framework Contract (later withdrawn) it emphasised the requirement for bidders to have a "an experienced Senior Management Team capable of directing and managing the delivery of the Services" and stated that experience of providing "healthcare laundry linen services" is essential: "Tenderers must demonstrate that they have successfully delivered contracts of a similar nature to the HSE's requirements during the last three years by providing a list of its previous significant contracts delivered over the last three years.". This was to be supported by references.<sup>590</sup> The OGP also emphasised the importance of experience in its ITT for the OGP's Framework. In this regard, the OGP's RFT states the following: "For Lots 1 & 2, Tenderers must declare by way of eESPD that they have successfully delivered in Appendix 1, must demonstrate in the Health Sector which may include a Department of Defence Military Hospital within the last 3 years."<sup>591</sup>

- 4.339 The Commission notes that the requirement for experience constitutes a barrier to entry for any linen laundry that is not already active in supplying hospitals, and that this would be a considerable barrier for all laundries with the exception of Berendsen, Celtic Linen and Kings Laundry. The Amárach Survey indicates that over 96% of healthcare customers said that experience of providing linen laundry services to organisations in the same sector was either essential or very important.<sup>592</sup>
- 4.340 Some healthcare customers require their suppliers of flat linen rental and maintenance services to healthcare customers in the State to have certain contingency plans in place. For instance, the Commission's market enquiry indicated that certain hospitals (e.g., [...]) require contingency plans to overcome possible plant failure, and as such, linen laundries with two healthcare facilities in separate locations tend to score higher in tenders.<sup>593</sup> In this regard, the OGP's RFT states that: *"Tenderers must have contingency provisions to be imitated [sic] should there be an interruption in the laundry processing process or in the distribution process that may result in potential interruption of service."<sup>594</sup> In addition, the Amárach Survey indicates that overall, over 4 in 5 customers (84%) when asked, said that good contingency plans were either essential or very important.<sup>595</sup>*
- 4.341 Finally, the Commission refers to Berendsen's internal documents in which the following was stated in respect of the healthcare market:

<sup>589</sup> Appendix 8, St Vincent's University Hospital ITT, 3 May 2018

<sup>&</sup>lt;sup>590</sup> https://ted.europa.eu/udl?uri=TED:NOTICE:429143-2018:TEXT:EN:HTML&src=0

<sup>&</sup>lt;sup>591</sup> Paragraph 3.2.B of the OGP's RFT. The OGP's RFT further specifies that each of these contracts must: (i) "Be similar in nature to the Services required in Appendix 1 and in the Health Sector which may include a Department of Defence Military Hospital"; (ii) "Have a minimum value of €144,000 per year of contracts"; (iii) "One of the contracts provided must be in place or have been in place for more than 2 years"; (iv) "Demonstrate drop and collect Services of Laundry and / or rental of Linen, Curtains, Workwear and Dust Mats of a similar size and nature to the requirements in Appendix 1, but in the Health Sector which may include a Department of Defence Military Hospital"; (v) "Tenderers must provide one example demonstrating experience of the laundry and / or rental, drop and collect Services of Linen, Curtains, Workwear and Dust Mats across multiple sites within one Province in Ireland (Munster, Leinster, Ulster, and Connacht)" and (vi) "Demonstrate the laundry and / or rental, drop and collect Services of Linen, Curtains, Workwear and Dust Mats to a Customer with locations in more than one county within Ireland".

<sup>&</sup>lt;sup>592</sup> 25 out of 26 healthcare customers that were asked this question. See slide 36 of the Amárach Survey where combined responses across healthcare and hospitality sectors are presented.

<sup>&</sup>lt;sup>593</sup> Please see record of a phone call with [...].

<sup>&</sup>lt;sup>594</sup> Page 53 of the OGP's RFT. The OGP indicated that it will consider access to alternative processing facilities within the provider's organisation amongst the list of other factors when evaluating bidders methodologies in respect of disaster recovery proposals.

<sup>&</sup>lt;sup>595</sup> Slide 36 of the Amárach Survey. Note this percentage relates to the whole survey not just healthcare customers.



"[...]"<sup>596</sup>

4.342 To conclude, there are a number of barriers to entry and expansion in the healthcare market including financial barriers (e.g., capital required to develop the necessary infrastructure) and non-financial barriers (e.g., reputation/experience which is an important factor for healthcare sector customers when they choose their suppliers of outsourced flat linen rental and maintenance services). The Commission notes that the Parties agreed that these barriers are relevant when assessing the likelihood of entry and expansion to the healthcare market.<sup>597</sup>

#### Evidence of Market Entry or Expansion

- 4.343 The Commission did not observe any evidence that linen laundries based in Northern Ireland are providing outsourced flat linen rental and maintenance services to healthcare customers in the State. In this regard, [...] noted that the last time it participated in the HSE's tender was over ten years ago.<sup>598</sup>
- 4.344 Market shares alone demonstrate that linen laundries such as [...]<sup>599</sup> and [...], while being somewhat successful in the hospitality sector have not yet been able to significantly penetrate the healthcare sector.<sup>600</sup>
- 4.345 [...] informed the Commission that it has no plans to expand its current activities (i.e., outside of "workwear, floorcare or washroom services") in the State.<sup>601</sup>
- 4.346 Fourth, the Commission notes that there has been no new entry in the healthcare market in the last 5 years.
- [...] expansion in the healthcare market
- 4.347 One service provider to nursing homes ([...]), informed the Commission that it intends to supply outsourced flat linen rental and maintenance services to hospitals in the State.<sup>602</sup> The Commission met with [...] to gather additional evidence about its potential expansion in the healthcare market.<sup>603</sup> The evidence presented by [...] can be summarised as follows:
  - a) [...] intends on moving into healthcare in the next 1-2 years;
  - b) [...] has space for additional machinery (e.g., CBW) that can be used to provide flat linen rental and maintenance services to healthcare customers;

<sup>&</sup>lt;sup>596</sup> See page 13 of Berendsen's [...] submitted by Berendsen in response to Question 9 of the First RFI.

<sup>&</sup>lt;sup>597</sup> Frontier Economics Report, Paragraph 6.3.

<sup>&</sup>lt;sup>598</sup> Please see phone call record with [...].

<sup>&</sup>lt;sup>599</sup> Contrary to the Parties' submission, [...] does not currently provide flat linen rental and maintenance services to nursing homes in the State. Please see [...] response to the First Competitor Questionnaire.

<sup>&</sup>lt;sup>600</sup> See the Commission's market share analysis in paragraphs 4.56 to 4.76 (hospitality) above and paragraphs 4.213 to 4.222 (healthcare) above.

<sup>&</sup>lt;sup>601</sup> See [...] to the First Competitor Questionnaire.

<sup>602</sup> See phone call record with [...].

<sup>603</sup> See meeting minutes with [...].



- c) [...] would require minimal investments to alter its facility in order to supply healthcare customers (e.g., complete barrier wall, purchase trolley washer etc.);
- d) [...] expressed interest in providing outsourced flat linen rental and maintenance services to [...] although it was not sure when this hospital's contract will be open for competition;
- e) [...] has not participated in any ITTs issued by hospitals to date;
- f) [...] stated that it is not in the process of obtaining the RABC accreditation and that it does not necessarily intend to get the RABC accreditation prior to competing for healthcare customers. In this regard, [...] opined that obtaining the RABC accreditation would not take a significant amount of time;
- g) [...] was not aware of the HSE's published Framework in Autumn of 2018 nor the OGP's intention to set up a Framework for linen laundry and hire services; and
- h) [...] expects to serve three or four additional nursing homes as one of its existing nursing home customers is adding additional sites.
- 4.348 At the outset, it is important to stress that unlike Berendsen, Celtic Linen and Kings Laundry, [...] has no experience in providing outsourced flat linen rental and maintenance service to healthcare hospital customers to date, and that its experience in the healthcare market is limited to a small number of nursing homes. As noted in paragraph 4.338 above, reputation/experience is an important factor for healthcare customers when they choose their suppliers of outsourced flat linen rental and maintenance services. Thus, [...] lack of experience in providing outsourced flat linen rental and maintenance services to healthcare customers is a significant factor in considering whether [...] expansion would be timely, likely and sufficient to prevent the identified SLC in the healthcare market.
- 4.349 Based on the detailed consideration of evidence provided by [...], the Commission expects [...] to attempt to expand in the healthcare market within the next two years as it has the ability and incentive to do so. However, the timeliness and likelihood of [...] expansion will depend on whether [...] will be deemed as a suitable service provider for healthcare customers, and particularly for hospital customers. For example, this could include whether [...] would apply for, and be admitted to the OGP's Framework. As noted in paragraph 2.12 above, the OGP has issued an RFT and set up the selection criteria on the basis of which the potential suppliers will be admitted to the OGP's Framework. The selection criteria include: (i) Economic and Financial Standing; and (ii) Technical and Professional Ability.<sup>604</sup> Having regard to the criteria set out in the OGP's RFT and in particular the requirement of experience of providing services of a similar nature, i.e. to hospitals, the Commission is of the view that [...] is unlikely to have the required experience to date nor does it meet the OGP's requirement of having at least 3 healthcare customer contracts with a minimum value of €144,000 per year each.
- 4.350 Finally, it should also be noted that the admission to the OGP's Framework in itself would not guarantee that [...] intended expansion would be sufficient to constraint the

<sup>&</sup>lt;sup>604</sup> Section 3.2 of the OGP's RFT.

merged entity. Admission to a Framework does not guarantee success in bidding for contracts. Evidence provided by [...] indicates that [...] plans gradual expansion in the healthcare market by targeting the hospital that is in close proximity to its facility and then using it as a reference to pursue other healthcare customers with a stronger possibility of success. To date, [...] has not made contact with its target customer, and is unaware of the customer's current laundry arrangements and likely future tender plans. The strategy expressed by [...] did not enable the Commission to conclude with sufficient confidence that [...] expansion would be sufficient to prevent the identified SLC in the healthcare market.

## *Entry by* [...] *to the healthcare market*

Coimisiún um

lomaíocht agus

Cosaint Tomhaltóirí

4.351 Another potential competitor, [...], informed the Commission of its intention to supply outsourced flat linen rental and maintenance services to nursing homes.<sup>605</sup> It does not currently serve any healthcare customers. At the same time, [...] also noted that it would not have the capacity to serve hospitals and would need to set up another facility in order to do so. Hence, the evidence provided by [...], did not enable the Commission to conclude with sufficient confidence that the entry by [...] would be timely, likely and sufficient to prevent the identified SLC in the healthcare market.

## Conclusion on market entry or expansion

4.352 The evidence provided to the Commission did not enable the Commission to conclude with sufficient confidence that any supplier will enter/expand to provide flat linen rental and maintenance services to healthcare customers in a timely, likely and sufficient manner to mitigate or prevent the identified SLC or prevent the loss of competitive pressure exerted by Kings Laundry pre-transaction. The OGP's Framework can potentially facilitate market entry/expansion. However, at this point it is uncertain which linen laundries will submit bids and which linen laundries will be admitted to the OGP's Framework. It is also important to note that admission to the OGP's Framework does not guarantee the award of any individual customer contracts. The Commission's view is that the barriers to entry/expansion are of sufficient magnitude to prevent timely market entry, and would not prevent the merged entity from increasing prices/reducing service quality following the implementation of the Proposed Transaction.

#### Countervailing buyer power

- 4.353 The Commission recognises that, in some circumstances, a customer may, because of its position in the market, be able to successfully resist supplier price increases, and may possess sufficient negotiating strength to enable it to constrain the behaviour of a supplier.<sup>606</sup> The assessment of CBP considers the extent to which buyer power postmerger would be present and sufficiently effective to prevent an SLC.
- 4.354 The Commission set out its approach to the assessment of potential CBP in paragraphs 4.196 to 4.197 above. The Commission has assessed the CBP of customers in the healthcare market by examining the following factors:

<sup>605</sup> See phone call record with [...].

<sup>&</sup>lt;sup>606</sup> See paragraph 7.1 of the Commission's Merger Guidelines.



- a) customers' ability to switch away from the merging parties, and/or switch to alternative suppliers (discussed in paragraphs 4.355 to 4.357 below);
- b) the number, nature and viability of alternative options available (discussed in paragraphs 4.358 to 4.366 below);
- c) the size and significance of individual customers (discussed in paragraphs 4.367 to 4.373 below); and
- d) the extent to which the CBP of some customers will benefit a sufficient number of customers to prevent an SLC in the market post-merger (discussed in paragraphs 4.374 to 4.376 below).

#### The ability of customers to switch supplier

- 4.355 The Commission found that customers in the healthcare sector are able to switch between suppliers of outsourced flat linen rental and maintenance services, typically at the end of a procurement exercise. In this regard, the majority of Customers that the Commission interviewed claimed that there were relatively low barriers to switching between alternative providers.<sup>607</sup> The Amárach Survey indicated that 43% of respondents from the healthcare sector thought that switching suppliers was easy.<sup>608</sup> The Commission also observed a number of switching examples with customers switching between Berendsen, Kings Laundry and Celtic Linen. However, despite the apparent ease of switching, almost one in five (19%) indicated that they had not changed or renewed laundry services in the past five years.<sup>609</sup>
- 4.356 The Commission notes that the HSE has not tendered for outsourced flat linen rental and maintenance services since 2014.<sup>610</sup> Since then, it has rolled over existing contracts with its two existing suppliers (Berendsen and Celtic Linen).<sup>611</sup> While this has clearly limited the ability of any other linen laundry to supply the HSE, it has also limited the HSE's ability to switch supplier.
- 4.357 Overall, the Commission is of the view that switching barriers between linen laundries that are active in the healthcare sector are relatively low.

#### The number, nature and viability of alternative options available

4.358 As noted in paragraph 4.224, the Commission found that the healthcare market is highly concentrated. Three suppliers (Berendsen, Celtic Linen and Kings Laundry) serve the largest and the majority of healthcare customers in the State. Hence, following the implementation of the Proposed Transaction, healthcare customers currently outsourcing their flat linen rental and maintenance requirements will essentially have two potential options when facing a price increase/service quality reduction: a) switch to the only remaining provider of outsourced flat linen rental and maintenance services to healthcare customers; or b) self-supply their flat linen requirements.

 $<sup>^{\</sup>rm 607}$  See paragraph 10.22 to 10.23 below.

<sup>&</sup>lt;sup>608</sup> Slide 46 of the Amárach Survey.

<sup>&</sup>lt;sup>609</sup> Slide 26 of the Amárach Survey.

 $<sup>^{\</sup>rm 610}$  See the Commission's note of phone call with [...] dated 7 May 2019.

 $<sup>^{\</sup>rm 611}$  See the Commission's note of phone call with [...] dated 7 May 2019.



- 4.359 The Commission notes that the importance of Kings Laundry's presence in the healthcare market was particularly highlighted at the time when Celtic Linen was in examinership and [...].<sup>612</sup> For example, [...] told the Commission that Kings Laundry was a very important third option when considering a supplier of outsourced flat linen rental and maintenance services in 2016 as it was able to meet all of [...] requirements.<sup>613</sup>
- 4.360 Having regard to the relevant counterfactual as set out in paragraphs 4.26 above, the Proposed Transaction will result in the elimination of a credible third service provider to all customers, including public and private hospitals and nursing homes in the healthcare market.
- 4.361 The Commission has considered the extent to which a switch to OPL would be an effective competitive constraint, and would demonstrate CBP. The Commission has already set out its view on why OPL is not part of the product market, and noted earlier that it was more appropriate to consider self-supply as a potential competition effect. This is consistent with the Commission's approach in its merger determination M/15/026 – Baxter Healthcare / Fannin Compounding<sup>614</sup>. Contrary to the Parties' views set out in the Report on OPLs, in that merger determination the Commission expressed concern "...about whether hospitals with no aseptic compounding facility (both public and private) have the ability to resist a price rise by Baxter for its compounded chemotherapy medicines post-transaction"<sup>615</sup>. Ultimately, the ability to self-supply was not determinative in this case. The Commission determined that the relevant transaction would not have substantially lessened competition in any market for goods or services in the State because "...the competitive structure in the commercial supply of compounded chemotherapy medicines in the State is likely to deteriorate to at least the same extent (and, possibly, to an even greater extent) in the absence of the proposed transaction, there is no basis for prohibiting the proposed transaction..."616
- 4.362 The Commission refers to paragraphs 3.47 to 3.63 above, and reiterates the following:
  - a) all of the Customers in the healthcare sector interviewed by the Commission stated that self-supply would not be a viable option in response to a 5% to 10% price increase of flat linen rental and maintenance services;
  - b) the Amárach Survey indicated that no respondents in the healthcare sector would set up OPL in response to a 5% to 10% price increase by their provider of outsourced flat linen rental and maintenance services;<sup>617</sup> and

<sup>&</sup>lt;sup>612</sup> See the Commission's note of phone call with [...].

<sup>&</sup>lt;sup>613</sup> See the Commission's note of phone call with [...].

<sup>&</sup>lt;sup>614</sup> The determination of this case can be accessed at: <u>https://www.ccpc.ie/business/mergers-acquisitions/merger-notifications/m15026-baxter-healthcare-fannin-compounding/.</u>

<sup>&</sup>lt;sup>615</sup> Paragraph 76 of M/15/026 – Baxter Healthcare / Fannin Compounding merger determination. The Commission also considered that hospitals (public and private) with no aseptic compounding facility were unlikely to have the ability to exercise effective CBP post-transaction to prevent Baxter from raising the price of compounded chemotherapy medicines. The Commission also considered that the HSE was unlikely to have the ability to exert effective CBP by intervening on behalf of public hospitals with no aseptic compounding facility in response to a price rise by Baxter. See paragraph 135 of M/15/026 – Baxter Healthcare / Fannin Compounding merger determination.

<sup>&</sup>lt;sup>616</sup> Paragraph 188 of M/15/026 – Baxter Healthcare / Fannin Compounding merger determination.

<sup>&</sup>lt;sup>617</sup> Slide 51 of the Amárach Survey.



- c) the Parties identified only one instance where a healthcare customer switched to OPL.<sup>618</sup> In contrast, as noted in paragraph 3.54 above, the Commission observed several cases where OPLs were closed in favour of healthcare customers switching to outsourced flat linen rental and maintenance services with such examples including [...] and several nursing homes.
- 4.363 In relation to costs for setting up OPLs as outlined in the Parties' Report on OPLs, the Commission notes that these costs relate to hospitality OPL rather than healthcare OPL. These costs do not take into account additional costs associated with ensuring that the OPL is compliant with the required hygiene standards such as RABC. In this regard, the Commission refers to the HSE's view that the likely costs of building a laundry compliant with HIQA/Infection control standards would be €6.8m.<sup>619</sup> The Commission also refers to operating costs cited by healthcare sector Self-suppliers.<sup>620</sup> These costs are significantly above costs cited in the Report on OPLs.
- 4.364 The Commission notes the Parties' statement that *"The HSE dictates the flat linen volumes that are brought to the market and the flat linen volumes that are kept inhouse."*<sup>621</sup> The Commission agrees that the HSE may choose not to outsource flat linen rental and maintenance services currently provided by its remaining OPLs. However, the Commission is of the view that the HSE is unlikely to be able to cease purchasing outsourced flat linen rental and maintenance services and self-supply all of its flat linen maintenance requirements in its existing OPLs instead. For example, the Commission has not observed any instances where the HSE withdrew, or threatened to withdraw, the ITT for procuring outsourced flat linen rental and maintenance services submitted by bidders.<sup>622</sup> Furthermore, as outlined in paragraph 3.59 above, the Commission notes that linen laundries expect the HSE to close its remaining OPLs rather than increase the proportion of self-supplied flat linen rental and maintenance services. In this regard, [...] noted that *"[...]"*<sup>623</sup>
- 4.365 For these reasons, the Commission is of the view that switching to self-supply is not a viable option for the vast majority of the largest customers in the healthcare market in response to a price increase. While an option of self-supply might be theoretically more viable to certain customers (e.g., smaller nursing homes), the Commission has not observed any evidence indicating that nursing homes would be in a position to effectively exercise such an option to prevent a price increase or service quality deterioration for the whole healthcare market.
- 4.366 Berendsen's internal documents [...] indicate that healthcare customers [...]. For instance, the Commission refers to Berendsen's internal document titled [...]<sup>624</sup> where it is noted that:

<sup>618</sup> See footnote 122.

<sup>&</sup>lt;sup>619</sup> https://www.lmfm.ie/news/hse-says-laundry-services-under-review/. Accessed on 1 March 2019.

<sup>&</sup>lt;sup>620</sup> See paragraph 10.39 below.

<sup>&</sup>lt;sup>621</sup> Paragraph 12 of the Report on OPLs.

<sup>&</sup>lt;sup>622</sup> In this regard, Berendsen stated that it "is not aware of any instances whereby the HSE took greater volumes of Flat Linen Solution Services in-house". See page 5 of document titled "Berendsen response to Competition and Consumer Protection Commission clarification questions of 8 March 2019" provided to the Commission on 13 March 2019.

<sup>&</sup>lt;sup>623</sup> See the Commission's minutes of meeting with [...].

<sup>&</sup>lt;sup>624</sup> Submitted by Berendsen in response to question 9 of the First RFI.



# ["..."] (emphasis added)

## The size and significance of individual customers

- 4.367 The Commission notes that the top 5 customers of the merged entity would collectively account for around [...]% of its total turnover from the provision of outsourced flat linen rental and maintenance services to healthcare customers in the State.<sup>625</sup> The largest customer is [...] which would account for approximately [...]% of the merged entity's turnover<sup>626</sup> with no other customer individually accounting for more than [...]% of the merged entity's turnover. The merged entity is therefore not likely to be overly dependent on individual customers which, in turn, weakens the negotiating power of individual customers.
- 4.368 Given the size of the HSE as a purchaser, the Commission has assessed the HSE's past behaviour in response to a price increase. By way of illustration, the Commission notes that, according to [...].<sup>627</sup>
- 4.369 The Commission observes that the HSE has a degree of discretion as to how its purchases of outsourced flat linen rental and maintenance services are distributed between linen laundries. While this might be expected to give the HSE bargaining power, as noted in paragraph 4.312 above, the HSE requires at least two suppliers to meet its demand for outsourced flat linen rental and maintenance services. This fact diminishes the HSE's bargaining power. In addition, the scale of any would-be CBP is diminished by virtue of the fact that some significant hospitals (e.g., St. James Hospital) carry out their own procurement. The OGP confirmed that there is no mandatory obligation on public bodies to use the OGP's Framework.<sup>628</sup> Thus, the Commission considers that the current practice of individual public healthcare hospitals issuing their own ITTs is likely to continue once the OGP's Framework is established.
- 4.370 The Commission notes that public healthcare customers could potentially exercise buyer power by threatening to outsource procurement to the HSE (or the OGP once the OGP's Framework is established). This strategy could only be a credible threat if: a) the affected customer would expect to get a better outcome from outsourced procurement; and b) the HSE or the OGP can credibly threaten the merged entity to switch a significant portion of its flat linen rental and maintenance requirements from the merged entity in response to a price increase to the affected customer.
- 4.371 In this regard, [...] noted<sup>629</sup> that "[...]" and "[...]". Thus, the Commission considers it unlikely that individual customer's threats to outsource their procurement to the HSE or the OGP can be used as a credible threat to prevent the merged entity from increasing prices/reducing service quality.

<sup>&</sup>lt;sup>625</sup> This is based on the Commission's analysis of information submitted by Berendsen in response to question 11 and 21 of the First RFI and question 26 of the second RFI as well as information submitted by Kings Laundry in response to question 11 and 24 of the First RFI. The analysis is based on the parties' turnover generated in 2017.

<sup>&</sup>lt;sup>626</sup> It should also be noted that the HSE does not issue ITTs on behalf of all public sector organisations. Berendsen listed 21 public healthcare organisations that issue ITTs and award contracts themselves. See Berendsen's response to Question 15 of the RFI. Thus, some of the biggest Berendsen's customers such as [...], are treated as individual customers.

<sup>&</sup>lt;sup>627</sup> Stated by [...] during the meeting with the Commission on [...].

<sup>&</sup>lt;sup>628</sup> See the Commission's note of phone call with [...].

<sup>&</sup>lt;sup>629</sup> See the Commission's note of phone call with [...].



- 4.372 Finally, the Commission understands that [...].<sup>630</sup> In the Commission's view, this indicates that [...] wished to retain [...] as a viable competitor, rather than intended to sponsor market entry. This is not indicative of bargaining power, because [...] is a very different proposition to sponsoring a new entrant.
- 4.373 The Commission is of the view that the HSE's bargaining power is likely to be limited, due to all of the factors discussed above. In addition, the implementation of the Proposed Transaction would reduce the number of options available to the HSE, which would further undermine the HSE's bargaining position in the purchase of flat linen rental and maintenance services in the State. The Commission acknowledges that the OGP's Framework might centralise public healthcare customers' procurement for outsourced flat linen rental and maintenance services. However, this development in itself is unlikely to give the OGP effective CBP that would prevent a price increase/service quality degradation to other healthcare customers (e.g., private healthcare customers).

The extent to which any buyer power of some customers will benefit a sufficient number of customers and prevent an SLC

4.374 Paragraph 7.4 of the Commission's Merger Guidelines states that:

"In a market where some but not all buyers possess significant countervailing buyer power, a merger may still result in increased prices (or other competitive harm) for those customers with little or no countervailing buyer power. For example, it may be that only large customers have the ability to exert countervailing buyer power and protect themselves from competitive harm. Small customers may not have sufficient negotiating strength to successfully exert countervailing buyer power."

- 4.375 The Commission has found that the healthcare market is typically characterised by tendering and/or bilateral negotiations, and that there is little transparency on pricing. As set out in the Commission's Merger Guidelines, in markets characterised by individual negotiations between suppliers and customers, and a lack of pricing transparency, any CBP possessed by one or more customers will typically not protect other customers from any anti-competitive effects that may arise post-merger.<sup>631</sup>
- 4.376 Even if the HSE (or the OGP once the OGP's Framework is established) were able to protect itself from competitive harm having regard to its size (which the Commission does not consider to be the case), following the implementation of the Proposed Transaction, the Commission is of the view that any bargaining power that the HSE/the OGP has/will have is unlikely to prevent a price increase/service quality degradation to other healthcare customers. These customers would be unable to leverage any price discount obtained by the HSE/the OGP primarily because of the lack of pricing transparency. In this regard, it should be noted that the healthcare market is characterised by a very significant number of small customers who would be unlikely to have any bargaining power vis-à-vis the merged entity. For instance, information provided by Berendsen indicates that in 2018, [...] out of [...] ([...]%) of Berendsen's

<sup>630</sup> Stated by [...] during the meeting with Commission on [...]. [...] also noted that currently [...].

<sup>&</sup>lt;sup>631</sup> See paragraph 7.6 of the Commission's Merger Guidelines



healthcare customers spent less than €25,000 on the purchase of flat linen rental and maintenance services with only [...] ([...]%) customers spending €100,000 or more.<sup>632</sup> The Commission acknowledges that the OGP's Framework might centralise public healthcare customers' procurement for outsourced flat linen rental and maintenance services. However, this development in itself is unlikely to give the OGP an effective CBP particularly if the OGP requires at least two suppliers to meet the demand for its outsourced flat linen rental and maintenance services.

## Conclusion on CBP

4.377 The Commission notes that of all healthcare customers, the HSE appears to have some bargaining power, but that its CBP is limited for the reasons discussed above. Furthermore, post transaction, all customers in the healthcare sector will face a reduction in the number of credible suppliers and will be left with a very limited choice of suppliers of outsourced flat linen rental and maintenance services. Thus, the Proposed Transaction would remove a key supplier and reduce healthcare customers' CBP (including that of the HSE or the OGP). The Commission's conclusion is therefore that customers do not have sufficient buyer power to prevent an SLC. The establishment of the OGP's Framework is highly unlikely to change this conclusion.

## Overall conclusion on Unilateral Effects in the healthcare market

- 4.378 For the reasons set out above and considering all of the evidence described above, the Commission considers that the Proposed Transaction is likely to result in significant unilateral effects in the healthcare market. In particular, the Commission is of the view that following the implementation of the Proposed Transaction the merged entity will have an ability and incentive to increase prices/reduce service quality in the healthcare market, which will ultimately result in consumer harm. This is based on the following findings:
  - a) the healthcare market can be characterised as a highly concentrated market with 3 significant players, with each of Berendsen, Kings Laundry and Celtic Linen being an important competitor prior to the Proposed Transaction;
  - b) based on the Commission's assessment of the nature of pre-merger competition, the Commission considers that suppliers take into account the risk of losing to each of their competitors, and not just to their closest competitor (which they would not generally be able to identify *ex ante* in the context of any given tender);
  - c) the Proposed Transaction would eliminate an important competitive force in an already very concentrated market. The credible options available to customers will be reduced to the merged entity and Celtic Linen;
  - d) the impact due to the removal of one of the Parties from the competitive process will be unlikely to be offset by competition from Celtic Linen in a timely, likely and sufficient manner, and Celtic Linen will also benefit from the removal of Kings Laundry;

<sup>&</sup>lt;sup>632</sup> See Berendsen's response to question 6 of the Second RFI.



Competition and Consumer Protection Commission

- e) there are a number of barriers to entry and expansion in the healthcare market including financial barriers i.e., capital required to develop the necessary infrastructure and non-financial barriers (e.g., reputation/experience which is an important factor for healthcare sector customers when they choose their suppliers of flat linen rental and maintenance services). The evidence provided to the Commission did not enable the Commission to conclude with sufficient confidence that any supplier will enter/expand into providing outsourced flat linen rental and maintenance services to a significant number of healthcare customers in a timely, likely and sufficient manner to mitigate or prevent the loss of competitive pressure exerted by Kings Laundry pre-transaction; and
- f) the HSE and other healthcare customers only have a limited choice of suppliers and the Proposed Transaction would remove a key supplier. The Commission has not observed evidence that healthcare customers have sufficient CBP to mitigate or prevent the identified likely resulting SLC.

# **Coordinated Effects**

4.379 Coordinated effects occur where a proposed transaction changes the nature of competition in the relevant market by making it more likely that the merged entity and some or all of its competitors will coordinate their behaviour by, for example, raising prices and/or decreasing output. Such coordination refers to actions that are profitable only as a result of each firm accommodating the reactions of others. Thus, the key question is whether a proposed transaction would materially increase the likelihood that firms in the healthcare market and the hospitality market, respectively, would successfully coordinate their behaviour or would strengthen existing coordination between firms in each aforementioned market.

#### No Evidence of Coordinated behaviour prior to the Proposed Transaction

4.380 The Commission has found no evidence of coordinated behaviour between competitors in the: (i) healthcare market; or (ii) hospitality market prior to the implementation of the Proposed Transaction. The Commission contacted a wide range of third parties during its investigation and none raised any concern about existing coordinated behaviour between linen laundries.

#### Conditions and Evidence Considered for Coordinated Behaviour

- 4.381 In assessing the potential for a proposed transaction to result in coordinated effects, the Commission will assess whether the conditions and incentives that are generally necessary for successful coordination between competitors are present or likely to arise, which includes assessing:
  - a) the ability to identify common terms of coordination; and
  - b) the ability to detect deviations from the terms of coordination, and the ability to punish deviations that would undermine the coordinated interaction.<sup>633</sup>

<sup>&</sup>lt;sup>633</sup> See paragraphs 4.25-4.31 of the Commission's Merger Guidelines for a detailed discussion of the conditions and incentives conducive to coordinated behaviour.



4.382 In this instance, the Commission assessed whether the Proposed Transaction would increase the possibility of coordination between competitors in the (i) healthcare market and, (ii) the hospitality market and then examined whether it would increase the sustainability of any such coordination.

#### The Possibility of Reaching Common Terms of Coordination

- 4.383 In order to coordinate behaviour, firms need to achieve an understanding as to how to do so. This need not involve explicit agreements among competitors, or any communication between them, nor need it involve all firms active in the relevant market or perfect coordination between firms. Coordinated behaviour can take many forms. In some markets, firms may coordinate their behaviour on prices in order to keep them above the competitive level. In other markets, coordination may aim at limiting production or the amount of new capacity brought to the market. Firms may also coordinate their behaviour by dividing the market between them, for instance by geographic area or other customer characteristics.
- 4.384 In this instance, linen laundries in the State could coordinate their behaviour by agreeing on prices charged for outsourced flat linen rental and maintenance services.

#### Healthcare Market

4.385 The Commission considered factors such as, market share symmetry, price transparency, product differentiation, market stability. Having assessed these factors, the Commission is of the view that the Proposed Transaction will not make it more likely that the merged entity and Celtic Linen will engage in coordinated behaviour in the healthcare market with the effect of substantially lessening competition in this market.

#### Hospitality Market

4.386 The Commission considered factors such as, market share symmetry, price transparency, product differentiation, market stability. Having assessed these factors, the Commission is of the view that the Proposed Transaction will not make it more likely that the merged entity and its competitors will engage in coordinated behaviour in the healthcare market with the effect of substantially lessening competition in this market.

#### Vertical Relationship

4.387 The Parties have stated in the notification that there is no vertical relationship between Berendsen and Kings Laundry. The Commission has not identified any vertical relationship between the Parties. On this basis, the Commission considers that the Proposed Transaction does not raise any vertical competition concerns in the State.

#### Efficiencies

4.388 Paragraphs 8.1 and 8.2 of the Commission's Merger Guidelines state that:

"A merger may generate various efficiencies for the merged entity. The Commission's analysis of efficiencies goes beyond the impact of efficiencies



Competition and Consumer Protection Commission

on the merged entity and focuses on whether verifiable efficiencies mitigate adverse competitive effects and prevent an SLC."

"The onus rests on the parties to show that claimed efficiencies are (i) merger specific, (ii) verifiable and (iii) benefit consumers sufficiently to prevent an SLC."

4.389 The Commission has not received any submission from the Parties on efficiencies which meets the criteria set out in paragraph 8.2 of the Commission's Merger Guidelines.



# 5. REMEDIES TO ADDRESS SLC CONCERNS

- 5.1 Section 22(3) of the Act provides that, on completion of the Phase 2 review period, the Commission shall make one of the following determinations:
  - The merger may be put into effect;
  - The merger may not be put into effect; or
  - The merger may be put into effect subject to conditions specified by the Commission.
- 5.2 The Proposed Transaction concerns the acquisition of the whole of Kings Laundry's business: that is, the supply of outsourced flat linen rental and maintenance services to both the hospitality and the healthcare markets. On completion of the Phase 2 review, the Commission has found that the Proposed Transaction will likely result in an SLC only in the healthcare market. The Commission notes that the merged entity would have a combined market share of just over [60-70]% in the healthcare market.<sup>634</sup> Kings Laundry accounts for just under [...]% of the healthcare market and so is a significant supplier. However, the Commission notes that Kings Laundry's healthcare business accounts for less than [...]% of its revenue, while the remaining [...]% of its revenue is generated from its hospitality business. The Commission, therefore, considers that, if the SLC concerns in the healthcare market are capable of being addressed either through proposals or specified conditions, it may not be proportionate to require that the merger may not be put into effect.

# **Proposals Submitted by Berendsen**

# First Proposals

5.3 Following several engagements with the parties to discuss the identified SLC concerns, Berendsen, on a without prejudice basis, submitted proposals on 14 June 2019 (the "First Proposals") pursuant to Section 20(3) of the Act. The submission of these proposals by Berendsen extended the deadline within which the Commission was required to make its determination of the Proposed Transaction in Phase 2. The extension added 15 working days to the Phase 2 period, bringing the review period to a total of 135 working days, in accordance with section 22(4B) of the Act. In the First Proposals Berendsen committed to either behavioural commitments, or to the divestment of certain contracts.

# Evaluation of the First Proposals

5.4 Following detailed consideration, the Commission concluded that the First Proposals would not address the identified SLC concerns in the healthcare market. The behavioural commitments were not appropriate to deal with the type of structural competition concerns identified in this case.<sup>635</sup> The proposed composition and scope of

<sup>634</sup> At the end of Q2 2018.

<sup>&</sup>lt;sup>635</sup> In this regard, in case <u>M/18/031 – Uniphar/SISK Healthcare</u> the Commission was concerned that this transaction could facilitate coordination of prices and terms and conditions by Uniphar. In order to address potential competition concerns and specifically the potential misuse / anti-competitive sharing of third parties' commercially sensitive information, the Commission accepted 139

the divestment package was insufficient to replace the loss of Kings Laundry as an important, independent, and credible competitor in the healthcare market.

5.5 The Commission therefore rejected the First Proposals on this basis that they were inappropriate and insufficient to address the identified SLC concerns in the healthcare market. In light of this, the Commission did not find it necessary to market test the First Proposals.

#### Second Proposals

Coimisiún um

Iomaíocht agus

Cosaint Tomhaltóirí

- 5.6 On 24 June 2019 Berendsen, on a without prejudice basis, submitted revised proposals (the "Second Proposals") to the Commission committing to dispose of a package of healthcare contracts, including [...].
- 5.7 The Commission considered that there were implementation risks associated with the Second Proposals which would render the remedy ineffective if they materialised. However, subject to the Parties addressing those implementation risks, the Second Proposals provided a divestment package which the Commission could market test.

#### Market testing of the Second Proposals

- 5.8 Over the period 25 June to 1 July 2019, the Commission market tested the Second Proposals in order to establish whether they were likely to be appropriate, proportionate and effective in addressing the identified SLC concerns in the healthcare market. Phone interviews were conducted with five competitors<sup>636</sup> and five customers.<sup>637</sup>
- 5.9 All of the customers interviewed expressed concern regarding the reduction in the number of significant suppliers in the healthcare market from 3 to 2. All of the customers expressed a preference for having more than 2 potential bidders in the market. Three of the five customers noted that if the number of possible bidders was to reduce from 3 to 2, they would be concerned about price increases and a degradation in quality of service.
- 5.10 Two customers thought the purchase of a divested package of contracts would not allow the establishment of a new competitor. Their reasons for coming to this view were that any new entrant would require significant financial resources and experience of serving healthcare customers. Two customers thought that a divested package of contracts that included hospitals would be sufficient to enable the purchaser of the divested contracts to become an effective competitor.
- 5.11 Healthcare customers also listed key criteria that they would expect a new supplier of outsourced flat linen rental and maintenance services to meet. Most commonly cited factors were: (i) adherence to infection control/prevention guidelines; (ii) service reliability; (iii) customer service; (iv) experience of providing flat linen rental and

binding behavioural commitments submitted by Uniphar that prevented the exchange of competitively sensitive information between Uniphar and SISK Healthcare.

<sup>&</sup>lt;sup>636</sup> [...]. [...] expressed no interest in entering the healthcare market. For the purposes of the market testing, competitors included laundries not currently active in the healthcare market, but with some potential to enter.

<sup>637 [...]. [...]</sup> and [...] did not respond.



- 5.12 Four out of five competitors thought that the divestment of contracts could be an effective remedy, but this was subject to the scope and nature of divested contracts (e.g., size and location of customers). Several competitors also noted that some assets (e.g., linen stock and trolleys) should be divested alongside contracts to reduce the level of investment that a prospective purchaser would need to incur when buying divested contracts.
- 5.13 Three out of five competitors [...] expressed interest in principle in purchasing healthcare contracts. All of these [competitors] noted that more information about the nature of the contracts to be divested (e.g., price, contract length, locations of customers, type of work, frequency of deliveries etc.) would be required before they could fully confirm interest.
- 5.14 On 2 July 2019 the Commission informed the Parties of the results of the market testing during a telephone conference.

#### Evaluation of the Second Proposals

Coimisiún um

lomaíocht agus

Cosaint Tomhaltóirí

- 5.15 The Commission was of the view that the divestment of the healthcare contracts envisaged under the Second Proposals (the "Divestment Package") could have been sufficient in order for a prospective suitable purchaser to replace the competitive constraint exerted by Kings Laundry in the healthcare market.
- 5.16 While the Commission considered that the substance of the Second Proposals may have addressed its competition concerns, it had significant remaining concerns regarding the implementation of the measures. These were post-completion proposals that posed serious implementation risks. The Commission engaged extensively with the Parties in relation to these concerns. Ultimately, however, the Parties were unable to address the implementation risks associated with the Second Proposals.

#### Third Proposals

- 5.17 On 8 July 2019, Berendsen submitted a revised set of proposals (the "Third Proposals"). In the Third Proposals, Berendsen committed to:
  - Divest three [...] healthcare contracts, [...] to a third party purchaser approved by the Commission;
  - Divest such additional healthcare contracts of an aggregate value (by reference to 2018 revenue figures), which, when aggregated with the value of the [...] healthcare contracts referred to above, have a total value of [...]; and
  - Divest the rights and title in ancillary items such as linen stock (but excluding, for the avoidance of doubt, any facilities or fixed assets (e.g., washers, dryers, or trucks)) as required by a Third Party Purchaser and to otherwise assist the Third Party Purchaser to enable it to provide the services in respect of the relevant Healthcare Contracts.



- or
- If it became apparent to Berendsen that it was unable to divest the three [...] healthcare contracts in the manner described above, it would enter into discussions with the Commission to identify an alternative package of healthcare contracts that could be divested to an approved third party purchaser(s) and that would address the Commission's competition concerns.
- 5.18 Berendsen committed to refraining from completing the Proposed Transaction unless and until the Commission has confirmed that it has complied with the Third Proposals.

#### Evaluation of the Third Proposals

- 5.19 The Commission considers that the Third Proposals submitted by the Parties address its SLC concerns in the healthcare market (the "Proposals"). The Proposals are included as Chapter 9 of this determination. The divestment of a package of healthcare contracts as described in the Proposals will allow the approved third party purchaser to provide effective competition to the merged entity and Celtic Linen. Berendsen's commitment to refrain from completing the Proposed Transaction until it has completed the divestment of the healthcare contracts addresses the Commission's concerns regarding implementation risks.
- 5.20 The Commission has taken the Proposals into account and, in light of the Proposals (which form part of the basis of the Commission's determination) has determined, in accordance with section 21(2)(a) of the Act, that the result of the proposed acquisition whereby Berendsen would acquire sole control of Kings Laundry will not be to substantially lessen competition in any market for goods or services in the State, and, accordingly, that the acquisition may be put into effect.



# 6. ANCILLARY RESTRAINTS

6.1 The SPA contains a number of restrictive obligations on Kings Laundry. These include non-compete and non-solicitation clauses. None of these restrictive obligations exceeds the maximum duration acceptable to the Commission<sup>638</sup>. The Commission considers these restrictions to be directly related to and necessary for the implementation of the Proposed Transaction insofar as they relate to the State.

<sup>&</sup>lt;sup>638</sup> In this respect, the Commission follows the approach adopted by the EU Commission in paragraphs 20 and 26 of its "Commission Notice on restrictions directly related and necessary to concentrations" (2005). For more information see <u>http://eurlex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:52005XC0305(02)&from=EN</u>


## 7. CONCLUSION

7.1 In light of the binding divestiture proposals submitted by Berendsen and in light of its analysis as set out in this Determination, the Commission has determined that the Proposed Transaction will not substantially lessen competition in any market for goods or services in the State.



### 8. DETERMINATION

Pursuant to section 20(3) of the Competition Act 2002, as amended (the "Act"), Berendsen Ireland Limited ("Berendsen") has submitted to the Competition and Consumer Protection Commission (the "Commission") the proposals set out below regarding measures to be taken to ameliorate any effects of the proposed acquisition on competition in markets for goods or services in the State, with a view to the said proposals becoming binding on Berendsen.

The Commission has taken the proposals into account and, in light of the said proposals (which form part of the basis of its determination), has determined, in accordance with section 22(3)(a) of the Act, that the result of the proposed acquisition, whereby Berendsen would acquire sole control of Kings Laundry Limited, will not be to substantially lessen competition in any market for goods or services in the State, and, accordingly, that the acquisition may be put into effect. Before making a determination in this matter, the Commission, in accordance with section 22(8) of the Act, had regard to any relevant international obligations of the State, and concluded that there were none.

For the Competition and Consumer Protection Commission

Isolde Goggin Chairperson Competition and Consumer Protection Commission



### 9. BERENDSEN'S PROPOSALS

#### M/18/063 – Berendsen (Elis)/Kings Laundry

#### Proposals by Berendsen to the Commission (the "Proposals")

#### 8 July 2019

#### Recitals

- A. The Proposed Transaction, as defined below, was notified to the Commission on 7 August 2018.
- B. Berendsen and Kings Laundry are both active in the supply of outsourced rental and maintenance of flat linen in the healthcare sector in the State.
- C. Pursuant to Section 20(3) of the Act, Berendsen, has submitted the Proposals to the Commission for the purpose of ameliorating the Commission's competition concerns in respect of the effect of the Proposed Transaction on competition in markets for goods or services in the State, with a view to the Proposals becoming binding on Berendsen on the date of the Determination.
- D. It is understood that Completion will only occur on condition that Berendsen has satisfied its obligations under paragraphs 6 or 7 below in relation to the Healthcare Contracts (as defined below) (and healthcare contracts, as the case may be). Therefore, in practical terms Completion and the satisfaction by Berendsen of its obligations to deal with the Healthcare Contracts in compliance with paragraph 6 or Healthcare Contracts (and healthcare contracts, as the case may be) in compliance with paragraph 7 of these Proposals are interconditional. Accordingly, it is the intention of Berendsen to arrange for Completion to happen contemporaneously with the satisfaction by it of its obligations to deal with the Healthcare with paragraphs 6 or 7 below. In any event, for the avoidance of doubt, Completion will not take place before Berendsen has fulfilled its obligations regarding the Healthcare Contracts (and healthcare contracts, as the case may be) in accordance with paragraphs 6 or 7 below.
- E. In its discussions with Berendsen, the Commission has confirmed that in the event that the provisions of paragraph 17 are invoked, the Commission will act reasonably in the context of assessing alternative package(s) which Berendsen may be required to put to the Commission to dispose of Healthcare Contracts (and healthcare contracts, as the case may be) and in particular as regards how those alternative package(s) to dispose of such contracts can be packaged together and as regards the number of Third Party Purchasers (as defined below) to whom the relevant contracts can be offered in order to address the Commission's competition concerns.

#### Definitions

For the purpose of these Proposals, the following terms shall have the following meaning:

"Act" means the Competition Act 2002, as amended;



"Berendsen" means Berendsen Ireland Limited, a wholly-owned subsidiary of Berendsen Limited (UK) and an indirect wholly-owned subsidiary of Elis;

"Berendsen Manager" has the meaning ascribed to it in paragraph 27;

"**Commission**" means the Competition and Consumer Protection Commission and its successors;

"Commission Monitoring Trustee" has the meaning ascribed to it in paragraph 20;

"**Completion**" means completion of the Proposed Transaction in accordance with the terms of the Share Purchase Agreement;

"**Determination**" means, for the purposes of these Proposals only, the determination of the Commission pursuant to Section 22(3)(a) of the Act that the Proposed Transaction may be put into effect taking into account the Proposals, which form part of the basis of the Determination;

"Elis" means Elis S.A.;

"Healthcare Contracts" means contracts in place between Berendsen and its [...] customers for the supply of rental and maintenance of flat linen, and "Healthcare Contract" shall be construed accordingly;

"Kings Laundry" means Kings Laundry Limited;

"Kings Laundry Business" means the outsourced flat linen laundry business acquired by Berendsen pursuant to the Share Purchase Agreement;

"Long Stop Date" means

- (a) the period of [...] from the date of the Determination or (i) such later date as the Commission, acting reasonably, may agree in writing or (ii) such extended period as may result from the delays referred to in paragraphs 14, 18, 19 and 20 below; or
- (b) in the case of where:
  - (i) a binding contract has been entered into for the sale of Healthcare Contracts; or
  - a binding contract has been entered into in respect of the sale of Healthcare Contracts (and healthcare contracts, as the case may be) pursuant to the review provisions herein,

it means, [...] from the date of entry into such binding contract (or such later date as may be agreed by the Commission acting reasonably following consultation with Berendsen or as extended under paragraph 10 below or such extended period as may result from the delays referred to in paragraphs 14, 18, 19 and 20); and

in any event, a date no later than [...] from the date of the Determination.

"Monitoring Mandate" has the meaning ascribed to it in paragraph 21;



"Monitoring Trustee" has the meaning ascribed to it in paragraph 21;

"Monitoring Trustee Reports" has the meaning ascribed to it in paragraph 23;

"New Monitoring Trustee" has the meaning ascribed to it in paragraph 19;

"Package A" and "Package A Contract(s)" have the meanings ascribed to them in paragraph 6;

"**Proposed Transaction**" means the proposed acquisition of sole control of Kings Laundry by Berendsen pursuant to the terms of the Share Purchase Agreement as notified to the Commission on 7 August 2018;

"Proposed Monitoring Trustee" has the meaning ascribed to it in paragraph 18;

"Relevant Customer" has the meaning ascribed to it in paragraph 8 and "Relevant Customers" shall be construed accordingly;

"Share Purchase Agreement" means the share purchase agreement, dated 25 July 2018, entered into between Berendsen and the owners of Kings Laundry for the purchase of the entire issued share capital of Kings Laundry;

"Third Party Purchaser" means an independent third party purchaser that is not connected to Elis, Berendsen or Kings Laundry within the meaning of section 220 of the Companies Act 2014 and "Third Party Purchasers" shall be construed accordingly;

"Third Party Sale" has the meaning ascribed to it in paragraph 8;

"Working Day" means a day (other than a Saturday or a Sunday or public holiday) on which banks are open for retail business in Dublin.

#### Part I

#### **Proposals relating to the Healthcare Contracts**

- Berendsen hereby undertakes to the Commission to procure, in advance of Completion, to the extent it is within its powers of procurement to do so, in compliance with paragraph 6 or 7, the divestment of the Healthcare Contracts (and the healthcare contracts, as the case may be) by the Long Stop Date to a Third Party Purchaser approved by the Commission, having regard to the provisions of paragraph 4, (whose approval shall not be unreasonably withheld or delayed).
- 2. Berendsen recognises that the acquisition of the Healthcare Contracts (and the healthcare contracts, as the case may be) by a Third Party Purchaser must not be likely to create, in light of information available to the Commission, prima facie competition concerns.
- 3. Berendsen recognises that for a Third Party Purchaser to meet with the Commission's approval, such purchaser shall be independent of and unconnected to Berendsen, Elis and Kings Laundry and is one that is reasonably likely to obtain all authorisations and

regulatory consents (if legally required) in relation to the acquisition of the Healthcare Contracts (and the healthcare contracts, as the case may be).

- 4. In assessing the suitability of a Third Party Purchaser, the Commission may have regard to factors such as its financial resources, expertise in the rental and maintenance of flat linen, contingency planning and whether such Third Party Purchaser is committed to maintaining and developing the Healthcare Contracts (and the healthcare contracts, as the case may be) and being an active competitive force in the healthcare market.
- 5. In assessing the suitability of a Third Party Purchaser, the Commission shall accept the decision of a Relevant Customer that its Healthcare Contract can be divested by Berendsen to the relevant Third Party Purchaser, subject to the provisions of paragraph 15 below.
- 6. Berendsen hereby undertakes to the Commission to procure, to the extent it is within its powers of procurement to do so, the divestment to a Third Party Purchaser by the Long Stop Date all of its rights, title and interests in respect of:
  - (i) three (3) Healthcare Contracts, [...]; and
  - such additional Healthcare Contracts of an aggregate value (by reference to 2018 revenue figures), which, when aggregated with the value of the Healthcare Contracts referred to in (i) above, have a total value of [...]. Berendsen also undertakes to the Commission to procure, to the extent it is within its powers of procurement to do so, the divestment of the rights and title in ancillary items such as linen stock (but excluding, for the avoidance of doubt, any facilities or fixed assets (e.g., washers, dryers, or trucks)) as required by a Third Party Purchaser and to otherwise assist the Third Party Purchaser to enable it to provide the services in respect of the relevant Healthcare Contracts ("Package A" and the relevant contract(s) therein being each a "Package A Contract(s)").
- 7. In circumstances where it becomes apparent to Berendsen (acting reasonably and in good faith) and to the Monitoring Trustee (who will confirm this to the Commission in writing) at any time prior to the Long Stop Date that it will not be possible to complete the divestment of Package A by the Long Stop Date, both Berendsen and the Commission undertake to consider, in good faith (and for this purpose the Commission further undertakes to invoke the review provisions in paragraph 17) an alternative package of Healthcare Contracts (and healthcare contracts, as the case may be) which could be divested by Berendsen in a form which would address the Commission's concerns. In this regard, the Commission shall have due regard to:
  - the combination of Healthcare Contracts and Berendsen's other healthcare contracts available at that time for the purposes of putting an alternative package(s) together to achieve the aggregate value of [...] (by reference to 2018 revenue figures); and

- (ii) the circumstances pertaining at the time and, in particular, to the status and standing of any Third Party Purchaser (in the relevant market(s)) who may be prepared to acquire the relevant contracts.
- 8. For the purposes of paragraphs 6 and 7, Berendsen acknowledges and agrees that it shall not actively solicit healthcare flat linen business from a customer who was party to a Healthcare Contract or to another of its healthcare contracts which has been divested pursuant to these Proposals (each a "Relevant Customer") for a period of [...] from the date of completion of a sale to a Third Party Purchaser (each a "Third Party Sale").
- 9. For the purposes of:

Coimisiún um

Iomaíocht agus

Cosaint Tomhaltóirí

- (i) Package A, if Berendsen enters into a binding contract (which for the avoidance of doubt may contain pre-conditions to its completion) with a Third Party Purchaser approved by the Commission for the sale of such Healthcare Contracts and such transaction completes within [...] of the date of entry into the relevant binding contract (or such longer period as may be agreed by the Commission acting reasonably following consultation with Berendsen or as extended under paragraph 10), and/or
- (ii) any alternative package(s) agreed under paragraph 7 and/or 17 below, if Berendsen enters into a binding contract (which for the avoidance of doubt may contain pre-conditions to its completion) with a Third Party Purchaser approved by the Commission for the sale of such Healthcare Contracts (and healthcare contracts, as the case may be) and such transaction completes within [...] of the date of entry into the relevant binding contract (or such longer period as may be agreed by the Commission acting reasonably following consultation with Berendsen or as extended under paragraph 10),

this shall constitute evidence that Berendsen has complied with Part I of these Proposals, provided that (for the avoidance of doubt) Completion has not taken place in advance of (i) or (ii) and the Commission has confirmed in writing that Berendsen has complied with Part I of these Proposals and may proceed to Completion. In this regard, the Commission agrees to provide this written confirmation promptly to facilitate near simultaneous completion of the divestment of the Healthcare Contracts (and healthcare contracts, as the case may be) and Completion.

- 10. The Commission acknowledges that it is the intention of Berendsen to complete a Third Party Sale contemporaneously with Completion and accordingly the Commission agrees to extend any relevant deadlines (subject to the Long Stop Date) set out herein in accordance with any reasonable request received from Berendsen in order for Completion and the completion of a Third Party Sale to happen contemporaneously.
- 11. Berendsen shall:
  - a. as soon as practicable, inform the Monitoring Trustee and the Commission in writing, with a fully documented and reasoned proposal, of any Third Party

Purchaser who indicates a genuine intention to purchase the Healthcare Contracts (and the healthcare contracts, as the case may be), enabling the Monitoring Trustee and the Commission to verify the suitability of the Third Party Purchaser; and

- b. when Berendsen and the Third Party Purchaser have entered into a binding contract for the sale of any Healthcare Contract, submit to the Monitoring Trustee and the Commission a copy of a binding contract(s) and a fully documented and reasoned report enabling the Monitoring Trustee and the Commission to verify that the requirements set out in these Proposals are fulfilled and that there has been no material change in the status of the Third Party Purchaser not reasonably foreseeable at the time the Commission assessed that purchaser's suitability, subject to the Commission agreeing to keep confidential all such information received.
- 12. Within [...] of receipt of a fully documented and reasoned proposal from Berendsen in accordance with paragraph 11(a), the Monitoring Trustee shall provide a written report to the Commission verifying the suitability of the Third Party Purchaser(s). The Commission shall communicate in writing its approval or non-approval of a Third Party Purchaser within [...] of receipt of a fully documented and reasoned proposal from Berendsen identifying that purchaser in accordance with paragraph 11(a).
- 13. Within [...] of receipt of a copy of a binding contract(s) and a fully documented and reasoned report from Berendsen in accordance with paragraph 11(b), the Monitoring Trustee shall provide a report to the Commission verifying whether the requirements set out in these Proposals have been fulfilled and that there has been no material change in the status of the Third Party Purchaser. Within [...] of receipt of a copy of a binding contract(s) and accompanying fully documented and reasoned report in accordance with paragraph 11(b), the Commission shall communicate in writing its view as to whether there has been any material change in the status of the Third Party Purchaser and whether the requirements set out in these Proposals have been fulfilled as provided for in paragraph 11(b).
- 14. Failure of the Commission to communicate its approval or non-approval of a Third Party Purchaser within ten (10) Working Days of receipt of a fully documented and reasoned proposal identifying such a purchaser in accordance with paragraph 11(a) shall postpone the running of the relevant periods in the definition of Long Stop Date until such date as the Commission communicates to Berendsen its approval or non-approval. However, if the Commission does not communicate its approval or non-approval within twenty (20) Working Days of receipt of a fully documented and reasoned proposal, such approval shall be deemed to have been given unconditionally.
- 15. It is acknowledged and agreed by Berendsen, for the purposes of divesting Package A, that the relevant Healthcare Contracts shall be divested to a single Third Party Purchaser. It is acknowledged and agreed by the Commission for the purposes of paragraph 7 above, that it will consider in good faith the possibility of Berendsen divesting the Healthcare Contracts (and healthcare contracts, as the case may be) to more than one (1) Third Party Purchaser so as to afford Berendsen the best opportunity



to divest such contracts. In the case of a plurality of offers from Third Party Purchasers to whom the Commission does not object, Berendsen shall be free to accept any offer or offers, or to select the offer or offers it considers best, provided that the Commission has confirmed in writing that its competition concerns would be addressed.

16. In circumstances where it has not been possible to complete the divestment of Package A, or the divestment of any alternative package of Healthcare Contracts (and healthcare contracts, as the case may be) agreed by Berendsen and the Commission under the review provisions in paragraph 17 by the Long Stop Date, Berendsen will terminate the Share Purchase Agreement.

#### **Proposals review clause**

17. The Commission may at its sole discretion, either in response to a reasoned request from Berendsen or the Monitoring Trustee or, in appropriate cases on its own initiative, extend any of the time periods provided for in these Proposals, with the exception of the time periods applying to the Commission mentioned in paragraphs 18, 19 and 20 below unless with the agreement of Berendsen (and such agreement not to be unreasonably withheld). The Commission may further, at its sole discretion, either in response to a reasoned request from Berendsen or the Monitoring Trustee showing good cause or, in appropriate cases on its own initiative (with the agreement of Berendsen and such agreement of substitute any provision in these Proposals.

#### Part II

#### Appointment of a Trustee in respect of the Proposals in Part I

#### **Appointment of a Monitoring Trustee**

- 18. Within [...] of the date of the Determination, Berendsen will propose to the Commission a trustee, who is independent of Berendsen, Elis and Kings Laundry (the "Proposed Monitoring Trustee"). The appointment of the Proposed Monitoring Trustee is subject to the approval of the Commission (such approval not to be unreasonably withheld or delayed). If the Commission does not reject the Proposed Monitoring Trustee by notice in writing within [...] of the date of the proposal by Berendsen, the Proposed Monitoring Trustee shall be deemed to have been approved.
- 19. If the Proposed Monitoring Trustee is rejected by the Commission, Berendsen will propose the name of a new trustee (the "**New Monitoring Trustee**") within [...] of being informed by the Commission of the rejection. If the Commission does not reject the New Monitoring Trustee by notice in writing to Berendsen within [...] of the new proposal, the New Monitoring Trustee shall be deemed to have been approved.
- 20. If the New Monitoring Trustee is rejected by the Commission, the Commission shall nominate a suitable trustee (the "**Commission Monitoring Trustee**") within [...], whom Berendsen will appoint or cause to be appointed.



- 21. Within [...] of the date on which the Commission has approved or is deemed to have approved either the Proposed Monitoring Trustee, the New Monitoring Trustee or the Commission Monitoring Trustee, Berendsen shall enter into a mandate agreement (the "Monitoring Mandate") with the approved trustee (the "Monitoring Trustee"), the terms of which shall have previously been agreed with the Commission, which confers on the Monitoring Trustee all the rights and powers necessary to permit the Monitoring Trustee to monitor compliance with the Proposals.
- 22. The Monitoring Trustee shall be independent of Berendsen, Elis and Kings Laundry, possess the necessary qualifications and experience to carry out its mandate and shall neither have nor become exposed to a conflict of interest.

#### Monitoring Trustee's Mandate

- 23. Pursuant to the Monitoring Mandate, throughout the duration of the Monitoring Trustee's appointment, the Monitoring Trustee shall, in respect of the Proposals:
  - a. Provide written reports (the "Monitoring Trustee Reports") to the Commission on the progress of the discharge of its duties under the Mandate, identifying any respects in which the Monitoring Trustee has been unable to discharge such duties. The Monitoring Trustee Reports shall be provided at monthly intervals, commencing one (1) month after the date of the appointment of the Monitoring Trustee, or at such other times or time periods as the Commission may specify and are notified in writing to Berendsen. Berendsen shall receive a non-confidential copy of such Monitoring Trustee Reports;
  - b. Monitor and advise the Commission as to the developments in selecting a Third Party Purchaser and as to the conduct of the negotiations;
  - c. Monitor and advise the Commission on whether a Third Party Purchaser with whom Berendsen intends to negotiate is likely to satisfy the Commission's requirements as to suitability, including providing a written report (pursuant to paragraph 12 above); and
  - d. Provide a written report verifying whether the requirements set out in these Proposals have been fulfilled and that there has been no material change in the status of the Third Party Purchaser (pursuant to paragraph 13 above).

#### Miscellaneous

- 24. The Monitoring Trustee's duties and functions as set out above shall not be extended or varied in any way by Berendsen save with the express written approval of the Commission (whose approval shall not be unreasonably withheld or delayed). Any instruction or request to the Monitoring Trustee from Berendsen which conflicts with the terms of the Monitoring Mandate and the duties and functions as set out above, will be considered null and void.
- 25. The Commission may, on its own initiative or at the request of the Monitoring Trustee, give any orders or instructions to the Monitoring Trustee that are required in order to



ensure compliance with the Proposals so long as Berendsen is first given a reasonable opportunity to comment on any such orders or instructions in advance.

- 26. In respect of the Proposals, the Monitoring Mandate shall expire once the Healthcare Contracts have been dealt with in compliance with paragraph 6 or the Healthcare Contracts (and healthcare contracts, as the case may be) have been dealt with in compliance with paragraph 7 and the Commission has confirmed in writing to Berendsen that it is satisfied that there has been compliance with the Proposals (it being understood that such confirmation from the Commission shall not be unreasonably withheld or delayed).
- 27. Berendsen shall appoint an employee within Berendsen with responsibility for the management of the Healthcare Contracts (the "Berendsen Manager"). The Berendsen Manager shall continue to manage all operations relating to the Healthcare Contracts in the best interest of those contracts with a view to ensuring their continued economic viability, marketability, and competitiveness.
- 28. The Monitoring Trustee shall have full and complete access to the Berendsen Manager and any other employee of Berendsen, as required, in order to ensure compliance with the obligation to maintain and preserve the economic viability, marketability and competitiveness of the Healthcare Contracts.
- 29. Berendsen will provide the Monitoring Trustee with all reasonable assistance and will procure (to the extent it is in its powers of procurement to do so) that all relevant third parties provide such assistance required to ensure compliance with the Proposals. Berendsen will provide or cause to be provided to the Monitoring Trustee all such assistance and information, including copies of all relevant documents accessible by Berendsen as the Monitoring Trustee may require in carrying out its Monitoring Mandate and will pay reasonable remuneration for the Monitoring Trustee's services.
- 30. In addition, at the expense of Berendsen, the Monitoring Trustee may appoint an advisor, subject to Berendsen's approval (whose approval shall not be unreasonably withheld or delayed) if the Monitoring Trustee considers the appointment of such advisor is strictly necessary and appropriate for the performance of its duties and obligations under the Monitoring Mandate, provided that any fees and other expenses incurred by the Monitoring Trustee are reasonable. Should Berendsen refuse to approve the advisor proposed by the Monitoring Trustee, the Commission may approve the appointment of such advisor, having first consulted with Berendsen. Only the Monitoring Trustee shall be entitled to issue instructions to any such advisor.
- 31. Notwithstanding the Monitoring Trustee's overall responsibility to discharge its functions and in particular notwithstanding the Monitoring Trustee's position as an independent unrelated third party, the Monitoring Trustee (who shall undertake in the Monitoring Mandate to do so) shall have to the extent possible, given the nature of its tasks, due regard to the commercial interests of Berendsen.

Competition and Consumer Protection Commission



- 32. Berendsen shall indemnify the Monitoring Trustee and its employees and agents (each an "Indemnified Party") and hold each Indemnified Party harmless against, and hereby acknowledges and agrees that an Indemnified Party shall have no liability to Berendsen for, any liabilities arising out of the performance of the Monitoring Trustee's duties under the Monitoring Mandate, except to the extent that such liabilities result from the wilful default, recklessness, negligence and/or bad faith of the Monitoring Trustee, its employees, agents and/or advisor.
- 33. The Monitoring Mandate shall be deemed to be discharged and the Monitoring Trustee's appointment shall be deemed to be terminated if Berendsen announces and demonstrates to the Commission that the Proposed Transaction has been irrevocably abandoned.

#### Monitoring Trustee review clause

34. Subject to the provisions of paragraph 17 above, the Commission may, at its sole discretion, in response to a reasoned request from Berendsen or the Monitoring Trustee showing good cause or, in appropriate cases on its own initiative (with the agreement of Berendsen and such agreement not to be unreasonably withheld) waive, modify or substitute any provision in relation to the appointment of the Monitoring Trustee.

Signed by:

Jack Byron

Director

Duly authorised for and on behalf of Berendsen

8 July 2019



Competition and Consumer Protection Commission

#### **Confidential Annex I**

#### Healthcare Contracts

[...]<sup>639</sup>

<sup>&</sup>lt;sup>639</sup> [...].



## 10. APPENDIX 1 – SUMMARY OF CUSTOMERS' AND SELF-SUPPLIERS' RESPONSES

10.1 This appendix provides a summary overview of the responses obtained from the Customers and Self-suppliers. These are considered and referred to throughout this Determination.

#### **Background and objectives**

- 10.2 In order to further inform the Commission's understanding of the structure of the market and the potential impact of the Proposed Transaction the Commission issued questionnaires to some of the larger organisations in healthcare and hospitality sectors. Some of these organisations are purchasing outsourced flat linen rental and maintenance services from the parties while some of these organisations have OPLs. The Commission's Market Enquiries were then supplemented by the Amárach Survey which targeted a broader range of organisations from the healthcare and hospitality sectors. The Amárach Survey is presented alongside this Determination.
- 10.3 The Commission's Market Enquiries examined, amongst other things:
  - a) Pricing structure and contract information;
  - b) Information on the criteria which are typically used by customers in the hospitality market and in the healthcare market when selecting providers of flat linen rental and maintenance services;
  - c) Information on procurement methods employed tenders or quotations etc.;
  - d) Information and views on switching providers in the sector;
  - e) Information and views on switching to: (i) self-supply by building an OPL; or (ii) wash-only launderettes, in response to a 5% 10 % price increase of laundry services; and
  - f) Information on the impact of the Proposed Transaction on purchasers of flat linen rental and maintenance services.
- 10.4 The Commission is mindful that information obtained from the Customers and Selfsuppliers, while a useful practical means of gathering information on customer and business preferences/behaviours, need to be interpreted with care and that stated preferences of Customers and Self-suppliers can differ from how they behave in practice. Therefore, the Commission does not solely or overly rely on information obtained from the Customers and Self-suppliers in forming its conclusions.



#### Research methodology and customer profile

- 10.5 The Commission relied on information provided by the Parties in the Merger Notification Form in order to prepare a list of purchasers of outsourced flat linen rental and maintenance services and organisations with OPLs in both healthcare and hospitality sectors.
- 10.6 The Commission circulated questionnaires to 20 Customers<sup>640</sup> and 11 Self-suppliers<sup>641</sup> in the healthcare and hospitality sectors. For each Customer and Self-supplier, the Parties identified relevant person responsible for making flat linen rental and maintenance purchasing decisions (or, in case of Self-suppliers, responsible for managing an OPL).
- 10.7 The responses were obtained by a number of means including respondents returning completed questionnaires, telephone interviews based on the questionnaire and, in two cases, face to face meetings. The information collected via phone calls have been summarised in notes with these summaries being confirmed as accurate reflections of interviews by interviewees.
- 10.8 In total, 15 responses from Customers (nine from the healthcare sector<sup>642</sup> and six from the hospitality sector<sup>643</sup>) and 8 responses from Self-suppliers (three from the healthcare sector<sup>644</sup> and five from the hospitality sector<sup>645</sup>) were obtained representing a response rate of 75% and 73% respectively. Not all respondents provided information in relation to all questions contained in the Commission's questionnaires and this is reflected in the analysis below. Table 21 and Table 22 present profiles of Customers and Self-suppliers that responded to the Commission's questionnaires.

Sector	Hospitality	Healthcare	Total
	6 (40%)	9 (60%)	15
Service provider <sup>646</sup>			
Berendsen	2 (40%)	3 (60%)	5
Kings Laundry	5 (50%)	5 (50%)	10
Other	3 (60%)	2 (40%)	5

#### *Table 21: Profile of Customers that provided responses*

Source: The Commission's analysis

643 These are [...].

<sup>&</sup>lt;sup>640</sup> 9 in the hospitality sector and 11 in the healthcare sector. See footnote 21 for the full list of contacted Customers.

<sup>&</sup>lt;sup>641</sup> 6 in the hospitality sector and 5 in the healthcare sector. See footnote 23 for the full list of contacted Customers.

<sup>&</sup>lt;sup>642</sup> These are the [...].

<sup>644</sup> These are [...].

<sup>645</sup> These are [...].

<sup>&</sup>lt;sup>646</sup> Four Customers use more than one service provider.



Sector	Hospitality	Healthcare	Total
	5 (62.5%)	3 (37.5%)	8
Previously purchased flat linen rental and maintenance services			
Yes	2 (100%)	0 (0%)	2
No	3 (50%)	3 (50%)	6

#### Table 22: Profile of Self-suppliers that provided responses

Source: The Commission's analysis

#### Summary of responses provided by Customers

10.9 The Commission assessed and summarised Customers' responses with a summary presented in paragraphs 10.10 to 10.36 below.

#### Services, pricing structure and terms of contracts

- 10.10 All 13 Customers that responded indicated that they opt for a service package that includes both the rental and maintenance of flat linen. Some customers purchase a mix of services (e.g., maintenance only service for some items (e.g. towels) and rental and maintenance service for other items (e.g. bed linen). Typical items rented by respondents include bed linen, table linen, towels and scrub suits (for healthcare sector Customers).
- 10.11 When asked about the pricing structure applied by linen laundries, all 13 Customers said that they are charged on a fee per item basis.
- 10.12 In relation to the term of contracts, 8 Customers (5 healthcare sector Customers, 3 hospitality sector Customers) indicated that the normal length of contract is three years, with four of these (2 healthcare sector Customers, 2 hospitality sector Customers) mentioning that there is an option of a contract extension. 4 Customers (2 healthcare sector Customers, 2 hospitality sector Customers, 2 hospitality sector Customers) indicated that they have rolling annual contracts and one healthcare sector Customer noted that its contract was for 2 years.

#### Tenders and quotations

- 10.13 The Customers have also been asked whether they tender their requirements for flat linen rental and maintenance services or whether they seek quotations for those services.
- 10.14 13 responses were received and 11 Customers indicated that they use a tender process (8 healthcare sector Customers, 3 hospitality sector Customers), one hospitality sector Customer indicated that it seeks quotes and one hospitality sector Customer said that it had no formal process recently and that bilateral negotiations are conducted with their current supplier. One healthcare sector Customers explained that while it uses formal procurement, their last ITT was issued several years ago. Another healthcare



sector Customer indicated that their latest contract renewal was made by mutual agreement with its current supplier of flat linen rental and maintenance services.

10.15 One healthcare sector Customer indicated that they typically tender the services on a regional basis. One hospitality sector Customer noted that the last time it issued an ITT, the decision was made to consolidate the purchase of flat linen services from a single provider (previously individual hotels were purchasing services from different suppliers). The decision to give the contract to one supplier was made for convenience of dealing with one provider, and to obtain better price.

Main criteria used when choosing a provider of flat linen rental and maintenance services

10.16 The typical criteria used by the customers when selecting a provider of flat linen rental and maintenance services is quality of linen, frequency of delivery, price and quality of service. The importance of pricing is evidenced by the following quotes:

*"that pricing would never be weighted less than 50%"<sup>647</sup>.* 

"pricing accounted for 40% of the marks with other criteria related to quality of service making 60% of the marks"<sup>648</sup>

"pricing (40% of marks) was important, but not the sole determinant"<sup>649</sup>.

- 10.17 14 Customers outlined their criteria used for selecting linen laundries, five (four healthcare sector Customers and 1 hospitality sector Customers) said they carry out site inspections and four (all healthcare sector Customers) mentioned that regulatory standard must be compiled with. Another healthcare sector Customer noted that infection control and service level agreement is a "*must have*" requirement. Table 23 illustrates examples of detailed criteria provided by respondents when selecting a provider of flat linen rental and maintenance services.
- 10.18 Two healthcare sector Customers, highlighted the importance of previous experience with one saying that previous experience in healthcare is essential and the other saying that one of its requirements included 3-year minimum experience criteria. This Customer also noted that they want the successful applicant to have experience managing linen/laundry/associated services for hospitals that are similar to them in terms of size.

Example A		Example B	
Ultimate Cost	40%	Cost	20%
Product Quality	25%	Innovation/CI	20%
Operational/Technical Infrastructure	15%	Quality of Service	20%

Table 23: Examples of criteria used in selection of provider of flat linen rental and maintenance services

<sup>&</sup>lt;sup>647</sup> See the Commission's phone record with minutes of meeting with [...] dated 6 December 2018.

<sup>&</sup>lt;sup>648</sup> See the Commission's phone record with [...] dated 19 December 2018.

<sup>&</sup>lt;sup>649</sup> See the Commission's phone record with [...] dated 7 January 2019.

Example A		Examp	ole B
Contract Specific Knowledge	5%	Proposed 10%	
		Mobilisation	
Data Management	5%	Auditing Tool	10%
Contract Delivery Model	5%	Management 10%	
		Methodology	
Business Continuity Plan	5%	Presentation	10%

Source: The Commission's analysis based on information provided by Customers.

10.19 In response to a question on the importance of the location of flat linen rental and maintenance services provider's facility when selecting service providers, from the 14 responses received, 10 Customers said that the location of the linen laundry's facilities was not important, once they could guarantee delivery, and four said it was important to some extent, (3 healthcare sector Customers, 1 hospitality sector Customer). Of those who mentioned it was important to some extent, two healthcare sector Customers indicated it was important in the context of contingency plans and one healthcare sector Customer said it was important because of the daily delivery requirement.

#### Switching providers

- 10.20 The Commission then asked questions on the ease/difficulty of switching between providers of flat linen rental and maintenance services.
- 10.21 8 (5 healthcare sector Customers, 3 hospitality sector Customers) responses were received in relation to the ease of switching while in contract. Five (3 healthcare sector Customers, 2 hospitality sector Customers) indicated that It was relatively easy to switch and one (healthcare sector Customers) said that it would be difficult, but not unachievable with no financial penalties, whereas two Customers said it may incur penalties for breach of contract.
- 10.22 12 (8 healthcare sector Customers, 4 hospitality sector Customers) responses were received in relation to ease of switching at the end of a contract. All, bar one, indicated that it was easy to switch. One healthcare sector Customer that did not find it easy to switch indicated that in general, switching provider is a significant task and they require their current service provider to give a 3-month notice if it decides to cease provision of services to them.
- 10.23 Of those Customers who mentioned that it was easy to switch, one made reference to payments for the unused linen stock and another mentioned that it takes up to 3 months to replace the linen stock of one supplier with the stock of another supplier. However, they noted that there are instances where, as a result of the tender, they changed suppliers and no service disruption occurred during the switching period. Therefore, overall, it was perceived that switching provider is relatively easy.
- 10.24 The Commission also asked whether Customers switched their provider of outsourced flat linen rental and maintenance services in the period from 1 January 2014 to 31 December 2017, 12 responses were received and eight<sup>650</sup> indicated that they switched in this period, (5 healthcare sector Customers, 3 hospitality sector Customers). The HSE

<sup>&</sup>lt;sup>650</sup> One other Hospitality respondent said that they switched from Celtic Linen to Kings Laundry, but said it was 6/7 years ago.



indicated that there were instances where, as a result of a tender process they changed suppliers to certain hospitals.

10.25 Of the other four healthcare sector Customers that responded to this question, three switched to Kings Laundry<sup>651</sup>, two from Celtic and one from CWS-boco, and one switched from Celtic to Berendsen. Two hospitality sector Customers switched to Kings Laundry (one from Celtic Linen and another from CWS-boco and Premier Linen). The reason given for the switch was that one customer was moving to one provider for all their hotels and the other Customer switched due for better quality of service and price. The third hospitality customer that switched, switched one hotel to OCL, the main reasons cited were price and keeping diversification of suppliers.

#### Likelihood of Switching to an OPL

- 10.26 When asked whether Customers self-supplied (i.e., had an OPL in the past), of the 14 responses received, 11 Customers stated that they did not self-supply in the recent past<sup>652</sup>. Of three Customers that previously had OPL (two healthcare sector Customers and one hospitality sector Customers); one healthcare sector Customers said that it did self-supply, but stopped due to high costs. [...] said that [...]. A hospitality sector Customer noted that 1 or 2 of its hotels currently self-supply linen laundry services, but that they intend to outsource these services.
- 10.27 The Commission also sought information on the possibility of switching to an onpremise laundry in response to a 5% - 10 % price increase of flat linen rental and maintenance services. Of the 13 responses received in relation to this question, all 13 said that switching to an on-premise laundry would not be a viable option. The main reasons given include:
  - Customer sites were unsuitable to host OPL facilities due to a lack of space required for such a facility;
  - There are substantial costs associated with switching to self-supply including (but not limited to): purchasing linen stock, equipment required for maintaining linen, OPL staff and other operational costs (e.g., electricity); and
  - lack of expertise in operating a laundry.

#### Likelihood of Switching to a wash-only launderette

- 10.28 In relation to the question of the possibility of switching to wash-only launderettes in response to a 5% 10 % price increase of flat linen rental and maintenance services, of the 12 responses received in relation to this question, all 12 Customers said that switching to wash-only launderettes would not be a viable option. The main reasons cited include:
  - Wash-only launderettes may not meet the necessary procedures/standards regarding infection control;

<sup>&</sup>lt;sup>651</sup> All three are private healthcare customers.

<sup>&</sup>lt;sup>652</sup> One healthcare sector Customer said that it might have had an OPL many years ago.



- Wash-only launderettes might not have sufficient capacity to handle the volume of linen required by Customers;
- the investment in linen stock that would be required if Customers were to purchase maintenance-only services.

#### Closest competitors of the parties

- 10.29 In response to the question on who are considered the closest competitors to the parties, 13 responses were received. Celtic Linen was the most frequently mentioned linen laundry with no other linen laundries mentioned by Healthcare sector customers. Healthcare sector customers indicated the following:
  - [...] said that Celtic Linen was the main competitor of Berendsen; and
  - [...] said that in the last tender Berendsen came distant second and Celtic Linen third and that they were not aware of any other laundry that could provide services to them<sup>653</sup>.
- 10.30 Hospitality sector Customers mentioned several linen laundries including Celtic Linen, OCL, and to a lesser extent CWS-boco and Sovereign Linen. One hospitality sector customer mentioned that Berendsen is not well known for serving customers in the hospitality sector and that it might [...]. In relation to smaller laundries (e.g., Premier Linen), this Customer noted that it would have reservations whether they can provide service quality similar to that of larger linen laundries. In addition, it also indicated that it would be reluctant to become the largest customer of a small laundry. Another hospitality sector Customer said that Berendsen, Kings Laundry, OCL and Celtic Linen submitted bids in response to its ITT. Lilliput told this Customer that it does not provide services in the State and did not submitted a bid. Ultimately, Celtic Linen won the tender. OCL was considered to be a good contender in the ITT and Customer had no concerns regarding their ability to deliver the required services.

#### Potential impact of the Proposed Transaction

10.31 Finally, Customers were asked to express their opinion about the Proposed Transaction and its possible impact on competition in the market<sup>654</sup> as well as prices of the flat linen rental and maintenance services. Table 24 illustrates Customers' views on the Proposed Transaction. In total 13 responses (8 healthcare sector Customers and 5 hospitality sector Customers) were received.

	Healthcare sector	Hospitality sector	Total
Don't know	0	1	1
Uncertain	0	1	1
No impact	1	2	3
Potentially negative	1	0	1
Negative	6	1	7

#### Table 24: Customers' views on the potential impact of the Proposed Transaction

<sup>&</sup>lt;sup>653</sup> Kings Laundry is their current provider.

<sup>&</sup>lt;sup>654</sup> Self-suppliers' were expressing views about competition in their respective sectors.



Source: The Commission's analysis based on views expressed by Self-suppliers

- 10.32 The main concerns expressed by Customers related to a reduction in competition and in particular for some healthcare customers the fact that the market structure will change (i.e., the number of suppliers will decrease from 3 to 2). It was also mentioned by a healthcare customer that if Celtic Linen were to go bankrupt, there will be a monopolist supplier of flat linen rental and maintenance services in the healthcare market. One healthcare sector Customer mentioned a potential reduction in quality of service. This Customer noted that Kings Laundry would have been the third largest player in the healthcare market and post-transaction there will just be two remaining players. Another healthcare sector Customer was of the view that Kings Laundry was trying to expand and may have tendered for the supply of linen, laundry and associated services for them in the future.
- 10.33 Hospitality sector Customers expressed lower level of concerns, with some saying it would have no impact and others saying it could be negative or positive. One hospitality sector Customer that thought the Proposed Transaction would have a negative impact on competition, noted that Kings Laundry is a dominant supplier in the hospitality sector and opined that Berendsen would likely to have an incentive to increase prices if the Proposed Transaction was put into effect
- 10.34 In relation to the question of the impact of the Proposed Transaction on the price of flat linen rental and maintenance services, 9 responses (6 healthcare sector Customers and 3 hospitality sector Customers) were received.
- 10.35 Of the six healthcare sector Customers, three Customers said they think prices would go up and two mentioned there would be less competition. One healthcare sector Customer said that if the parties were to try imposing higher prices, they would need to meet and negotiate terms and conditions of provided services and also look for potential alternative providers. Of the three Hospitality sector Customers one said that they would expect prices to increase, one said that it had some concerns about potential price increase and one said that it would expect no change.
- 10.36 No other concerns were identified by the Customers, with some of them reiterating that the Proposed Transaction will result in the reduction of competition. When asked whether any signs of co-ordination between linen laundries were observed when receiving bids/quotations, Customers noted that they did not see any signs of coordination.

#### Summary of responses provided by Self-suppliers

#### Description of OPLs

10.37 The Commission asked Self-suppliers to describe the equipment used in their OPLs. Table 25 indicates that all Self-suppliers have washing machines, tumble dryers, ironers/rollers and folders in their OPLs. All Self-suppliers own linen items.

Table 25: Equipment used in OPLs

[]	[]	[]	[]	[]	[]	[]	[]
8 washing	4 washer	4 washing	3 dryers,	washing	The hotel	The	Six washing
machines,	extractors	machines,	3	machines,	has 3	facility is	machines
3 tumble	5 washing	4 dryers, 1	washing	laundry,	washing	stand	Six tumble
dryers, 1	machines	ironer, 1	machines	roller,	machines,	alone, it	dryers
tunnel	1 barrier	folder	and 1	iron	5 Dryers,	consists	1 Bed sheet
finisher	washer		large		one	of a roller,	ironer/folder
and 2	5 Tumble		roller		roller, and	four	1 towel ironer
ironing	dryers				an	dryers,	
machines	5 Ironers				automatic	four	
	2 Bed				folder.	washing	
	sheet					machines.	
	folders						
	1 Shrink						
	Wrap						
	Machine						

Source: The Commission's analysis based on information provided by Self-suppliers

10.38 Only one out of 8 responding Self-suppliers noted that it provides flat linen and rental services to external third parties. In this regard, Merrion Hotel noted that it does linen laundry for a neighbouring restaurant. However, this was only possible, because it is a relatively small job<sup>655</sup>.

#### Costs of operating OPLs

10.39 Three Self-suppliers were able to provide estimates of operating OPLs. One hospitality Self-supplier noted that its annual OPL operating costs are circa €120,000. Two healthcare Self-suppliers noted that their OPL operating costs are in excess of €1 million. The major categories of operating costs cited by Self-suppliers were labour, utility costs (e.g., electricity) and linen stock replacement.

#### Reasons for switching to OPLs

- 10.40 2 out of 8 Self-suppliers (both from the hospitality sector) noted that they have previously purchased outsourced flat linen rental and maintenance services. Both Self-suppliers indicated that they switched to OPLs primarily due to insufficient level of service quality provided by linen laundries (e.g., delayed deliveries and/or linen cleanliness not meeting required standards) rather than high prices.
- 10.41 Both hotels switched to OPLs as part of a wider exercise of hotel refurbishment/extension. Both hotels noted that setting up OPLs required significant financial investments. In this regard, [...] was of the view that smaller hotels could never afford to set up OPLs. It also noted that most hotels don't have the room to expand or built a dedicated laundry facility, which is a big constraint.<sup>656</sup> [...] noted that it switched to self-supply because the opportunity to extend the hotel came up. Without the space they never would have switched to OPL.

<sup>&</sup>lt;sup>655</sup> See the Commission's phone record with [...] dated 24 January 2019.

<sup>&</sup>lt;sup>656</sup> See the Commission's phone record with [...] dated 24 January 2019.



Competition and Consumer Protection Commission

Intentions to switching from OPLs

10.42 One out of 8 Self-suppliers (Self-supplier from the healthcare sector) indicated that it considers switch to outsourced flat linen rental and maintenance services, but noted that concrete details of OPL closure are not established yet.

### Potential impact of the Proposed Transaction

10.43 Table 26 illustrates Self-suppliers' views on the Proposed Transaction. The majority of Self-suppliers' that expressed views on this matter (5 out of 7) noted that the proposed Transaction will not affect competition in healthcare and/or hospitality markets<sup>657</sup>. One Self-supplier from the hospitality sector did not have strong views, but noted that the reduced number of competitors could lead to an increase in prices in the hospitality market. Two Self-suppliers from the hospitality sector opined that if the merged entity were to increase its prices, the option of self-supply could become a viable alternative, particularly for larger hotels.

	Healthcare sector	Hospitality Sector	Total
Don't know	1	0	1
No impact	2	3	5
Potentially negative	0	1	1

Table 26: Self-suppliers' views on the potential impact of the Proposed Transaction

Source: The Commission's analysis based on views expressed by Self-suppliers

10.44 Self-suppliers in the healthcare sector noted that the Proposed Transaction did not affect them directly and thus, have relatively limited views on the impact of the Proposed Transaction on competition in the healthcare market.

<sup>&</sup>lt;sup>657</sup> Self-suppliers' were expressing views about competition in their respective sectors.



# 11. APPENDIX 2 – THE AMÁRACH SURVEY

11.1 The Amárach Survey conducted for the Commission by Amárach Research is presented alongside this Determination.



### **12.** APPENDIX 3 – ECONOMIC REPORTS

12.1 The report titled "Some observations on Frontier Economics' 'bidding markets' argument" conducted on behalf of the Commission by DotEcon Ltd. and the report titled "EXTERNAL ECONOMIST STUDY FOR THE CCPC BERENDSEN (ELIS)/KINGS LAUNDRY (M/18/063)" conducted on behalf of the Commission by PMCA Economic Consulting are presented alongside this Determination.<sup>658</sup>

 $<sup>^{\</sup>rm 658}$  That is the confidential version of this Determination issued to the Parties.



### **13.** APPENDIX 4 – ORGANISATIONS WITH OPLS

- 13.1 This appendix provides an overview of organisations in healthcare and hospitality sectors that operate OPLs. The organisations referred to in this appendix were identified by the Parties and Competitors during the Commission's investigation. Thus, they do not represent the definitive list of all organisations with OPLs in the State.
- 13.2 Table 27 lists hotels and hospitals with OPLs.

Organisation	Location	County	Size (in terms of beds)	Sector
[]	[]	Wexford	50-60	Hospitality
[]	[]	Wicklow	29	Hospitality
[]	[]	Wexford	78	Hospitality
[]	[]	Wexford	118	Hospitality
[]	[]	Dublin	270	Hospitality
[]	[]	Limerick	104	Hospitality
[]	[]	Dublin	142	Hospitality
[]	[]	Kilkenny	139	Hospitality
[]	[]	Kerry	244	Hospitality
[]	[]	Dublin	208	Hospitality
[]	[]	Dublin	192	Hospitality
[]	[]	Dublin	9	Hospitality
[]	[]	Kildare	39	Hospitality
[]	[]	Kilkenny	85	Hospitality
[]	[]	Limerick	89	Hospitality
[]	[]	Westmeath	35	Healthcare
[]	[]	Sligo	359	Healthcare
[]	[]	Limerick	75	Healthcare
[]	[]	Waterford	532	Healthcare
[]	[]	Galway	655	Healthcare
[]	[]	Drogheda	340	Healthcare
[]	[]	Kilkenny	285	Healthcare
[]	[]	Sligo	Unknown	Hospitality
[]	[]	Tipperary	Unknown	Healthcare
[]	[]	Kilkenny	285	Healthcare
[]	[]	Tipperary	Unknown	Healthcare
[]	[]	Tipperary	Unknown	Healthcare
[]	[]	Galway	Unknown	Healthcare

#### Table 27: Organisations with OPLs

Source: The Commission's analysis based on information provided by the parties and Competitors

13.3 Table 28 illustrates instances when organisations switched from or to OPLs in the period from 1 January 2014 to 30 June 2018 known to the Commission. It shows that in the hospitality market there were 9 hotels that switched to OPLs while 2 hotels switched from OPLs to purchase of outsourced flat linen rental and maintenance services. In the healthcare market, the Commission is aware of only one instance where healthcare



organisation switched to OPL. There were 4 instances where healthcare organisations switched from OPLs to purchase of outsourced flat linen rental and maintenance services.

Hospitality switching	customers g to OPLs		customers from OPLs	customer	thcare s switching OPLs		e customers g from OPLs
Name	Turnover (in €000's)	Name	Turnover (in €000's)	Name	Turnover (in €000's)	Name	Turnover (in €000's)
[]	€23.0	[]	€37.0	[]	Unknown	[]	€38.0
[]	€273.2	[]	€18.0			[]	€12.0
[]	€53.7					[]	€3.0
[]	€77.2					[]	€1.0
[]	€97.1						
[]	€15.0						
[]	€45.0						
[]	€57.0						
[]	€11.0						

Table 28: Organisations switching from/to OPLs, 1 January 2014 to 30 June 2018

Source: The Commission's analysis based on information provided by the parties and Competitors

13.4 Table 29 lists recently opened hotels and one care centre that decided to purchase outsourced flat linen rental and maintenance services. It shows that several significant hotels (in terms of potential turnover from linen laundry's perspective) have decided not to build OPLs.

Table 29: Recently opened hospitals and care centres that purchase outsourced flat linen rental
and maintenance services

Customer	Location	County	Size (in terms of projected revenue (in €000's))	Sector
[] <sup>659</sup>	[]	[]	[]	Hospitality
[]	[]	Wexford	20	Healthcare
[]	[]	Dublin	200	Hospitality
[]	[]	Dublin	140	Hospitality
[]	[]	Dublin	120	Hospitality
[]	[]	Cork	40.8	Hospitality

Source: The Commission's analysis based on information provided by the parties and Competitors

<sup>659 [...]</sup> will open in 2019.



### 14. APPENDIX 5 – GLOSSARY

14.1 In this Determination, the following terms shall have the definitions set out below. Where it is helpful for the reader to reference a defined term in the text, such term may also be defined in the text.

#### Table 30: Glossary of frequently used terms

Term	Definition	
Ashton Linen	Ashton Linen Service Limited	
Amárach	Amárach Research	
Berendsen	Berendsen Ireland Limited.	
Carraig Linen	Carraig Linen & Laundry Services Limited.	
СВР	Countervailing Buyer Power	
Celtic Linen	Celtic Linen Limited.	
СМА	The United Kingdom Competition and Markets Authority, previously the Competition Commission.	
Commission	The Competition and Consumer Protection Commission.	
Competitors	Linen laundries contacted by the Commission (apart from the parties)	
Counterfactual	Assessment of the most likely competitive situation in the absence of the Proposed Transaction.	
Customers	healthcare and hospitality customers contacted directly by the Commission	
CWS-boco	CWS-boco Ireland Limited.	
Dingle Linen	Dingle Linen Service.	
EC	European Commission	
Elis	Elis S.A.	
First Competitor Questionnaire	Questionnaire issued by the Commission dated 14 September 2018 to potential competitors of the merging parties.	



Term	Definition
First RFI	The Commission's Requirement for Further Information dated 14 September 2018 served on Berendsen and Kings Laundry, pursuant to section 20(2) of the Act.
flat linen	flat linen is an industry term describing offered linen products including flat-ironed bed sheets, pillowcases and tablecloths. In this Assessment this term also includes items such as towels, bath robes, scrub suits and other miscellaneous items.
GUPPI	Gross Upward Price Pressure Index
healthcare customers	Organisations in the healthcare sector in the State purchasing flat linen rental and maintenance services.
нні	The Herfindahl-Hirschman index
hospitality customers	Organisations in the hospitality sector in the State purchasing flat linen rental and maintenance services.
HSE	The Health Service Executive
IPR test	Indicative Price Rise test
ІТТ	Invitation to tender
Kings Laundry	Kings Laundry Limited, a private limited company registered in the State.
Lilliput	Lilliput (Dunmurry) Limited.
Limavady Linen	Limavady Linen Services Ltd.
Limerick Laundry	Limerick Laundry & Dry Cleaning Services Limited.
Linencare	Linencare Laundries Limited.
linen laundry	Provider of flat linen rental and maintenance services in the State.
maintenance	The term maintenance refers to the collection and processing (i.e., sorting, washing, drying, ironing, folding and packing) of used and soiled items and the delivery of clean items. Items may also be repaired as part of maintenance.
Merger Notification Form	Merger notification form submitted by the parties to the Commission dated 7 August 2018 in respect of the Proposed Transaction.



Competition and Consumer Protection Commission

Term	Definition
Millbrook Linen	Millbrook Linen Limited
NCA(s)	National competition authorities.
OCL	O.C.L. Laundry Services Limited.
OPL	On-premise laundry.
Phase 1	The investigation, by the Commission, of the Proposed Transaction to determine whether it will not lead to a SLC in any market for goods or services in the State.
Phase 1 Decision	The Commission's phase 1 decision dated 9 January 2019 to carry out the full investigation of the Proposed Transaction
Phase 2	A full investigation by the Commission of the Proposed Transaction following the Phase 1 Decision
Premier Linen	Premier Linen
РМСА	PMCA Economic Consulting
Questionnaire to Customers	Questionnaire issued by the Commission dated 24 September 2018 to customers purchasing flat linen rental and maintenance services
RABC	the risk analysis bio-contamination control
Report on Celtic Linen	Report on Celtic Linen submitted by the parties on 18 January 2019
Report on OPLs	Report on OPLs submitted by the parties on 18 January 2019
respondents	Organisations that participated in the Commission's Market Research
RFQ	Request for quotations
Second Competitor Questionnaire	Questionnaire issued by the Commission dated 12 February 2019 to competitors of the merging parties.
Second RFI	The Commission's Requirement for Further Information dated 12 February 2019 served on Berendsen and Kings Laundry, respectively, pursuant to section 20(2) of the Act.
Self-suppliers	Hotels and hospitals currently operating their own OPLs contacted directly by the Commission



Term	Definition
service providers	Providers of flat linen rental and maintenance services in the State.
SLC	Substantial lessening of competition
SSNIP	Small but significant non-transitory increase in price.
The Authority	the Commission's predecessor the Competition Authority
The Act	The Competition Act 2002, as amended.
The Amárach Survey	Independent piece of market research commissioned by the Commission and undertaken by Amárach Research.
The Assessment	The Commission's assessment of the Proposed Transaction issued to the Parties on 28 March 2019
The Commission's Merger Guidelines	"Guidelines for Merger Analysis" adopted by the Commission on 31 October 2014
The DotEcon Report	The report "Some observations on Frontier Economics' 'bidding markets' argument" conducted on behalf of the Commission by DotEcon Ltd.
The healthcare market	The market for the supply of outsourced flat linen rental and maintenance services to healthcare customers in the State.
The hospitality market	The market for the supply of outsourced flat linen rental and maintenance services to hospitality customers in the State.
The OFT	The Office of Fair Trading
The OGP	The Office of Government Procurement
The Parties	Comprises Berendsen and Kings Laundry.
The PMCA Report	The Report "EXTERNAL ECONOMIST STUDY FOR THE CCPC BERENDSEN (ELIS)/KINGS LAUNDRY (M/18/063)" conducted on behalf of the Commission by PMCA Economic Consulting
The Proposed Transaction	refers to the proposed transaction as notified to the Commission on 7 August 2018 whereby Berendsen would acquire sole control of Kings Laundry
The State	The Republic of Ireland
wash-only launderettes	Providers of linen maintenance services